UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington D.C. 20549

FORM 8-K

CURRENT REPORT

Pursuant to Section 13 or 15(d) of The Securities Exchange Act of 1934

Date of Report (Date of earliest event reported): April 29, 2013

Terra Tech Corp.

(Exact name of registrant as specified in its charter)

Nevada

(State or other jurisdiction of incorporation)

000-54358

(Commission File Number)

<u>26-3062661</u>

(IRS Employer Identification No.)

18101 Von Karman, Third Floor Irvine, California 92612

(Address of principal executive offices)(Zip Code)

(855) 447-6967

Registrant's telephone number, including area code

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions (see General Instruction A.2. below):		
	Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)	
	Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)	
	Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))	

Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

Item 1.01 Entry into a Material Definitive Agreement

Share Exchange Agreement with Edible Garden Corp.

On April 23, 2013, Terra Tech Corp., a Nevada corporation (the "Company"), entered into a Share Exchange Agreement, dated March 23, 2013 (the "Share Exchange Agreement"), by and among the Company, Edible Garden, LLC, a Nevada limited liability company ("Edible Garden"), and the holders of common stock of Edible Garden. The principle holders of the common stock of Edible Garden consisted of Ken VandeVrede, Mike VandeVrede, Steve VandeVrede and Dan VandeVrede, all of whom are brothers and each of whom held 23% of the issued and outstanding shares of common stock of Edible Garden. Additionally, Beverly Willekes, the sister of the four VandeVrede brothers, and David VandeVrede the father of the four VandeVrede brothers and Beverly Willekes, each held 4% of the issued and outstanding shares of common stock of Edible Garden.

Under the terms and conditions of the Share Exchange Agreement, the Company offered and sold 1,250,000 shares of common stock of the Company in consideration for all the issued and outstanding shares in Edible Garden. Separately, Amy Almsteier, offered and sold 6,750,000 of her 12,500,000 shares of Series B Preferred Stock to Ken VandeVrede, Mike VandeVrede, Steve VandeVrede and Dan VandeVrede, Beverly Willekes, and David VandeVrede (collectively, the "Edible Garden Shareholders"), each of whom acquired the Series B Preferred Stock on a pro-rata basis, based on their respective parentage equity interest in Edible Garden immediately prior to the consummation of the Share Exchange Agreement.

Each share of Series B Preferred Stock is convertible, at any time, at the option of the holder, on a 1-for-5.384325537 basis, into shares of common stock and has voting rights equal to 100 shares of common stock. The 6,750,000 shares of Series B Preferred Stock is convertible at any time into 41,290,091 shares of common stock and have voting power equal to 675,000,000 shares of common stock. The effect of the issuance of the 1,250,000 shares of common stock of the Company and the third-party offer and sale of the 6,750,000 shares of Series B Preferred Stock is that Edible Garden Shareholders now hold approximately 48.9% of the issued and outstanding shares of common stock of the Company and 53.7% of the voting power of the Corporation.

As a result of the share exchange Edible Garden is now a wholly-owned subsidiary of the Company. Edible Garden is a retailer seller of its line of locally grown hydroponic produce, which is distributed throughout the Northeast United States. Edible Garden, a premier brand of sustainably grown produce, has rapidly expanded its product availability from just over 100 retail stores to covering close to 400 retailers throughout New Jersey, Connecticut, Delaware, Maryland, New York and Pennsylvania in less than a year. The Company believes that this rapid expansion has been due to consumers demanding fresher produce grown locally using environmentally sustainable methods.

A condition to closing of the Share Exchange Agreement was that the Company shall have entered into an equity line financing of an amount not less than \$4,000,000, which condition was met when, on April 29, 2012, the Company entered into that Common Stock Purchase Agreement dated as of April 26, 2013 with Hanover Holdings I, LLC, a New York limited liability company (the "Investor").

In connection with the share exchange with Edible Garden, the Company and Amy Almsteier entered Stock Redemption Agreement dated May 6, 2013, pursuant to which the Company redeemed 500,000 shares of common stock of the Company from Ms. Almsteier for consideration of \$5.00.

Articles of Exchange were filed with the Secretary of the State of Nevada on April 24, 2013.

Common Stock Purchase Agreement with Hanover Holdings

On April 29, 2013 (the "Closing Date"), the Company entered into a Common Stock Purchase Agreement dated as of April 26, 2013 (the "Purchase Agreement") with Hanover Holdings I, LLC, a New York limited liability company (the "Investor"). The Purchase Agreement provides that, upon the terms and subject to the conditions set forth therein, the Investor is committed to purchase up to \$5,000,000 (the "Total Commitment") worth of the Company's common stock, \$0.001 par value (the "Shares"), over the 36-month term of the Purchase Agreement.

From time to time over the term of the Purchase Agreement, commencing on the trading day immediately following the date on which the initial registration statement is declared effective by the Securities and Exchange Commission (the "Commission"), as further discussed below, the Company may, in its sole discretion, provide the Investor with draw down notices (each, a "Draw Down Notice") to purchase a specified dollar amount of the Shares (the "Draw Down Amount") over a 10 consecutive trading day period, commencing on the trading day specified in the applicable Draw Down Notice (the "Pricing Period"), with each draw down subject to the limitations discussed below. The maximum amount of Shares requested to be purchased pursuant to any single Draw Down Notice cannot exceed 300% of the average daily trading volume of the Company's common stock for the ten trading days immediately preceding the date of the Draw Down Notice (the "Maximum Draw Down Amount").

Once presented with a Draw Down Notice, the Investor is required to purchase a pro rata portion of the applicable Draw Down Amount on each trading day during the applicable Pricing Period on which the daily volume weighted average price for our common stock (the "VWAP") equals or exceeds a floor price determined by the Company for such draw down (the "Floor Price"). If the VWAP falls below the applicable Floor Price on any trading day during the applicable Pricing Period, the Purchase Agreement provides that the Investor will not be required to purchase the pro rata portion of the applicable Draw Down Amount allocated to that trading day. The per share purchase price for the Shares subject to a Draw Down Notice shall be equal to 92.5% of the arithmetic average of the lowest VWAPs that equal or exceed the applicable Floor Price during the applicable Pricing Period, thowever, that if the VWAP does not equal or exceed the applicable Floor Price for at least five trading days during the applicable Pricing Period, then the per share purchase price shall be equal to 92.5% of the arithmetic average of all VWAPs that equal or exceed the applicable Floor Price during such Pricing Period. Each purchase pursuant to a draw down shall reduce, on a dollar-for-dollar basis, the Total Commitment under the Purchase Agreement.

The Company is prohibited from issuing a Draw Down Notice if (i) the amount requested in such Draw Down Notice exceeds the Maximum Draw Down Amount, (ii) the sale of Shares pursuant to such Draw Down Notice would cause the Company to issue or sell or the Investor to acquire or purchase an aggregate dollar value of Shares that would exceed the Total Commitment, or (iii) the sale of Shares pursuant to the Draw Down Notice would cause the Company to sell or the Investor to purchase an aggregate number of shares of the Company's common stock which would result in beneficial ownership by the Investor of more than 4.99% of the Company's common stock (as calculated pursuant to Section 13(d) of the Securities Exchange Act of 1934, as amended, and the rules and regulations thereunder). The Company cannot make more than one draw down in any Pricing Period and must allow 24 hours to elapse between the completion of the settlement of any one draw down and the commencement of a Pricing Period for any other draw down.

If the Company issues a Draw Down Notice and fails to deliver the shares to the Investor on the applicable settlement date, and such failure continues for 10 trading days, the Company agreed to pay the Investor, in addition to all other remedies available to the Investor under the Purchase Agreement, an amount in cash equal to 2.0% of the purchase price of such shares for each 30-day period the shares are not delivered, plus accrued interest.

Additionally, the Company paid to the Investor a commitment fee equal to \$125,000 (or 2.5 % of the Total Commitment under the Purchase Agreement) in the form of 595,239 restricted shares of the Company's common stock (the "Initial Commitment Shares"), calculated at a price equal to \$0.21 per share. The Initial Commitment Shares will be registered for resale in the Registration Statement, as discussed below.

The Purchase Agreement also provides for indemnification of the Investor and its affiliates in the event that the Investor incurs losses, liabilities, obligations, claims, contingencies, damages, costs and expenses related to a breach by the Company of any of its representations and warranties under the Purchase Agreement or the other related transaction documents or any action instituted against the Investor or its affiliates due to the transactions contemplated by the Purchase Agreement or other transaction documents, subject to certain limitations.

The Company agreed to pay up to \$15,000 of reasonable attorneys' fees and expenses incurred by the Investor in connection with the preparation, negotiation, execution and delivery of the Purchase Agreement and related transaction documentation.

Registration Rights Agreement with Hanover Holdings

In connection with the execution of the Purchase Agreement, on April 29, 2013, the Company and the Investor also entered into a Registration Rights Agreement dated as of the Closing Date (the "Registration Rights Agreement"). Pursuant to the Registration Rights Agreement, the Company has agreed to file an initial registration statement ("Registration Statement") with the Commission to register an agreed upon number of Shares, on or prior to April 29, 2013 (the "Filing Deadline") and have it declared effective at the earlier of (A) the 90th calendar day after the earlier of (1) the Filing Deadline and (2) the date of which such initial Registration Statement is filed with the Commission and (B) the fifth business day after the date the Company is notified by the Commission that such Registration Statement will not be reviewed or will not be subject to further review (the "Effectiveness Deadline").

If at any time all of the Registrable Securities (as defined in the Registration Rights Agreement) are not covered by the initial Registration Statement, the Company has agreed to file with the Commission one or more additional Registration Statements so as to cover all of the Registrable Securities not covered by such initial Registration Statement, in each case, as soon as practicable, but in no event later than the applicable filing deadline for such additional Registration Statements as provided in the Registration Rights Agreement.

The Company also agreed, among other things, to indemnify the Investor from certain liabilities and fees and expenses of the Investor incident to the Company's obligations under the Registration Rights Agreement, including certain liabilities under the Securities Act of 1933, as amended (the "Securities Act"). The Investor has agreed to indemnify and hold harmless the Company and each of its directors, officers and persons who control the Company against certain liabilities that may be based upon written information furnished by the Investor to the Company for inclusion in a registration statement pursuant to the Registration Rights Agreement, including certain liabilities under the Securities Act.

The foregoing descriptions of the Purchase Agreement and Registration Rights Agreement are qualified in their entirety by reference to the provisions of the Purchase Agreement and Registration Rights Agreement filed as exhibits 10.1 and 10.2 to this Current Report on the Form 8-K (this "Report"), respectively, which are incorporated herein by reference.

Offer and Sale of 6% Senior Secured Convertible Debenture

On March 26, 2013, the Company filed a Current Report on Form 8-K (file No. 000-54258; the "March 26, 2013 Form 8-K"), disclosing, among other things, that the Company had entered into a Securities Purchase Agreement (the "Securities Purchase Agreement") with certain accredited investors (the "Purchasers") pursuant to which the Company raised \$825,000 in a private placement financing (the "Offering"). The Company further disclosed that, on March 22, 2013, after the satisfaction of certain closing conditions, the Offering closed and the Company issued to the Purchasers three 6% Senior Secured Convertible Debentures for aggregate proceeds of \$825,000. Each debenture accrues interest at a rate of 6% per annum and is convertible into shares of common stock of the Company, at any time at the election of the holder thereof, at a purchase prices equal to 62% of the lowest daily volume weighted average price (or the "VWAP") of the Company's common stock as quoted by Bloomberg L.P. for the ten (10) trading days immediately preceding the conversion date.

On April 30, 2013, the Company filed a Current Report on Form 8-K disclosing that on April 19, 2013, the Company offered and sold an additional 6% Senior Secured Convertible Debenture for aggregate proceeds of \$250,000. The \$250,000 sale was made pursuant to the Securities Purchase Agreement with one purchaser, who also was one of three purchasers of debentures disclosed in the March 26, 2013 Form 8-K..

On May 3, 2013, the Company offered and sold, pursuant to Securities Purchase Agreement, an additional 6% Senior Secured Convertible Debenture to one purchaser, for aggregate proceeds of \$200,000. In connection with the sale, the purchaser became a party to that certain General Security Agreement, dated March 22, 2013, by and among the Company and the purchasers of 6% Senior Secured Convertible Debentures identified in the March 26, 2013 Form 8-K, and that certain General Security Agreement, dated March 22, 2013, by and among GrowOp Technology Ltd., a wholly-owned subsidiary of the Company and the purchasers of 6% Senior Secured Convertible Debentures identified in the March 26, 2013 Form 8-K. Under each General Security Agreement, the Company and GrowOp Technology, respectively, granted security interests in all of their respective assets, rights, interests and after-acquired assets and properties as collateral for repayment of the principal and interest owed under the debentures.

Aegis Capital Corp. (the "Placement Agent") served as the placement agent of the Company for the sale of the \$200,000 debenture. In consideration for services rendered as the Placement Agent, the Company agreed to: (i) pay to the Placement Agent a cash commission equal to \$16,000, or 8.0% of the gross proceeds received in the sale of the \$200,000 debenture, and (ii) issue to the Placement Agent, or its designee, a Warrant to purchase that number of shares of common stock of the Company equal to 5% of the aggregate number of shares of the Company's common stock underlying the debentures sold in the offering, at an exercise price equal to the purchase price of the common stock underlying the debentures, for a term of 5 years (the "Agent Warrants").

The shares of common stock underlying the \$200,000 debenture are restricted securities and cannot be resold without registration under the Securities Act of 1933, as amended, or an exemption from registration.

Item 3.02 Unregistered Sales of Equity Securities

Reference is made to the disclosure set forth under Items 1.01 of this Report, which disclosure is incorporated herein by reference.

The company offered and sold the 1,250,000 shares of common stock to Edible Garden Shareholders pursuant to the exemption for registration afforded by Section 4(2) of the Securities Act of 1933, as amended (the "Securities Act"), in a non-public offering to 6 investors, where the investors had registration-type information about the Company.

The Company offered and sold the Initial Commitment Shares and the sale of the Shares to the Investor under the Purchase Agreement pursuant to the exemption for registration afforded by Section 4(2) of the Securities Act and Rule 506 of Regulation D, promulgated under the Securities Act, where the Investor was an "accredited investor" within the meaning of Rule 501 of Regulation D.

The Company offered and sold the \$200,000 6% Senior Secured Convertible Debenture, to one purchaser, pursuant to the exemption for registration afforded by Section 4(2) of the Securities Act and Rule 506 of Regulation D, promulgated under the Securities Act, where the purchaser was an "accredited investor" within the meaning of Rule 501 of Regulation D.

Item 5.01 Changes in Control of Registrant

Reference is made to the disclosure set forth under Items 1.01 of this Report, which disclosure is incorporated herein by reference. The effect of the issuance of the 1,250,000 shares of common stock of the Company and the third-party offer and sale of the 6,750,000 shares of Series B Preferred Stock is that Edible Garden Shareholders now hold approximately 48.9% of the issued and outstanding shares of common stock of the Company and 53.7% of the voting power of the Corporation.

While the Edible Garden shareholders still voting control of the Company, Amy Almsteier, the Company's current Secretary, Treasurer and a Director, and Derek Peterson, the Company's current President and Chief Executive officer, still each hold 50 shares of Series A Preferred Stock, which amounts to all of the issued and outstanding shares Series A Preferred Stock of the Company.

The following is a summary of the material rights and restrictions associated with our Series A Preferred Stock. Each share of Series A Preferred Stock is convertible on a one-for-one basis into common stock and has all of the voting rights that the holders of our common stock has and so long as any shares of Series A Preferred Stock are outstanding, the Company shall not, without first obtaining the approval (by vote or written consent, as provided by law) of the holders of at least a majority of the then outstanding shares of Series A Preferred Stock, voting as a separate class:

- (1) amend our Articles of Incorporation or Bylaws;
- (2) change or modify the rights, preferences or other terms of the Series A Preferred Stock, or increase or decrease the number of authorized shares of Series A Preferred Stock;

- (3) reclassify or recapitalize any outstanding equity securities, or authorize or issue, or undertake an obligation to authorize or issue, any equity securities (or any debt securities convertible into or exercisable for any equity securities) having rights, preferences or privileges senior to or on a parity with the Series A Preferred Stock;
- (4) authorize or effect any transaction constituting a Deemed Liquidation (as defined in this subparagraph), or any other merger or consolidation of the Corporation. For purposes of this paragraph, a "Deemed Liquidation" shall mean: (A) the closing of the sale, transfer or other disposition of all or substantially all of the Corporation's assets (including an irrevocable or exclusive license with respect to all or substantially all of the Corporation's intellectual property); (B) the consummation of a merger, share exchange or consolidation with or into any other corporation, limited liability company or other entity (except one in which the holders of capital stock of the Corporation as constituted immediately prior to such merger, share exchange or consolidation continue to hold at least 50% of the voting power of the capital stock of the Corporation or the surviving or acquiring entity (or its parent entity)), (C) authorize or effect any transaction liquidation, dissolution or winding up of the Company, either voluntary or involuntary, provided, however, that none of the following shall be considered a Deemed Liquidation: (i) a merger effected exclusively for the purpose of changing the domicile of the Corporation, or (ii) a transaction or other event deemed to be exempt from the definition of a Deemed Liquidation by the holders of at least a majority of the then outstanding Series A Preferred Stock;
- (5) increase or decrease the size of the Board of Directors as provided in the Bylaws of the Corporation or remove any of the Series A Directors (unless approved by the Board of Directors including the Series A Directors);
- (6) declare or pay any dividends or make any other distribution with respect to any class or series of capital stock (unless approved by the Board of Directors including the Series A Directors);
- (7) redeem, repurchase or otherwise acquire (or pay into or set aside for a sinking fund for such purpose) any outstanding shares of capital stock (other than the repurchase of shares of common stock from employees, consultants or other service providers pursuant to agreements approved by the Board of Directors under which the Corporation has the option to repurchase such shares at no greater than original cost upon the occurrence of certain events, such as the termination of employment) (unless approved by the Board of Directors including the Series A Directors);
- (8) amend any stock option plan of the Corporation, if any (other than amendments that do not require approval of the stockholders under the terms of the plan or applicable law) or approve any new equity incentive plan;
 - (9) replace the President and/or Chief Executive Officer of the Corporation (unless approved by the Board of Directors including the Series A Directors); or
 - (10) transfer assets to any subsidiary or other affiliated entity.

Item 5.02 Departure of Directors or Certain Officers; Election of Directors; Appointment of Certain Officers; Compensatory Arrangements of Certain Officers.

On April 24, 2013, the board of directors of the Company appointed Steve VandeVrede and Mike VandeVrede as directors of the Company.

Steve VandeVrede, age 27, has served as Vice-President of Naturally Beautiful Plant Products LLC, since January 2007. Mr. Vande Vrede is also currently an owner of Gro-rite Landscape Services LLC. Since September 2010, Mr. Vande Vrede has also served as Director of New Business and Marketing at Edible Garden, which is a sustainable hydroponic herbs and lettuce grower that provides fresh, locally grown herbs and leafy greens to supermarkets and restaurants. From 2003 to 2005, Mr. Vande Vrede attended Quinnipiac University, and from 2005 to 2007, he attended William Patterson University, where he obtained a degree in Business Finance Management. Mr. VandeVrede's experience in finance, gardening retail and specialty farming led to our conclusion that Mr. Vande Vrede should be serving as a member of our board of directors in light of our business and structure.

Mike VandeVrede, age 33, has served as President of Naturally Beautiful Plant Products LLC, since January 2007. Mr. Vande Vrede is also currently an owner of Gro-rite Landscape Services LLC. Since September 2010, Mr. Vande Vrede has also served as Director at Edible Garden, which is a sustainable hydroponic herbs and lettuce grower that provides fresh, locally grown herbs and leafy greens to supermarkets and restaurants. Mr. VandeVrede's experience as President of Naturally Beautiful Plant Products LLC led to our conclusion that Mr. Vande Vrede should be serving as a member of our board of directors in light of our business and structure.

Steve VandeVrede and Mike VandeVrede are brothers of Ken VandeVrede, currently a director of the Company. These three VandeVrede brothers currently hold three of the seven seats of the board of directors of the company.

Neither Steve VandeVrede nor Mike VandeVrede has any agreement, arrangement or understanding with the Company in connection with being appointed a director of the Company, except that being appointed a director was a condition of closing of the share exchange under the Share Exchange Agreement referenced in Item 1.01 of this Current Report on Form 8-K.

Item 9.01 Financial Statements and Exhibits

(d) Exhibits:

Exhibit	Description
2.1	Share Exchange Agreement, dated April 24, 2013, by and among the Registrant, Edible Garden Corp., a Nevada corporation, and the holders of common stock of Edible Garden Corp.
2.2	Form of Articles of Share Exchange
10.1	Common Stock Purchase Agreement dated April 26, 2013 (the "Purchase Agreement") by and between the Registrant and Hanover Holdings I, LLC, a New York limited liability company
10.2	Registration Rights Agreement dated April 26, 2013, by and between the Registrant and Hanover Holdings I, LLC, a New York limited liability company
10.3	Stock Redemption Agreement dated May 6, 2013, by and between the Registrant and Amy Almsteier
10.4	Form of Securities Purchase Agreement (1)
10.5	Form of 6% Senior Secured Convertible Debenture (1)
10.6	Form of General Security Agreement (1)
10.7	Form of Stock Pledge Agreement (1)

⁽¹⁾ Incorporated by reference to Current report on Form 8-K (File No. 000-54258), filed with the Commission on March 26, 2013.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

> Terra Tech Corp. (Registrant)

Date: May 6, 2013

By: /s/ Derek Peterson

Name: Derek Peterson Title: President and Chief Executive Officer

EXHIBIT INDEX

Exhibit	Description
	-
2.1	Share Exchange Agreement, dated April 24, 2013, by and among the Registrant, Edible Garden Corp., a Nevada corporation, and the holders of common stock
	of Edible Garden Corp.
2.2	Form of Articles of Share Exchange
10.1	Common Stock Purchase Agreement dated April 26, 2013 (the "Purchase Agreement") by and between the Registrant and Hanover Holdings I, LLC, a New
	York limited liability company
10.2	Registration Rights Agreement dated April 26, 2013, by and between the Registrant and Hanover Holdings I, LLC, a New York limited liability company
10.3	Stock Redemption Agreement dated May 6, 2013, by and between the Registrant and Amy Almsteier
10.4	Form of Securities Purchase Agreement (1)
10.5	Form of 6% Senior Secured Convertible Debenture (1)
10.6	Form of General Security Agreement (1)
10.7	Form of Stock Pledge Agreement (1)

⁽¹⁾ Incorporated by reference to Current report on Form 8-K (File No. 000-54258), filed with the Commission on March 26, 2013.

SHARE EXCHANGE AGREEMENT

THIS SHARE EXCHANGE AGREEMENT (the "Agreement") dated as of April 24, 2013, is entered into by and among Terra Tech Corp., a Nevada corporation ("Terra Tech"), Edible Garden Corp., a Nevada corporation ("Edible Garden"), and the shareholders of Edible Garden listed on Schedule 1 to this Agreement (each, a "Shareholder" and, collectively, the "Shareholders").

RECITALS

- A. The principal Shareholders have substantial experience in hydroponic farming and have formed Edible Garden to facilitate implementation of a large scale plan to produce and supply fresh herbs and leafy greens to supermarkets and restaurants through implementation of a branded marketing strategy and investments in greenhouse facilities and a state-of-the-art semi-automated hydroponic growing system.
- B. The Shareholders own the number of shares of capital stock of Edible Garden (the "Shares") set forth opposite each Shareholder's name or <u>Schedule 1</u>, which Shares collectively constitute all of the issued and outstanding shares of capital stock in Edible Garden.
- C. Terra Tech desires to purchase from the Shareholders, and the Shareholders desire to sell to Terra Tech, the Shares in exchange for shares of Terra Tech Common Stock, all on the terms and subject to the conditions set forth in this Agreement (the "Exchange").
 - D. As a result of the Exchange, Terra Tech will become the sole shareholder of Edible Garden.
 - E. Certain capitalized terms used in this Agreement are defined on Exhibit A hereto.

AGREEMENT

In consideration of the agreements, provisions and covenants set forth below, Terra Tech, Edible Garden and the Shareholders, hereby agree as follows:

ARTICLE I. EXCHANGE OF SHARES

1.1 Agreement to Sell.

Upon the terms and subject to all of the conditions contained herein, each of the Shareholders hereby agrees to sell, assign, transfer and deliver to Terra Tech, and Terra Tech hereby agrees to purchase and accept from each of the Shareholders, on the Closing Date, the Shares.

1.2 Purchase Price.

As full consideration for the sale, assignment, transfer and delivery of the Shares by the Shareholders to Terra Tech, and upon the terms and subject to all of the conditions contained herein, Terra Tech shall issue to the Shareholders an aggregate of 1,250,000 shares of Terra Tech common stock (the "Acquisition Shares") on a pro rata basis based upon their respective beneficial ownership interest in Edible Garden, as certified by the President of Edible Garden, at the Closing.

1.3 Mechanics of Exchange.

At the Closing, each Shareholder shall deliver to Terra Tech the duly endorsed certificate or certificates that immediately prior to the Closing represented the Edible Garden Shares of Common Stock (the "Certificates") and Terra Tech shall cause to be issued and delivered to each Shareholder a certificate(s) evidencing ownership by each Shareholder of a proportionate share of the Acquisition Shares.

1.4 No Fractional Shares.

No fraction of a share of Terra Tech Common Stock shall be issued in the Exchange. In lieu of fractional shares, the Shareholders upon surrender of their Certificates as set forth in Section 1.3 shall be issued that number of shares of common stock resulting by rounding up to the nearest whole number of shares of Acquisition Shares that each such Shareholder shall receive as a result of the Exchange.

1.5 Closing.

The closing of the transactions contemplated by this Agreement (the "Closing") shall take place at 9:00 a.m., Pacific Standard Time, at the principal administrative offices of Terra Tech, or at a location mutually agreement upon by Terra Tech and Edible Garden, on or before May 1, 2013 (the "Closing Date"); provided, however, that if all of the other conditions set forth in Articles VI and VII hereof are not satisfied or waived, unless this agreement has been terminated under Section 9 hereof, the Closing Date shall be the business day following the day on which all such conditions have been satisfied or waived, or at such other date, time and place as Terra Tech, Edible Garden and the Shareholders shall agree.

ARTICLE II. REPRESENTATIONS AND WARRANTIES OF EDIBLE GARDEN AND SHAREHOLDERS

Except as set forth in the Disclosure Schedule regarding Edible Garden, attached hereto (the "Edible Garden Disclosure Schedule"), Edible Garden represents and warrants to Terra Tech, and the Shareholders represent and warrant severally and to their knowledge (each Shareholder's representations and warranties hereunder pertain solely to that Shareholder and Edible Garden and not to other Shareholders), as follows:

2.1 Organization and Qualification.

Edible Garden is duly incorporated, validly and in good standing existing under the laws of Nevada, has all requisite authority and power (corporate and other), governmental licenses, authorizations, consents and approvals to carry on its business as presently conducted and as contemplated to be conducted, to own, hold and operate its properties and assets as now owned, held and operated by it, to enter into this Agreement, to carry out the provisions hereof except where the failure to be in good standing or to have such governmental licenses, authorizations, consents and approvals will not, in the aggregate, either (i) have a Material Adverse Effect on the business, assets or financial condition of Edible Garden, or (ii) impair the ability of Edible Garden to perform its material obligations under this Agreement. Edible Garden is duly qualified, licensed or domesticated as a foreign corporation in good standing in each jurisdiction wherein the nature of its activities or its properties owned or leased requires such qualification, licensing or domestication, except where the failure to be so qualified, licensed or domesticated will not have a Material Adverse Effect. Set forth as part of the Edible Garden Disclosure Schedule is a list of those jurisdictions in which each of Edible Garden presently conducts its business, owns, holds and operates its properties and assets.

2.2 Subsidiaries

Edible Garden does not own directly or indirectly, any equity or other ownership interest in any corporation, partnership, joint venture or other entity or enterprise. Edible Garden does not have any direct or indirect interests of stock ownership or otherwise in any corporation, partnership, joint venture, firm, association or business enterprise, and is not party to any agreement to acquire such an interest.

2.3 Articles of Incorporation and Bylaws.

The copies of the charter document and corporate governance document of Edible Garden (collectively, the "Organizational Documents") that have been delivered to Terra Tech prior to the execution of this Agreement are true and complete and have not been amended or repealed. Edible Garden is not in violation or breach of any of the provisions of the Organizational Documents, except for such violations or breaches which, in the aggregate, will not have a Material Adverse Effect on Edible Garden.

2.4 Authorization and Validity of this Agreement.

This Agreement and each of the Transaction Agreements constitute the legal, valid and binding obligation of each person or entity who is a party thereto (other than Terra Tech), enforceable against each such person or entity in accordance with its terms, except as such enforcement is limited by general equitable principles, or by bankruptcy, insolvency and other similar laws affecting the enforcement of creditors rights generally. Each Edible Garden shareholder has all requisite legal capacity to execute and deliver this Agreement and the Transaction Agreements to which he or she is a party, and to perform its, his or her obligations hereunder and thereunder. The execution and delivery by each of Edible Garden and each of the Shareholders of this Agreement and the Transaction Agreements (to the extent either is a party thereto), and the consummation of the transactions contemplated herein and therein (the "Transactions") have been authorized by all necessary corporate or other action on the part of Edible Garden and each of the Shareholders. This Agreement and the Transaction Agreements have been duly executed and delivered by the parties thereto (other than Terra Tech).

2.5 No Violation.

Neither the execution nor delivery of this Agreement or the Transaction Agreements, nor the consummation or performance of any of the Transactions by Edible Garden or the Shareholders will directly or indirectly:

(i) violate or conflict with any provision of the Organizational Documents of Edible Garden; (B) result in (with or without notice or lapse of time) a violation or breach of, or conflict with or constitute a default or result in the termination or in a right of termination or cancellation of, or accelerate the performance required by, or require notice under, any agreement, promissory note, lease, instrument or arrangement to which Edible Garden or any of its assets are bound or result in the creation of any Liens upon Edible Garden or any of its assets; (C) violate any order, writ, judgment, injunction, ruling, award or decree of any Governmental Body; ("Governmental Body"); (D) violate any statute, law or regulation of any jurisdiction as such statute, law or regulation that relates to the Shareholders or Edible Garden or any of the assets of Edible Garden; or (E) result in cancellation, modification, revocation or suspension of any permits, licenses, registrations, consents, approvals, authorizations or certificates issued or granted by any Governmental Body which are held by or granted to the Shareholders or Edible Garden or which are necessary for the conduct of Edible Garden's business; or

(ii) to the knowledge of Edible Garden or any of the Shareholders, cause Edible Garden to become subject to, or to become liable for the payment of, any Tax (as hereinafter defined) or cause any of the assets owned by Edible Garden to be reassessed or revalued by any taxing authority or other Governmental Body.

None of Edible Garden or the Shareholders is or will be required to give any notice to or obtain any approval, consent, ratification, waiver or other authorization (a "Consent") from any person or entity (including, without limitation, any Governmental Body) in connection with (i) the execution and delivery of this Agreement or any of the Transaction Agreements, or (ii) the consummation or performance of any of the Transactions.

2.6 Capitalization and Related Matters.

(a) Capitalization. Edible Garden has issued and outstanding one hundred (100) shares of common stock. Except as set forth in the preceding sentence, no other class of capital stock or other security of Edible Garden is authorized, issued, reserved for issuance or outstanding. The Shareholders, as of the Closing Date, are the lawful, record and beneficial owners of the number of Edible Garden Shares of Common Stock set forth opposite each Seller's name on Schedule 1 attached hereto. The Shareholders have, as of the date hereof and as of the Closing Date, valid and marketable title to their respective Shares, free and clear of all Liens (including, without limitation, any claims of spouses under applicable community property laws) and are the lawful, record and beneficial owners of all of the Shares. Except as is issued to and held by the Shareholders or Edible Garden, no other class of capital stock or other security of Edible Garden, as applicable, is authorized, issued, reserved for issuance or outstanding. At the Closing, Terra Tech will be vested with good and marketable title to the Shares, free and clear of all Liens (including, without limitation, any claims of spouses under applicable community property laws). No legend or other reference to any purported Lien appears upon any certificate representing the Shares. Each of the Shares has been duly authorized and validly issued and is fully paid and nonassessable.

(b) No Redemption Requirements. There are no authorized or outstanding options, warrants, equity securities, calls, rights, commitments or agreements of any character by which Edible Garden or any of the Shareholders is obligated to issue, deliver or sell, or cause to be issued, delivered or sold, any shares of capital stock or other securities of Edible Garden. There are no outstanding contractual obligations (contingent or otherwise) of Edible Garden to retire, repurchase, redeem or otherwise acquire any outstanding shares of capital stock of, or other ownership interests in, Edible Garden or to provide funds to or make any investment (in the form of a loan, capital contribution or otherwise) in any other entity.

2.7 Compliance with Laws and Other Instruments.

Except as would not have a Material Adverse Effect, the business and operations of Edible Garden has been and are being conducted in accordance with all applicable foreign, federal, provincial and local laws, rules and regulations and all applicable orders, injunctions, decrees, writs, judgments, determinations and awards of all courts and governmental agencies and instrumentalities. There are no permits, bonuses, registrations, consents, approvals, authorizations, certificates, or any waiver of the foregoing, which are required to be issued or granted by a Governmental Body for the conduct of the business as presently conducted or the ownership of the assets of Edible Garden. Except as would not have a Material Adverse Effect, Edible Garden is not, and has not received notice alleging that it is, in violation of, or (with or without notice or lapse of time or both) in default under, or in breach of, any term or provision of the Organizational Documents or of any indenture, loan or credit agreement, note, deed of trust, mortgage, security agreement or other material agreement, lease, license or other instrument, commitment, obligation or arrangement to which Edible Garden is a party or by which any of Edible Garden's properties, assets or rights are bound or affected. To the knowledge of Edible Garden, no other party to any material contract, agreement, lease, license, commitment, instrument or other obligation to which Edible Garden is a party is (with or without notice or lapse of time or both) in default thereunder or in breach of any term thereof. Edible Garden is not subject to any obligation or restriction of any kind or character, nor is there, to the knowledge of Edible Garden, any event or circumstance relating to Edible Garden that materially and adversely affects in any way its business, properties, assets or prospects or that prohibits Edible Garden from entering into this Agreement and the Transaction Agreements or would prevent or make burdensome its performance of or compliance with all or any pa

2.8 Certain Proceedings.

There are no outstanding or pending preceding that has been commenced against or involving Edible Garden or any of its assets and, to the knowledge of Edible Garden and the Shareholders, no matters of the foregoing nature are contemplated or threatened. None of Edible Garden or the Shareholders have been charged with, and is not threatened with, or under any investigation with respect to, any allegation concerning any violation of any provision of any federal, provincial, local or foreign law, regulation, ordinance, order or administrative ruling, and is not in default with respect to any order, writ, injunction or decree of any Governmental Body.

2.9 No Brokers or Finders.

None of Edible Garden, the Shareholders, or any officer, director, independent contractor, consultant, agent or employee of Edible Garden has agreed to pay, or has taken any action that will result in any person or entity becoming obligated to pay or entitled to receive, any investment banking, brokerage, finder's or similar fee or commission in connection with this Agreement or the Transactions. Edible Garden and the Shareholders shall jointly and severally indemnify and hold Terra Tech harmless against any liability or expense arising out of, or in connection with, any such claim.

2.10 Title to and Condition of Properties.

Edible Garden has good, valid and marketable title to all of its properties and assets (whether real, personal or mixed, and whether tangible or intangible) reflected as owned in its books and records, free and clear of all Liens. Edible Garden owns or holds under valid leases or other rights to use all real property, plants, machinery, equipment and all assets necessary for the conduct of its business as presently conducted, except where the failure to own or hold such property, plants, machinery, equipment and assets would not have a Material Adverse Effect on Edible Garden.

2.11 Absence of Undisclosed Liabilities.

Edible Garden has no debt, obligation or liability (whether accrued, absolute, contingent, liquidated or otherwise, whether asserted or unasserted, whether due or to become due, whether or not known to Edible Garden) arising out of any transaction entered into prior to the Closing Date or any act or omission prior to the Closing Date which individually or taken together would constitute a Material Adverse Effect on Edible Garden, except to the extent specifically set forth on or reserved against in the books and records of Edible Garden.

The books and records of Edible Garden fairly present in all material respects the financial condition, assets and liabilities of Edible Garden, as applicable, taken as a whole, as of the dates and periods indicated, and were prepared in accordance with GAAP (except as otherwise indicated therein or in the notes thereto).

2.12 Changes.

Edible Garden has not, since its formation:

- (a) Ordinary Course of Business. Conducted its business or entered into any transaction other than in the Ordinary Course of Business, except for this Agreement and the Transaction Agreements.
- (b) Adverse Changes. Suffered or experienced any change in, or affecting, its condition (financial or otherwise), properties, assets, liabilities, business, operations, results of operations or prospects which would have a Material Adverse Effect;
- (c) Loans. Made any loans or advances to any Person other than travel advances and reimbursement of expenses made to employees, officers and directors in the Ordinary Course of Business;
- (d) <u>Compensation and Bonuses</u>. Made any payments of any bonuses or compensation other than regular salary payments, or increase in the salaries, or payment on any of its debts in the Ordinary Course of Business, to any of its shareholders, directors, officers, employees, independent contractors or consultants or entry into by it of any employment, severance, or similar contract with any director, officer, or employee, independent contractor or consultant; Adopted, or increased in the payments to or benefits under, any profit sharing, bonus, deferred compensation, savings, insurance, pension, retirement, or other employee benefit plan for or with any of its employees;
 - (e) Liens. Created or permitted to exist any Lien on any of its properties or assets other than Permitted Liens;
- (f) <u>Capital Stock</u>. Issued, sold, disposed of or encumbered, or authorized the issuance, sale, disposition or encumbrance of, or granted or issued any option to acquire any shares of its capital stock or any other of its securities or any Equity Security, or altered the term of any of its outstanding securities or made any change in its outstanding shares of capital stock or its capitalization, whether by reason of reclassification, recapitalization, stock split, combination, exchange or readjustment of shares, stock dividend or otherwise; changed its authorized or issued capital stock; granted any stock option or right to purchase shares of its capital stock; issued any security convertible into any of its capital stock; granted any registration rights with respect to shares of its capital stock; purchased, redeemed, retired, or otherwise acquired any shares of its capital stock; declared or paid any dividend or other distribution or payment in respect of shares of capital stock of any other entity;
 - (g) Dividends. Declared, set aside, made or paid any dividend or other distribution to any of its shareholders;
- (h) <u>Material Contracts</u>. Terminated or modified any of its Material Contract except for termination upon expiration in accordance with the terms of such agreements, a description of which is included in the Edible Garden's Disclosure Schedule;

- (i) <u>Claims</u>. Released, waived or cancelled any claims or rights relating to or affecting Edible Garden in excess of \$1,000 in the aggregate or instituted or settled any Proceeding involving in excess of \$10,000 in the aggregate;
- (j) <u>Discharged Liabilities</u>. Paid, discharged, cancelled, waived or satisfied any claim, obligation or liability in excess of \$1,000 in the aggregate, except for liabilities incurred prior to the date of this Agreement in the Ordinary Course of Business;
- (k) <u>Indebtedness</u>. Created, incurred, assumed or otherwise become liable for any Indebtedness or commit to any endeavor involving a commitment in excess of \$1,000 in the aggregate, other than contractual obligations incurred in the Ordinary Course of Business;
 - (1) Guarantees. Guaranteed or endorsed in a material amount any obligation or net worth of any Person;
 - (m) Acquisitions. Acquired the capital stock or other securities or any ownership interest in, or substantially all of the assets of, any other Person;
- (n) <u>Accounting</u>. Changed its method of accounting or the accounting principles or practices utilized in the preparation of its financial statements, other than as required by GAAP;
 - (o) Agreements. Entered into any agreement, or otherwise obligated itself, to do any of the foregoing.

2.13 Material Contracts.

Edible Garden has delivered to Terra Tech, prior to the date of this Agreement, true, correct and complete copies of each of its Material Contracts.

(a) No Defaults. The Material Contracts of Edible Garden are valid and binding agreements of Edible Garden, as applicable, and are in full force and effect and are enforceable in accordance with their terms. Except as would not have a Material Adverse Effect, Edible Garden is not in breach or default of any of its Material Contracts to which it is a party and, to the knowledge of Edible Garden, no other party to any of its Material Contracts is in breach or default thereof. Except as would not have a Material Adverse Effect, no event has occurred or circumstance has existed that (with or without notice or lapse of time) would (a) contravene, conflict with or result in a violation or breach of, or become a default or event of default under, any provision of any of its Material Contracts or (b) permit Edible Garden or any other Person the right to declare a default or exercise any remedy under, or to accelerate the maturity or performance of, or to cancel, terminate or modify any of its Material Contracts. Edible Garden has not received any notice and has no knowledge of any pending or threatened cancellation, revocation or termination of any of its Material Contracts to which it is a party, and there are no renegotiations of, or attempts to renegotiate.

2.14 Tax Returns and Audits.

- (a) <u>Tax Returns</u>. Edible Gardens was organized during 2013 and, to date, has not filed any Tax Returns and, to the knowledge of Edible Gardens and the Shareholders, has not incurred any obligations to pay any Taxes.
- (b) No Disputes. There is no pending audit, examination, investigation, dispute, proceeding or claim with respect to any Taxes of or Tax Return required to be filed by Edible Garden, nor is any such claim or dispute pending or contemplated.
- (c) No Tax Allocation, Sharing. Edible Garden is not a party to any Tax allocation or sharing agreement. Edible Garden (a) has not been a member of a Tax Group filing a consolidated income Tax Return under Section 1501 of the Code (or any similar provision of provincial, local or foreign law), and (b) does not have any liability for Taxes for any Person under Treasury Regulations Section 1.1502-6 (or any similar provision of provincial, local or foreign law) as a transferee or successor, by contract or otherwise.

2.15 Material Assets.

The books and records of Edible Garden reflect the material properties and assets (real and personal) owned or leased by them.

2.16 Insurance Coverage.

Edible Garden has no insurance or general liability policies maintained by Edible Garden on its properties and assets.

2.17 Litigation; Orders.

There is no Proceeding (whether federal, provincial, local or foreign) pending or, to the knowledge of Edible Garden, threatened or appealable against or affecting Edible Garden or any of its properties, assets, business or employees. To the knowledge of Edible Garden, there is no fact that might result in or form the basis for any such Proceeding. Edible Garden is not subject to any Orders and has not received any written opinion or memorandum or legal advice from their legal counsel to the effect that Edible Garden is exposed, from a legal standpoint, to any liability which would be material to its business. Edible Garden is not engaged in any legal action to recover monies due it or for damages sustained by any of them.

2.18 Licenses.

Edible Garden is not presently required to possess, and does not presently possess, any licenses, permits, authorizations, approvals, franchises and rights (collectively "Permits") although it is contemplated that it may be necessary for Edible Garden to obtain and possess certain Permits to engage in its business as contemplated in the future and to permit it to own and use its properties and assets in the manner in which it contemplates owning and using such properties and assets.

2.19 Bank Accounts and Safe Deposit Boxes

The Edible Garden Disclosure Schedule discloses the title and number of each bank or other deposit or financial account, and each lock box and safety deposit box used by Edible Garden, the financial institution at which that account or box is maintained and the names of the persons authorized to draw against the account or otherwise have access to the account or box, as the case may be.

2.20 Stock Option Plans; Employee Benefits.

Edible Garden does not have any employee benefit plans or arrangements covering their present and former employees or providing benefits to such persons in respect of services provided to Edible Garden. Edible Garden has no commitment, whether formal or informal and whether legally binding or not, to create any additional plan, arrangement or practice similar to the Approved Plans.

2.21 Employee Matters.

- (a) No former or current employee of Edible Garden is a party to, or is otherwise bound by, any agreement or arrangement (including, without limitation, any confidentiality, non-competition or proprietary rights agreement) that in any way adversely affected, affects, or will affect (i) the performance of his, her or its duties to Edible Garden, or (ii) the ability of Edible Garden to conduct its business.
- (b) Edible Garden has no employees, directors, officers, consultants, independent contractors, representatives or agents whose contract of employment or engagement cannot be terminated by three months' notice.
- (c) Edible Garden is in compliance with all applicable laws respecting employment and employment practices, terms and conditions or employment and wages and hours, and is not engaged in any unfair labor practice. There is no labor strike, dispute, shutdown or stoppage actually pending or, to the knowledge of Edible Garden or the Shareholders, threatened against or affecting Edible Garden.

2.22 Environmental and Safety Matters.

Except as would not have a Material Adverse Effect:

- (a) Edible Garden has at all times been and is in compliance with all Environmental Laws and Orders applicable to Edible Garden, as applicable.
- (b) There are no Proceedings pending or, to the knowledge of Edible Garden, threatened against Edible Garden alleging the violation of any Environmental Law or Environmental Permit applicable to Edible Garden or alleging that Edible Garden is a potentially responsible party for any environmental site contamination. None of Edible Garden or the Shareholders are aware of, or has ever received notice of, any past, present or future events, conditions, circumstances, activities, practices, incidents, actions or plans which may interfere with or prevent continued compliance, or which may give rise to any common law or legal liability, or otherwise form the basis of any claim, action, suit, proceeding, hearing or investigation, based on or related to the manufacture, processing, distribution, use, treatment, storage, disposal, transport, or handling, or the emission, discharge, release or threatened release into the environment, of any pollutant, contaminant, or hazardous or toxic material or waste.

(c) Neither this Agreement nor the consummation of the transactions contemplated by this Agreement shall impose any obligations to notify or obtain the consent of any Governmental Body or third Persons under any Environmental Laws applicable to Edible Garden.

2.23 Money Laundering Laws.

The operations of Edible Garden are and have been conducted at all times in compliance with applicable financial record-keeping and reporting requirements of the money laundering statutes of all U.S. and non-U.S. jurisdictions, the rules and regulations thereunder and any related or similar rules, regulations or guidelines, issued, administered or enforced by any Governmental Body (collectively, the "Money Laundering Laws") and no Proceeding involving Edible Garden with respect to the Money Laundering Laws is pending or, to the knowledge of Edible Garden, threatened.

2.24 Disclosure.

- (a) Any information set forth in this Agreement, the Edible Garden Disclosure Schedule, or the Transaction Agreements shall be true, correct and complete in all material respects.
- (b) No statement, representation or warranty of Edible Garden or the Shareholders in this Agreement (taken with the Schedules) or the Transaction Agreements or any exhibits or schedules thereto contain any untrue statement of a material fact or omits to state a material fact necessary to make the statements herein or therein, taken as a whole, in light of the circumstances in which they were made, not misleading.
- (c) Except as set forth in the Edible Garden Disclosure Schedule, the Shareholders and Edible Garden have no knowledge of any fact that has specific application to Edible Garden (other than general economic or industry conditions) and that adversely affects the assets or the business, prospects, financial condition, or results of operations of Edible Garden.
- (d) In the event of any inconsistency between the statements in the body of this Agreement and those in the Schedules (other than an exception expressly set forth as such in the Schedules with respect to a specifically identified representation or warranty), the statements in the Schedules shall control.
- (e) The books of account, minute books and stock record books of Edible Garden, all of which have been made available to Terra Tech, are complete and accurate and have been maintained in accordance with sound business practices. Without limiting the generality of the foregoing, the minute books of Edible Garden contain complete and accurate records of all meetings held, and corporate action taken, by the shareholders, the boards of directors, and committees of the boards of directors of Edible Garden, as applicable, and no meeting of any such shareholders, board of directors, or committee has been held for which minutes have not been prepared and are not contained in such minute books.

ARTICLE III. REPRESENTATIONS AND WARRANTIES OF TERRA TECH

Terra Tech hereby represents and warrants to the Shareholders as of the date hereof:

3.1 Organization; Good Standing.

Terra Tech is duly incorporated, validly existing and in good standing under the laws of Nevada, has all requisite authority and power (corporate and other), governmental licenses, authorizations, consents and approvals to carry on its business as presently conducted and as contemplated to be conducted, to own, hold and operate its properties and assets as now owned, held and operated by it, to enter into this Agreement, to carry out the provisions hereof except where the failure to be in good standing or to have such governmental licenses, authorizations, consents and approvals will not, in the aggregate, either (i) have a Material Adverse Effect on the business, assets or financial condition of Terra Tech, or (ii) impair the ability of Terra Tech to perform its material obligations under this Agreement. Terra Tech is duly qualified, licensed or domesticated as a foreign corporation in good standing in each jurisdiction wherein the nature of its activities or its properties owned or leased requires such qualification, licensing or domestication, except where the failure to be so qualified, licensed or domesticated will not have a Material Adverse Effect.

3.2 Terra Tech Capitalization.

As of the date hereof and on the Closing Date, the authorized and outstanding capital of Terra Tech consists of:

(a) authorized capital stock which consists of 350,000,000 shares of common stock, par value \$0.001 per share (the "Common Stock"), and 25,000,000 shares of preferred stock, par value \$0.001 per share (the "Preferred Stock"), of which 85,611,781 shares of Common Stock and 14,750,100 shares of Preferred Stock are issued and outstanding. 100 shares of Preferred Stock have been designated as Series A Preferred Stock (convertible into 100 shares of common stock), and all 100 of such shares are issued and outstanding. 24,999,900 shares of Preferred Stock have been designated as Series B Preferred Stock, and 14,750,000 shares of such Series B Preferred Stock (convertible into 79,418,802 shares of Common Stock) are issued and outstanding. All of such outstanding shares have been validly issued and are fully paid and nonassessable. No shares of Common Stock are subject to preemptive rights or any other similar rights or any liens or encumbrances suffered or permitted by Terra Tech;

(b) outstanding warrants (the "Warrants") to purchase 6,780,704 shares of Common Stock at an exercise price of \$0.33 with a term of 4 years, for aggregate gross proceeds of \$2,237,632; and

(c) outstanding 6% Senior Secured Convertible Debentures (the "Debentures" and, together with the outstanding Common Stock, Series A Preferred Stock, Series B Preferred Stock and Warrants, the "Terra Tech Securities") in an aggregate amount of \$1,075,000. Each Debenture accrues interest at a rate of 6% per annum and is convertible into shares of common stock of Terra Tech, at any time at the election of the holder thereof, at a purchase prices equal to 62% of the lowest daily volume weighted average price (or the "VWAP") of Terra Tech's common stock as quoted by Bloomberg L.P. for the ten (10) trading days immediately preceding the conversion date.

The Acquisition Shares, when issued in connection with this Agreement and the other Transactional Agreements, will be duly authorized, validly issued, fully paid and nonassessable. The Terra Tech Securities are not, and the Acquisition Shares are not and will not be as of the Closing, subject to any preemptive or subscription right. Other than the Terra Tech Securities, there is no outstanding voting trust agreement or other contract, agreement, arrangement, option, warrant, call, commitment or other right of any character obligating or entitling Terra Tech to issue, sell, redeem or repurchase any of its securities, and there is no outstanding security of any kind convertible into or exchangeable for the Common Stock of Terra Tech, nor has Terra Tech, or any of its agents orally agreed to issue any of the foregoing. There are no declared or accrued unpaid dividends with respect to any shares of Terra Tech Common Stock. There are no agreements, written or oral, between Terra Tech and any of its stockholders or among any stockholders relating to the acquisition (including without limitation rights of first refusal or preemptive rights), or disposition, or registration under the Securities Act or voting of the capital stock of Terra Tech. There are no outstanding shares of Common Stock that are subject to vesting. Terra Tech has no capital stock other than the Common Stock and Preferred Stock authorized, issued or outstanding.

3.3 Authority; Binding Nature of Agreements.

- (a) The execution, delivery and performance of this Agreement, the Transactional Agreements, and all other agreements and instruments contemplated to be executed and delivered by Terra Tech in connection herewith have been duly authorized by all necessary corporate action on the part of Terra Tech and its board of directors.
- (b) This Agreement, the Transactional Agreements, and all other agreements and instruments contemplated to be executed and delivered by Terra Tech constitute the legal, valid and binding obligation of Terra Tech, enforceable against Terra Tech in accordance with their terms, except to the extent that enforceability may be limited by applicable bankruptcy, insolvency, moratorium or other laws affecting the enforcement of creditors' rights generally and by general principles of equity regardless of whether such enforceability is considered in a proceeding in law or equity.

(c) There is no pending Proceeding, and, to Terra Tech's knowledge, no Person has threatened to commence any Proceeding that challenges, or that may have
the effect of preventing, delaying, making illegal or otherwise interfering with, the Exchange or Terra Tech's ability to comply with or perform its obligations and covenants under
the Transactional Agreements, and, to the knowledge of Terra Tech, no event has occurred, and no claim, dispute or other condition or circumstance exists, that might directly or
indirectly give rise to or serve as a basis for the commencement of any such Proceeding.

3.4 Non-contravention; Consents.

The execution and delivery of this Agreement and the other Transactional Agreements, and the consummation of the Exchange, by Terra Tech will not, directly or indirectly (with or without notice or lapse of time):

- (a) contravene, conflict with or result in a material violation of (i) Terra Tech's Certificate of Incorporation or Bylaws, or (ii) any resolution adopted by Terra Tech Board or any committee thereof or the stockholders of Terra Tech;
- (b) to the knowledge of Terra Tech, contravene, conflict with or result in a material violation of, or give any Governmental Body the right to challenge the Exchange or to exercise any remedy or obtain any relief under, any legal requirement or any Order to which Terra Tech or any material assets owned or used by it are subject;
- (c) to the knowledge of Terra Tech, cause any material assets owned or used by Terra Tech to be reassessed or revalued by any taxing authority or other Governmental Body;
- (d) to the knowledge of Terra Tech, contravene, conflict with or result in a material violation of any of the terms or requirements of, or give any Governmental Body the right to revoke, withdraw, suspend, cancel, terminate or modify, any Governmental Authorization that is held by Terra Tech or that otherwise relates to Terra Tech's business or to any of the material assets owned or used by Terra Tech, where such contraventions, conflict, violation, revocation, withdrawal, suspension, cancellation, termination or modification would have a Material Adverse Effect on Terra Tech;
 - (e) contravene, conflict with or result in a material violation or material breach of, or material default under, any Contract to which Terra Tech is a party;
- (f) give any Person the right to any payment by Terra Tech or give rise to any acceleration or change in the award, grant, vesting or determination of options, warrants, rights, severance payments or other contingent obligations of any nature whatsoever of Terra Tech in favor of any Person, in any such case as a result of the Exchange; or
 - (g) result in the imposition or creation of any material Lien upon or with respect to any material asset owned or used by Terra Tech.

Except for Consents, filings or notices required under the state and federal securities laws or any other laws or regulations or as otherwise contemplated in this Agreement and the other Transactional Agreements, Terra Tech will not be required to make any filing with or give any notice to, or obtain any Consent from, any Person in connection with the execution and delivery of this Agreement and the other Transactional Agreements or the consummation or performance of the Exchange.

3.5 Finders and Brokers.

- (a) Neither Terra Tech nor any Person acting on behalf of Terra Tech has engaged any finder, broker, intermediary or any similar Person in connection with the Exchange.
 - (b) Terra Tech has not entered into a contract or other agreement that provides that a fee shall be paid to any Person or Entity if the Exchange is consummated.

3.6 Reports and Financial Statements; Absence of Certain Changes.

(a) Terra Tech has filed all reports required to be filed with the SEC pursuant to the Exchange Act since December 23, 2008 (all such reports, including those to be filed prior to the Closing Date and all registration statements and prospectuses filed by Terra Tech with the SEC, are collectively referred to as the "Terra Tech SEC Reports). All of the Terra Tech SEC Reports, as of their respective dates of filing (or if amended or superseded by a filing prior to the date of this Agreement, then on the date of such filing): (i) complied in all material respects as to form with the applicable requirements of the Securities Act or Exchange Act and the rules and regulations thereunder, as the case may be, and (ii) did not contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading. The audited financial statements of Terra Tech included in the Terra Tech SEC Reports comply in all material respects with the published rules and regulations of the SEC with respect thereto, and such audited financial statements (i) were prepared from the books and records of Terra Tech, (ii) were prepared in accordance with GAAP applied on a consistent basis (except as may be indicated therein or in the notes or schedules thereto) and (iii) present fairly the financial position of Terra Tech as of the dates thereof and the results of operations and cash flows for the periods then ended. The unaudited financial statements (i) were prepared from the books and records of Terra Tech, (ii) were prepared in accordance with GAAP, except as otherwise permitted under the Exchange Act and the rules and regulations thereunder, on a consistent basis (except as may be indicated therein or in the notes or schedules thereto) and (iii) present fairly the financial position of Terra Tech as of the dates thereof and the results of operations and cash flows (or changes in financia

(b) Except as specifically contemplated by this Agreement or reflected in the Terra Tech SEC Reports, since December 23, 2008 there has not been (i) any material adverse change in Terra Tech's business, assets, liabilities, operations, and, to the knowledge of Terra Tech, no event has occurred that is likely to have a material adverse effect on Terra Tech's business, assets, liabilities or operations, (ii) any declarations setting aside or payment of any dividend or distribution with respect to the Terra Tech Common Stock other than consistent with past practices, (iii) any material change in Terra Tech's accounting principles, procedures or methods, (iv) cancellation in writing of any material customer contract or (v) the loss of any customer relationship which would have a material adverse effect on Terra Tech's business, assets, liabilities or operations.

3.7 Compliance with Applicable Law.

Except as disclosed in the Terra Tech SEC Reports filed prior to the date of this Agreement and except to the extent that the failure or violation would not in the aggregate have a Material Adverse Effect on the business, results of operations or financial condition of Terra Tech, to Terra Tech's knowledge Terra Tech holds all Governmental Authorizations necessary for the lawful conduct of its business under and pursuant to, and the business of Terra Tech is not being conducted in violation of, any Governmental Authorization applicable to Terra Tech.

3.8 Complete Copies of Requested Reports.

Terra Tech has delivered or made available true and complete copies of each document that has been reasonably requested by Edible Garden or the Shareholders.

3.9 Full Disclosure

- (a) Neither this Agreement (including all Schedules and exhibits hereto) nor any of the Transactional Agreements contemplated to be executed and delivered by Terra Tech in connection with this Agreement contains any untrue statement of material fact; and none of such documents omits to state any material fact necessary to make any of the representations, warranties or other statements or information contained therein not misleading.
- (b) All of the information regarding Terra Tech and the business, condition, assets, liabilities, operations, financial performance, net income and prospects that has been furnished to Edible Garden or the Shareholders by or on behalf of Terra Tech or any of the Terra Tech's Representatives, is accurate and complete in all material respects.

3.10 Acknowledgments Regarding Edible Garden

Terra Tech acknowledges and is aware that:

- (a) Edible Garden is a newly formed entity organized to commercialize certain concepts, strategies and rights conceived and/or contributed to Edible Garden by one or more of the Shareholders and, except for such organizational and capitalization activities, has conducted no material operations to date;
- (b) Other than this Agreement and the other Transactional Agreements and as contemplated by this Agreement and the other Transactional Agreements, Edible Garden has no material Contracts, has no licenses or permits, has no customers and has no material assets, whether tangible or intangible; and

(c) The Shareholders have direct and indirect interests in, or relationships to, one or more entities which may compete, do business or otherwise have potential conflicts, with Edible Garden; nothing herein shall preclude the Shareholders from continuing such relationships or interests following closing of the Exchange.

ARTICLE IV. COVENANTS OF EDIBLE GARDEN

4.1 Access and Investigation.

Edible Garden shall ensure that, at all times during the Pre-Closing Period:

- (a) Edible Garden and its Representatives provide Terra Tech and its Representatives access, at reasonable times and with twenty-four (24) hours' notice from Terra Tech to Edible Garden, to all of the premises and assets of Edible Garden, to all existing books, records, Tax Returns, work papers and other documents and information relating to Edible Garden, and to responsible officers and employees of Edible Garden, and Edible Garden and its Representatives provide Terra Tech and its Representatives with copies of such existing books, records, Tax Returns, work papers and other documents and information relating to Edible Garden as Terra Tech may request in good faith;
- (b) Each of Edible Garden and its Representatives confer regularly with Terra Tech upon its request, concerning operational matters and otherwise report regularly (not less than semi-monthly and as Terra Tech may otherwise request) to Terra Tech and discuss with Terra Tech and its Representatives concerning the status of the business, condition, assets, liabilities, operations, and financial performance of Edible Garden, and promptly notify Terra Tech of any material change in the business, condition, assets, liabilities, operations, and financial performance of Edible Garden, or any event reasonably likely to lead to any such change.

4.2 Operation of the Business.

Edible Garden shall ensure that, during the Pre-Closing Period:

- (a) It conducts its operations in the Ordinary Course of Business and in the same manner as such operations have been conducted prior to the date of this Agreement;
- (b) It uses its commercially reasonable efforts to preserve intact its current business organization, keep available and not terminate the services of its current officers and employees and maintain its relations and goodwill with all suppliers, customers, landlords, creditors, licensors, licensees, employees and other Persons having business relationships with Edible Garden;

- (c) It does not declare, accrue, set aside or pay any dividend or make any other distribution in respect of any shares of its capital stock, and does not repurchase, redeem or otherwise reacquire any shares of its capital stock or other securities, except with respect to the repurchase of shares of Edible Garden Common Stock upon termination of employees at the original purchase price pursuant to agreements existing at the date hereof;
 - (d) It does not sell or otherwise issue (or grant any warrants, options or other rights to purchase) any shares of capital stock or any other securities;
- (e) It does not amend its charter document, corporate governance document or other Organizational Documents, and does not affect or become a party to any recapitalization, reclassification of shares, stock split, reverse stock split or similar transaction;
 - (f) It does not form any subsidiary or acquire any equity interest or other interest in any other Entity;
- (g) It does not establish or adopt any Employee Benefit Plan, and does not pay any bonus or make any profit sharing or similar payment to, or increase the amount of the wages, salary, commissions, fringe benefits or other compensation or remuneration payable to, any of its directors, officers or employees;
 - (h) It does not change any of its methods of accounting or accounting practices in any respect;
 - (i) It does not make any Tax election;
 - (j) It does not commence or take any action or fail to take any action which would result in the commencement of any Proceeding;
- (k) It does not (i) acquire, dispose of, transfer, lease, license, mortgage, pledge or encumber any fixed or other assets, other than in the Ordinary Course of Business; (ii) incur, assume or prepay any indebtedness, Indebtedness or obligation or any other liabilities or issue any debt securities, other than in the Ordinary Course of Business; (iii) assume, guarantee, endorse for the obligations of any other person, other than in the Ordinary Course of Business; (iv) make any loans, advances or capital contributions to, or investments in, any other Person, other than in the Ordinary Course of Business; or (v) fail to maintain insurance consistent with past practices for its business and property;
 - (l) It pays all debts and Taxes, files all of its Tax Returns (as provided herein) and pays or performs all other obligations, when due;
- (m) It does not enter into or amend any agreements pursuant to which any other Person is granted distribution, marketing or other rights of any type or scope with respect to any of its services, products or technology;
 - (n) It does not hire any new officer-level employee;

- (o) It does not revalue any of its assets, including, without limitation, writing down the value of inventory or writing off notes or accounts receivable, except as required under GAAP and in the Ordinary Course of Business;
 - (p) Except as otherwise contemplated hereunder, it does not enter into any transaction or take any other action outside the Ordinary Course of Business; and
- (q) It does not enter into any transaction or take any other action that likely would cause or constitute a Breach of any representation or warranty made by it in this Agreement.

4.3 Filings and Consents; Cooperation.

Edible Garden shall ensure that:

- (a) Each filing or notice required to be made or given (pursuant to any applicable Law, Order or contract, or otherwise) by Edible Garden or the Shareholders in connection with the execution and delivery of any of the Transactional Agreements, or in connection with the consummation or performance of the Exchange, is made or given as soon as possible after the date of this Agreement;
- (b) Each Consent required to be obtained (pursuant to any applicable Law, Order or contract, or otherwise) by Edible Garden or the Shareholders in connection with the execution and delivery of any of the Transactional Agreements, or in connection with the consummation or performance of the Exchange, is obtained as soon as possible after the date of this Agreement and remains in full force and effect through the Closing Date;
- (c) It promptly delivers to Terra Tech a copy of each filing made, each notice given and each Consent obtained by Edible Garden during the Pre-Closing Period; and
- (d) During the Pre-Closing Period, it and its Representatives cooperate with Terra Tech and Terra Tech's Representatives, and prepare and make available such documents and take such other actions as Terra Tech may request in good faith, in connection with any filing, notice or Consent that Terra Tech is required or elects to make, give or obtain.

4.4 Notification; Updates to Disclosure Schedules.

- (a) During the Pre-Closing Period, Edible Garden shall promptly notify Terra Tech in writing of:
- (i) the discovery by it of any event, condition, fact or circumstance that occurred or existed on or prior to the date of this Agreement which is contrary to any representation or warranty made by it in this Agreement or in any of the other Transactional Agreements, or that would upon the giving of notice or lapse of time, result in any of its representations and warranties set forth in this agreement to become untrue or otherwise cause any of the conditions of Closing set forth in Article VIII or Article VIII not to be satisfied:

(ii) any event, condition, fact or circumstance that occurs, arises or exists after the date of this Agreement (except as a result of actions taken pursuant to the express written consent of Terra Tech) and that is contrary to any representation or warranty made by it in this Agreement, or that would upon the giving of notice or lapse of time, result in any of its representations and warranties set forth in this agreement to become untrue or otherwise cause any of the conditions of Closing set forth in Article VII or Article VIII not to be satisfied;

- (b) If any event, condition, fact or circumstances that is required to be disclosed pursuant to Section 4.4(a) requires any material change in the Edible Garden Disclosure Schedule, or if any such event, condition, fact or circumstance would require such a change assuming the Edible Garden Disclosure Schedule were dated as of the date of the occurrence, existence or discovery of such event, condition, fact or circumstances, then Edible Garden, as applicable, shall promptly deliver to Terra Tech an update to the Edible Garden Disclosure Schedule specifying such change (a "Disclosure Schedule Update").
 - (c) It will promptly update any relevant and material information provided to Terra Tech after the date hereof pursuant to the terms of this Agreement.

4.5 Commercially Reasonable Efforts.

During the Pre-Closing Period, Edible Garden shall use its commercially reasonable efforts to cause the conditions set forth in Article VII and Article VIII to be satisfied on a timely basis and so that the Closing can take place on or before May 1, 2013, in accordance with Section 1.5, and shall not take any action or omit to take any action, the taking or omission of which would or could reasonably be expected to result in any of the representations and warranties of Edible Garden set forth in this Agreement becoming untrue, or in any of the conditions of Closing set forth in Article VIII or Article VIII not being satisfied.

ARTICLE V. COVENANTS OF TERRA TECH

5.1 Notification.

During the Pre-Closing Period, Terra Tech shall promptly notify Edible Garden in writing of:

- (a) the discovery by Terra Tech of any event, condition, fact or circumstance that occurred or existed on or prior to the date of this Agreement which is contrary to any representation or warranty made by Terra Tech in this Agreement; and,
- (b) any event, condition, fact or circumstance that occurs, arises or exists after the date of this Agreement (except as a result of actions taken pursuant to the written consent of Edible Garden) and that is contrary to any representation or warranty made by Terra Tech in this Agreement;

5.2 Filings and Consents; Cooperation.

Terra Tech shall ensure that:

- (a) Each filing or notice required to be made or given (pursuant to any applicable Law, Order or contract, or otherwise) by Terra Tech in connection with the execution and delivery of any of the Transactional Agreements, or in connection with the consummation or performance of the Exchange, is made or given as soon as possible after the date of this Agreement;
- (b) Each Consent required to be obtained (pursuant to any applicable Law, Order or contract, or otherwise) by Terra Tech in connection with the execution and delivery of any of the Transactional Agreements, or in connection with the consummation or performance of the Exchange, is obtained as soon as possible after the date of this Agreement and remains in full force and effect through the Closing Date;
- (c) Terra Tech promptly delivers to Edible Garden and a copy of each filing made, each notice given and each Consent obtained by Terra Tech during the Pre-Closing Period; and
- (d) During the Pre-Closing Period, Terra Tech and its Representatives cooperate with Edible Garden and their Representatives, and prepare and make available such documents and take such other actions as Edible Garden may request in good faith, in connection with any filing, notice or Consent that Edible Garden is required or elects to make, give or obtain.

5.3 Commercially Reasonable Efforts.

During the Pre-Closing Period, Terra Tech shall use its commercially reasonable efforts to cause the conditions set forth in Article VII and Article VIII to be satisfied on a timely basis and so that the Closing can take place on or before May 1, 2013, or as soon thereafter as is reasonably practical, in accordance with Section 1.5, and shall not take any action or omit to take any action, the taking or omission of which would or could reasonably be expected to result in any of the representations and warranties or Terra Tech set forth in this Agreement becoming untrue or in any of the conditions of closing set forth in Article VIII or Article VIII not being satisfied.

ARTICLE VI. MUTUAL COVENANTS

6.1 Disclosure of Confidential Information.

(a) Each of Terra Tech and the Shareholders acknowledges and agrees that it may receive Confidential Information in connection with this Transaction including without limitation, the Edible Garden Disclosure Schedule and any information disclosed during the due diligence process, the public disclosure of which will harm the disclosing party's business. The Receiving Party may use Confidential Information only in connection with the Transaction. The results of the due diligence review may not be used for any other purpose other than in connection with the Transaction. Except as expressly provided in this Agreement, the Receiving Party shall not disclose Confidential Information to anyone without the Disclosing Party's prior written consent. The Receiving Party shall take all reasonable measures to avoid disclosure, dissemination or unauthorized use of Confidential Information, including, at a minimum, those measures it takes to protect its own confidential information of a similar nature. The Receiving Party shall not export any Confidential Information in any manner contrary to the export regulations of the governmental jurisdiction to which it is subject.

- (b) The Receiving Party may disclose Confidential Information as required to comply with binding orders of governmental entities that have jurisdiction over it, provided that the Receiving Party (i) gives the Disclosing Party reasonable notice (to the extent permitted by law) to allow the Disclosing Party to seek a protective order or other appropriate remedy, (ii) discloses only such information as is required by the governmental entity, and (iii) uses commercially reasonable efforts to obtain confidential treatment for any Confidential Information so disclosed.
- (c) All Confidential Information shall remain the exclusive property of the Disclosing Party. The Disclosing Party's disclosure of Confidential Information shall not constitute an express or implied grant to the Receiving Party of any rights to or under the Disclosing Party's patents, copyrights, trade secrets, trademarks or other intellectual property rights.
- (d) The Receiving Party shall notify the Disclosing Party immediately upon discovery of any unauthorized use or disclosure of Confidential Information or any other breach of this Agreement by the Receiving Party. The Receiving Party shall cooperate with the Disclosing Party in every reasonable way to help the Disclosing Party regain possession of such Confidential Information and prevent its further unauthorized use.
- (e) The Receiving Party shall return or destroy all tangible materials embodying Confidential Information (in any form and including, without limitation, all summaries, copies and excerpts of Confidential Information) promptly following the Disclosing Party's written request; provided, however, that, subject to the provisions of this Agreement, the Receiving Party may retain one copy of such materials in the confidential, restricted access files of its legal department for use only in the event a dispute arises between the parties related to the Transaction and only in connection with that dispute. At the Disclosing Party's option, the Receiving Party shall provide written certification of its compliance with this Section.

6.2 Indemnification.

(a) Subject to the limitations set forth in Section 6.2(b) below, each of Edible Garden and the Shareholders, jointly and severally, each shall defend, indemnify and hold harmless Terra Tech, and its respective employees, officers, directors, stockholders, controlling persons, affiliates, agents, successors and assigns (collectively, the "Terra Tech Indemnified Persons"), and shall reimburse the Terra Tech Indemnified Person, for, from and against any loss, liability, claim, damage, expense (including costs of investigation and defense and reasonable attorneys' fees) or diminution of value, whether or not involving a third-party claim (collectively, "Damages"), directly or indirectly, relating to, resulting from or arising out of:

(i) any untrue representations, misrepresentations or breach of warranty by or of Edible Garden or the Shareholders contained in or pursuant to this Agreement, and the Edible Garden Disclosure Schedule; (ii) any breach or nonfulfillment of any covenant, agreement or other obligation by or of Edible Garden or the Shareholders (only to the extent made or occurring prior to or at the Closing) contained in or pursuant to this Agreement, the Transaction Agreements executed by Edible Garden or any of the Shareholders in their individual capacity, the Edible Garden Disclosure Schedule, or any of the other agreements, documents, schedules or exhibits to be entered into by Edible Garden or any of the Shareholders in their individual capacity pursuant to or in connection with this Agreement;

(iii) all of Pre-Closing liabilities of Edible Garden or the Shareholders; and

- (iv) any liability, claim, action or proceeding of any kind whatsoever, whether instituted or commenced prior to or after the Closing Date, which directly or indirectly relates to, arises or results from, or occurs in connection with facts or circumstances relating to the conduct of business of Edible Garden or the assets of Edible Garden, or events or circumstances existing on or prior to the Closing Date.
- (b) Notwithstanding anything in Section 6.2(a) to the contrary, no Shareholder shall bear any obligation to provide indemnity under Section 6.2(a) unless, and only to the extent, such obligation arises from the knowing misrepresentation or act, or failure to act, of that particular Shareholder; for clarity, no Shareholder shall be liable for indemnity under Section 6.2(a) with respect to (i) any misrepresentation, act or omission to act or breach by any other Shareholder, or (ii) any misrepresentation by Edible Garden unless the same is known to said Shareholder.
- (c) Terra Tech shall defend, indemnify and hold harmless Edible Garden and the Shareholders and their respective affiliates, agents, successors and assigns (collectively, the "Edible Garden Indemnified Persons"), and shall reimburse the Edible Garden Indemnified Persons, for, from and against any Damages, directly or indirectly, relating to, resulting from or arising out of:
 - (i) any untrue representation, misrepresentation or breach of warranty by or of Terra Tech contained in or pursuant to this Agreement;
- (ii) any breach or nonfulfillment of any covenant, agreement or other obligations by or of Terra Tech contained in or pursuant to this Agreement, the Transaction Agreements or any other agreements, documents, schedules or exhibits to be entered into or delivered to pursuant to or in connection with this Agreement.
- (d) Promptly after receipt by an indemnified Party under Section 6.2 of this Agreement of notice of a claim against it ("Claim"), such indemnified Party shall, if a claim is to be made against an indemnifying Party under such Section, give notice to the indemnifying Party of such Claim, but the failure to so notify the indemnifying Party will not relieve the indemnifying Party of any liability that it may have to any indemnified Party, except to the extent that the indemnifying Party demonstrates that the defense of such action is prejudiced by the indemnified Party's failure to give such notice.

(e) A claim for indemnification for any matter not involving a third-party claim may be asserted by notice to the Party from whom indemnification is sought.

6.3 Publicity.

Except as agreed by Terra Tech and Edible Garden, no party hereto, nor any of their Representatives, shall issue or disseminate any press release or other publicity or otherwise makes any disclosure of any nature (to any of its suppliers, customers, landlords, creditors or employees or to any other Person) regarding the Exchange; except in each case to the extent that a Party is required by law to make any such disclosure regarding such transactions or as separately agreed by the parties; provided, however, that if it is required by law to make any such disclosure, the Party proposing to make such disclosure shall advise the other Party as early as possible before making such disclosure, of the nature and content of the intended disclosure.

ARTICLE VII. CLOSING CONDITIONS OF TERRA TECH

Terra Tech's obligations to affect the Closing and consummate the Exchange are subject to the satisfaction of each of the following conditions:

7.1 Accuracy of Representations and Warranties.

The representations and warranties of Edible Garden and the Shareholders in this Agreement shall have been true and correct as of the date of this Agreement and shall be true and correct on and as of the Closing. Edible Garden and the Shareholders shall have performed all obligations in this Agreement required to be performed or observed by them on or prior to the Closing.

7.2 Additional Conditions to Closing.

(a) All necessary approvals under federal and state securities laws and other authorizations relating to the issuance of the Acquisition Shares shall have been received.

(b) No preliminary or permanent injunction or other order by any federal, state or foreign court of competent jurisdiction which prohibits the consummation of the Exchange shall have been issued and remain in effect. No statute, rule, regulation, executive order, stay, decree, or judgment shall have been enacted, entered, issued, promulgated or enforced by any court or governmental authority which prohibits or restricts the consummation of the Exchange. All authorizations, consents, orders or approvals of, or declarations or filings with, and all expirations of waiting periods imposed by, any Governmental Body which are necessary for the consummation of the Exchange, other than those the failure to obtain which would not materially adversely affect the consummation of the Exchange or in the aggregate have a material adverse effect on Terra Tech and its subsidiaries, taken as a whole, shall have been filed, occurred or been obtained (all such permits, approvals, filings and consents and the lapse of all such waiting periods being referred to as the "Requisite Regulatory Approvals") and all such Requisite Regulatory Approvals shall be in full force and effect.

(c) There shall not be any action taken, or any statute, rule, regulation or order enacted, entered, enforced or deemed applicable to the Exchange, by any Governmental Body which, in connection with the grant of a Requisite Regulatory Approval, imposes any material condition or material restriction upon Terra Tech or its subsidiaries or Edible Garden, including, without limitation, requirements relating to the disposition of assets, which in any such case would so materially adversely impact the economic or business benefits of the Exchange as to render inadvisable the consummation of the Exchange.

7.3 Performance of Agreements.

Edible Garden or the Shareholders, as the case may be, shall have executed and delivered each of the agreements, instruments and documents required to be executed and delivered, and performed all actions required to be performed by Edible Garden or any of the Shareholders, as the case may be, pursuant to this Agreement, except as Terra Tech has otherwise consented in writing.

7.4 Consents.

Each of the Consents identified or required to have been identified in the Edible Garden Disclosure Schedule shall have been obtained and shall be in full force and effect, other than those Consents, which have been expressly waived by Terra Tech.

7.5 No Material Adverse Change and Satisfactory Due Diligence.

There shall not have been any material adverse change in the business, condition, assets, liabilities, operations or financial performance of Edible Garden since the date of this Agreement as determined by Terra Tech in its discretion. Terra Tech shall be satisfied in all respects with the results of its due diligence review of Edible Garden.

7.6 Edible Garden Closing Certificate.

In addition to the documents required to be received under this Agreement, Terra Tech shall also have received the following documents:

- (a) copies of resolutions of Edible Garden, certified by a Secretary, Assistant Secretary or other appropriate officer of Edible Garden, authorizing the execution, delivery and performance of this Agreement and other Transactional Agreements;
 - (b) good standing certificate with respect to Edible Garden issued by the State of Nevada; and

(c) such other documents as Terra Tech may request in good faith for the purpose of(i) evidencing the accuracy of any representation or warranty made by Edible Garden, (ii) evidencing the compliance by Edible Garden, or the performance by Edible Garden of, any covenant or obligation set forth in this Agreement or any of the other Transactional Agreements, (iii) evidencing the satisfaction of any condition set forth in Article VIII or this Article VII, or (iv) otherwise facilitating the consummation or performance of the Exchange.

7.7 Transactional Agreements.

Each Person (other than Terra Tech) shall have executed and delivered prior to or on the Closing Date all Transactional Agreements to which it is to be a party.

7.8 Delivery of Stock Certificates, Minute Book and Corporate Seal.

The Shareholders shall have delivered to Terra Tech the stock books, stock ledgers, minute books and corporate seals of Edible Garden.

ARTICLE VIII. CLOSING CONDITIONS OF THE SHAREHOLDERS

The Shareholders' obligations to affect the Closing and consummate the Exchange are subject to the satisfaction of each of the following conditions:

8.1 Accuracy of Representations and Warranties.

The representations and warranties of Terra Tech in this Agreement shall have been true and correct as of the date of this Agreement and shall be true and correct on and as of the Closing and Terra Tech shall have performed all obligations in this Agreement required to be performed or observed by them on or prior to the Closing.

8.2 Additional Conditions to Closing.

- (a) All necessary approvals under federal and state securities laws and other authorizations relating to the issuance and transfer of the Acquisition Shares by Terra Tech and the transfer of the Shares by the Shareholders shall have been received.
- (b) No preliminary or permanent injunction or other order by any federal, state or foreign court of competent jurisdiction which prohibits the consummation of the Exchange shall have been issued and remain in effect. No statute, rule, regulation, executive order, stay, decree, or judgment shall have been enacted, entered, issued, promulgated or enforced by any court or governmental authority which prohibits or restricts the consummation of the Exchange. All Requisite Regulatory Approvals shall have been filed, occurred or been obtained and all such Requisite Regulatory Approvals shall be in full force and effect.
- (c) There shall not be any action taken, or any statute, rule, regulation or order enacted, entered, enforced or deemed applicable to the Exchange, by any federal or state Governmental Body which, in connection with the grant of a Requisite Regulatory Approval, imposes any condition or restriction upon Terra Tech or its subsidiaries or Edible Garden, including, without limitation, requirements relating to the disposition of assets, which in any such case would so materially adversely impact the economic or business benefits of the Exchange as to render inadvisable the consummation of the Exchange.

- (d) Terra Tech shall have entered into an equity line financing of an amount not less than \$4,000,000, and a condition of such financing shall be that the securities being sold pursuant to such financing shall be included in an offering registered with the Securities and Exchange Commission under a Registration Statement on From S-1.
 - (e) Michael Vande Vrede and Steve VandeVrede shall have appointed directors of Terra Tech.

8.3 Terra Tech Closing Certificates.

The Shareholders shall have received the following documents:

- (a) copies of resolutions of Terra Tech, certified by a Secretary, Assistant Secretary or other appropriate officer of Terra Tech, authorizing the execution, delivery and performance of this Agreement, the Transactional Agreements and the Exchange;
 - (b) good standing certificate with respect to Terra Tech issued by the State of Nevada; and
- (c) such other documents as Edible Garden may request in good faith for the purpose of (i) evidencing the accuracy of any representation or warranty made by Terra Tech, (ii) evidencing the compliance by Terra Tech with, or the performance by Terra Tech of, any covenant or obligation set forth in this Agreement or any of the other Transactional Agreements, (iii) evidencing the satisfaction of any condition set forth in Article VI or this Article VII, or (iv) otherwise facilitating the consummation or performance of the Exchange.

8.4 No Material Adverse Change.

There shall not have been any material adverse change in Terra Tech's business, condition, assets, liabilities, operations or financial performance since the date of this Agreement.

8.5 Performance of Agreements.

Terra Tech shall have executed and delivered each of the agreements, instruments and documents required to be executed and delivered, and performed all actions required by Terra Tech pursuant to this Agreement, except as Edible Garden and the Shareholders have otherwise consented in writing.

8.6 Consents.

Each of the Consents identified or required to have been identified in Section 3.4 shall have been obtained and shall be in full force and effect, other than those Consents the absence of which shall not have a material adverse effect on Terra Tech.

8.7 Terra Tech Stock.

On the Closing Date, shares of Terra Tech Common Stock shall be eligible for quotation on the Over-the-Counter Bulletin Board.

ARTICLE IX. FURTHER ASSURANCES

Each of the parties hereto agrees that it will, from time to time after the date of the Agreement, execute and deliver such other certificates, documents and instruments and take such other action as may be reasonably requested by the other party to carry out the actions and transactions contemplated by this Agreement, including the closing conditions described in Articles VII and VIII. Edible Garden and the Shareholders shall reasonably cooperate with Terra Tech in its obtaining of the books and records of Edible Garden, or in preparing any solicitation materials to be sent to the shareholders of Terra Tech in connection with the approval of the Exchange and the transactions contemplated by the Transactional Agreements.

ARTICLE X. TERMINATION

10.1 Termination.

This Agreement may be terminated and the Exchange abandoned at any time prior to the Closing Date:

- (a) by mutual written consent of Terra Tech, Edible Garden and the Shareholders;
- (b) by Terra Tech if (i) there is a material Breach of any covenant or obligation of Edible Garden or the Shareholders; provided however, that if such Breach or Breaches are capable of being cured prior to the Closing Date, such Breach or Breaches shall not have been cured within 10 days of delivery of the written notice of such Breach, or (ii) Terra Tech reasonably determines that the timely satisfaction of any condition set forth in Article VII has become impossible or impractical (other than as a result of any failure on the part of Terra Tech to comply with or perform its covenants and obligations under this Agreement or any of the other Transactional Agreements);
- (c) by Edible Garden if (i) there is a material Breach of any covenant or obligation of Terra Tech; provided, however, that if such Breach or Breaches are capable of being cured prior to the Closing Date, such Breach or Breaches shall not have been cured within 10 days of delivery of the written notice of such Breach, or (ii) Edible Garden reasonably determines that the timely satisfaction of any condition set forth in Article VIII has become impossible or impractical (other than as a result of any failure on the part of Edible Garden or any Shareholder to comply with or perform any covenant or obligation set forth in this Agreement or any of the other Transactional Agreements);

- (d) by Terra Tech if the Closing has not taken place on or before May 1, 2013 (except if as a result of any failure on the part of Terra Tech to comply with or perform its covenants and obligations under this Agreement or in any other Transactional Agreement);
- (e) by Edible Garden if the Closing has not taken place on or before May 1, 2013 (except if as a result of the failure on the part of Edible Garden or the Shareholders to comply with or perform any covenant or obligation set forth in this Agreement or in any other Transactional Agreement);
- (f) by any of Terra Tech, on the one hand or Edible Garden, on the other hand, if any court of competent jurisdiction in the United States or other United States governmental body shall have issued an order, decree or ruling or taken any other action restraining, enjoining or otherwise prohibiting the Exchange and such order, decree, ruling or any other action shall have become final and non-appealable; provided, however, that the party seeking to terminate this Agreement pursuant to this clause (f) shall have used all commercially reasonable efforts to remove such order, decree or ruling; or
- (g) The parties hereby agree and acknowledge that a breach of the provisions of Articles 4.1, 4.2, 4.3 and 4.4 are, without limitation, material Breaches of this Agreement.

10.2 Termination Procedures.

If Terra Tech wishes to terminate this Agreement pursuant to Section 10.1, Terra Tech shall deliver to the Shareholders and Edible Garden a written notice stating that Terra Tech is terminating this Agreement and setting forth a brief description of the basis on which Terra Tech is terminating this Agreement. If Edible Garden or the Shareholders wish to terminate this Agreement pursuant to Section 10.1, Edible Garden or the Shareholders shall deliver to Terra Tech a written notice stating that Edible Garden and the Shareholders are terminating this Agreement and setting forth a brief description of the basis on which Edible Garden and the Shareholders are terminating this Agreement.

10.3 Effect of Termination.

In the event of termination of this Agreement as provided above, this Agreement shall forthwith have no further effect. Except for a termination resulting from a Breach by a party to this Agreement, there shall be no liability or obligation on the part of any party hereto. In the event of a breach, the remedies of the non-breaching party shall be to seek damages from the breaching party or to obtain an order for specific performance, in addition to or in lieu of other remedies provided herein. Upon request after termination, each party will redeliver or, at the option of the party receiving such request, destroy all reports, work papers and other material of any other party relating to the Exchange, whether obtained before or after the execution hereof, to the party furnishing same; provided, however, that each Party shall, in all events, remain bound by and continue to be subject to Section 6.1, 6.2 and 6.3.

Notwithstanding the above, both Terra Tech, on the one hand, and Edible Garden and the Shareholders, on the other hand, shall be entitled to announce the termination of this Agreement by means of a mutually acceptable press release.

ARTICLE XI. MISCELLANEOUS

11.1 Survival of Representations and Warranties.

All representations and warranties of Terra Tech, Edible Garden and the Shareholders in this Agreement and the Edible Garden Disclosure Schedule shall survive indefinitely. The right to indemnification, reimbursement or other remedy based on such representations and warranties will not be affected by any investigation conducted by the parties.

11.2 Expenses

Except as otherwise set forth herein, each of the parties to the Exchange shall bear its own expenses incurred in connection with the negotiation and consummation of the transactions contemplated by this Agreement.

11.3 Entire Agreement.

This Agreement and the other Transactional Agreements contain the entire agreement of the parties hereto, and supersede any prior written or oral agreements between them concerning the subject matter contained herein, or therein. There are no representations, agreements, arrangements or understandings, oral or written, between the parties to this Agreement, relating to the subject matter contained in this Agreement and the other Transaction Agreements, which are not fully expressed herein or therein. The schedules and each exhibit attached to this Agreement or delivered pursuant to this Agreement are incorporated herein by this reference and constitute a part of this Agreement.

11.4 Counterparts.

This Agreement may be executed in any number of counterparts, each of which shall be deemed an original but all of which shall constitute one and the same instrument.

11.5 Descriptive Headings.

The Article and Section headings in this Agreement are for convenience only and shall not affect the meanings or construction of any provision of this Agreement.

11.6 Notices.

Any notices required or permitted to be given under this Agreement shall be in writing and shall be deemed sufficiently given on the earlier to occur of the date of personal delivery, the date of receipt or three (3) days after posting by overnight courier or registered or certified mail, postage prepaid, addressed as follows:

If to Terra Tech:

Terra Tech Corp. 112 North Curry Street Carson City, Nevada 89703

If to Edible Garden or the Shareholders:

Edible Garden Inc. 30 Hillview Road Lincoln Park, New Jersey 07035

With a copy to:

Stuart Brown, Esq. Brown, Moskowitz & Kallen, P.C. 75 Main Street, Suite 203 Milburn, New Jersey 07041

To such address or addresses as a party shall have previously designated by notice to the sender given in accordance with this section.

11.7 Choice of Law.

This Agreement shall be construed in accordance with and governed by the laws of the State of Nevada without regard to choice of law principles. Each of the parties hereto consents to the jurisdiction of the courts of the State of California, County of Orange and to the federal courts located in the County of Orange, State of California.

11.8 Binding Effect; Benefits.

This Agreement shall inure to the benefit of and be binding upon the parties and their respective successors and permitted assigns. Nothing in this Agreement, express or implied, is intended to confer on any Person other than the parties or their respective successors and permitted assigns, the Shareholders and other Persons expressly referred to herein, any rights, remedies, obligations or liabilities under or by reason of this Agreement.

11.9 Assignability.

Neither this Agreement nor any of the parties' rights hereunder shall be assignable by any party without the prior written consent of the other parties and any attempted assignment without such consent shall be void.

11.10 Waiver and Amendment.

Any term or provision of this Agreement may be waived at any time by the party, which is entitled to the benefits thereof. The waiver by any party of a breach of any provision of this Agreement shall not operate or be construed as a waiver of any subsequent breach. The parties may, by mutual agreement in writing, amend this Agreement in any respect. Edible Garden and the Shareholders hereby acknowledge their intent that this Agreement includes as a party any holder of capital stock in Edible Garden at the time of Closing. Terra Tech, Edible Garden and the Shareholders therefore agree that this Agreement may be amended, without the further consent of any party to this Agreement, (i) to add as a new Shareholder any existing shareholder of Edible Garden and (ii) to modify Schedule 1 to reflect the addition of such shareholder.

11.11 Attorney' Fees.

In the event of any action or proceeding to enforce the terms and conditions of this Agreement, the prevailing party shall be entitled to an award of reasonable attorneys' and experts' fees and costs, in addition to such other relief as may be granted.

11.12 Severability.

If any provision of this Agreement is held invalid or unenforceable by any court of competent jurisdiction, the other provisions of this Agreement will remain in full force and effect. Any provision of this Agreement held invalid or unenforceable only in part or degree will remain in full force and effect to the extent not held invalid or unenforceable.

11.13 Construction.

In executing this Agreement, the parties severally acknowledge and represent that each: (a) has fully and carefully read and considered this Agreement; (b) has or has had the opportunity to consult independent legal counsel regarding the legal effect and meaning of this document and all terms and conditions hereof; (c) has been afforded the opportunity to negotiate as to any and all terms hereof; and (d) is executing this Agreement voluntarily, free from any influence, coercion or duress of any kind. The language used in this Agreement will be deemed to be the language chosen by the parties to express their mutual intent, and no rule of strict construction will be applied against any party.

[signature page follows]

IN WITNESS WHEREOF, this Agreement has been executed by the parties hereto as of the day and year first above written.

Terra Tech:

TERRA TECH CORP.

By: /s/Derek Peterson

Name: Derek Peterson

Title: President and Chief Executive Officer

Edible Garden:

EDIBLE GARDEN CORP.

By: /s/Ken Vande Vrede

Name: Ken Vande Vrede

Title: President

Shareholders of Edible Garden Corp.:

/s/ Ken Vande Vrede

Name: Ken Vande Vrede

(holder of 23 shares of common stock of Edible Garden Corp.)

/s/ Mike Vande Vrede

Name: Mike Vande Vrede

(holder of 23 shares of common stock of Edible Garden Corp.)

/s/ Steve Vande Vrede

Name: Steve Vande Vrede

(holder of 23 shares of common stock of Edible Garden Corp.)

/s/ Dan Vande Vrede

Name: Dan Vande Vrede

(holder of 23 shares of common stock of Edible Garden Corp.)

/s/ Beverly Willekes Name: Beverly Willekes

(holder of 4 shares of common stock of Edible Garden Corp.)

/s/ David Vande Vrede

Name: David Vande Vrede

(holder of 4 shares of common stock of Edible Garden Corp.)

EXHIBIT A

CERTAIN DEFINITIONS

For purposes of the Agreement (including this Exhibit A):

Agreement. "Agreement" shall mean the Share Exchange Agreement to which this Exhibit A is attached (including all Disclosure Schedules and all Exhibits), as it may be amended from time to time.

Approved Plans. "Approved Plans" shall mean a stock option or similar plan for the benefit of employees or others, which has been approved by the shareholders of Edible Garden.

Edible Garden Shares of Common Stock. "Edible Garden Shares of Common Stock" shall mean the shares of common stock of Edible Garden.

Breach. There shall be deemed to be a "Breach" of a representation, warranty, covenant, obligation or other provision if there is or has been any inaccuracy in or breach of, or any failure to comply with or perform, such representation, warranty, covenant, obligation or other provision.

Certificates. "Certificates" shall have the meaning specified in Section 1.3 of the Agreement.

Terra Tech. "Terra Tech" shall have the meaning specified in the first paragraph of the Agreement.

Terra Tech Common Stock. "Terra Tech Common Stock" shall mean the shares of common stock of Terra Tech.

Terra Tech SEC Reports. "Terra Tech SEC Reports" shall have the meaning specified in Section 3.6 of the Agreement.

Closing. "Closing" shall have the meaning specified in Section 1.5 of the Agreement.

Closing Date. "Closing Date" shall have the meaning specified in Section 1.5 of the Agreement.

Code. "Code" shall mean the Internal Revenue Code of 1986 or any successor law, and regulations issued by the IRS pursuant to the Internal Revenue Code or any successor law.

Confidential Information. "Confidential Information" shall mean all nonpublic information disclosed by one party or its agents (the "Disclosing Party") to the other party or its agents (the "Receiving Party") that is designated as confidential or that, given the nature of the information or the circumstances surrounding its disclosure, reasonably should be considered as confidential. Confidential Information includes, without limitation (i) nonpublic information relating to the Disclosing Party's technology, customers, vendors, suppliers, business plans, intellectual property, promotional and marketing activities, finances, agreements, transactions, financial information and other business affairs, and (ii) third-party information that the Disclosing Party is obligated to keep confidential.

Confidential Information does not include any information that (i) is or becomes publicly available without breach of this Agreement, (ii) can be shown by documentation to have been known to the Receiving Party at the time of its receipt from the Disclosing Party, (iii) is received from a third party who, to the knowledge of the Receiving Party, did not acquire or disclose such information by a wrongful or tortious act, or (iv) can be shown by documentation to have been independently developed by the Receiving Party without reference to any Confidential Information.

Consent. "Consent" shall mean any approval, consent, ratification, permission, waiver or authorization (including any Governmental Authorization).

Disclosure Schedule Update. "Disclosure Schedule Update" shall have the meaning specified in Section 4.4 of the Agreement.

Edible Garden Disclosure Schedule. "Edible Garden Disclosure Schedule" shall have the meaning specified in introduction to Article II of the Agreement.

Entity. "Entity" shall mean any corporation (including any non-profit corporation), general partnership, limited partnership, limited liability partnership, joint venture, estate, trust, cooperative, foundation, society, political party, union, company (including any limited liability company or joint stock company), firm or other enterprise, association, organization or entity.

Environmental Laws. "Environmental Laws" shall mean any Law or other requirement relating to the protection of the environment, health, or safety from the release or disposal of hazardous materials.

Environmental Permit. "Environmental Permit" means all licenses, permits, authorizations, approvals, franchises and rights required under any applicable Environmental Law or Order.

Equity Securities. "Equity Security" shall mean any stock or similar security, including, without limitation, securities containing equity features and securities containing profit participation features, or any security convertible into or exchangeable for, with or without consideration, any stock or similar security, or any security carrying any warrant, right or option to subscribe to or purchase any shares of capital stock, or any such warrant or right.

Exchange Act. "Exchange Act" means the United States Securities Exchange Act of 1934, as amended.

GAAP. "GAAP" shall mean United States Generally Accepted Accounting Principles, applied on a consistent basis.

Governmental Authorization. "Governmental Authorization" shall mean any:

- (a) permit, license, certificate, franchise, concession, approval, consent, ratification, permission, clearance, confirmation, endorsement, waiver, certification, designation, rating, registration, qualification or authorization that is issued, granted, given or otherwise made available by or under the authority of any Governmental Body or pursuant to any Law; or
 - (b) right under any contract with any Governmental Body.

Governmental Body. "Governmental Body" shall mean any:

- (a) nation, principality, state, commonwealth, province, territory, county, municipality, district or other jurisdiction of any nature;
- (b) federal, state, local, municipal, foreign or other government;
- (c) governmental or quasi-governmental authority of any nature (including any governmental division, subdivision, department, agency, bureau, branch, office, commission, council, board, instrumentality, officer, official, representative, organization, unit, body or Entity and any court or other tribunal); or
- (d) individual, Entity or body exercising, or entitled to exercise, any executive, legislative, judicial, administrative, regulatory, police, military or taxing authority or power of any nature, including any court, arbitrator, administrative agency or commissioner, or other governmental authority or instrumentality.

<u>Indebtedness</u>. "Indebtedness" shall mean any obligation, contingent or otherwise. Any obligation secured by a Lien on, or payable out of the proceeds of, or production from, property of the relevant party will be deemed to be Indebtedness.

Intellectual Property. "Intellectual Property" means all industrial and intellectual property, including, without limitation, all U.S. and non-U.S. patents, patent applications, patent rights, trademarks, trademarks, trademarks, trademarks, trademarks, trademarks, service marks, service marks, service marks, service marks, and the goodwill associated therewith, copyrights, in both published and unpublished works, whether registered or unregistered, copyright applications, franchises, licenses, know-how, trade secrets, technical data, designs, customer lists, confidential and proprietary information, processes and formulae, all computer software programs or applications, layouts, inventions, development tools and all documentation and media constituting, describing or relating to the above, including manuals, memoranda, and records, whether such intellectual property has been created, applied for or obtained anywhere throughout the world.

Knowledge. "Knowledge" means, (a) with respect to an individual, the actual knowledge of that individual and (b) with respect to a corporation, the actual knowledge of a director or officer of such corporation; in either event, without any obligation to conduct any investigation.

<u>Laws</u>. "Laws" means, with respect to any Person, any U.S. or non-U.S. federal, national, state, provincial, local, municipal, international, multinational or other law (including common law), constitution, statute, code, ordinance, rule, regulation or treaty applicable to such Person.

Lien. "Lien" shall mean any mortgage, pledge, security interest, encumbrance, lien or charge, right of first refusal, encumbrance or other adverse claim or interest of any kind, including, without limitation, any conditional sale or other title retention agreement, any lease in the nature thereof and the filing of or agreement to give any financing statement under the Uniform Commercial Code of any jurisdiction and including any lien or charge arising by Law.

Material Adverse Effect. "Material Adverse Effect" means any change, effect or circumstance which, individually or in the aggregate, would reasonably be expected to (a) have a material adverse effect on the business, assets, financial condition or results of operations of the affected party, in each case taken as a whole or (b) materially impair the ability of the affected party to perform its obligations under this Agreement and the Transaction Agreements, excluding any change, effect or circumstance resulting from (i) the announcement, pendency or consummation of the transactions contemplated by this Agreement, (ii) changes in the United States securities markets generally, or (iii) changes in general economic, currency exchange rate, political or regulatory conditions in industries in which the affected party operates.

Material Contract: "Material Contract" means any and all agreements, contracts, arrangements, understandings, leases, commitments or otherwise, providing for potential payments by or to the company in excess of \$1,000, and the amendments, supplements and modifications thereto.

Order. "Order" shall mean any award, decision, injunction, judgment, order, ruling, subpoena, or verdict entered, issued, made, or rendered by any Governmental Body.

Ordinary Course of Business. "Ordinary Course of Business" shall mean an action taken by Edible Garden if (i) such action is taken in normal operation, consistent with past practices, (ii) such action is not required to be authorized by the Shareholders, Board of Directors or any committee of the Board of the Directors or other governing body of Edible Garden and (iii) does not require any separate or special authorization or consent of any nature by any Governmental Body or third party.

Permitted Liens. "Permitted Liens" shall mean (a) Liens for Taxes not yet payable or in respect of which the validity thereof is being contested in good faith by appropriate proceedings and for the payment of which the relevant party has made adequate reserves; (b) Liens in respect of pledges or deposits under workmen's compensation laws or similar legislation, carriers, warehousemen, mechanics, laborers and materialmen and similar Liens, if the obligations secured by such Liens are not then delinquent or are being contested in good faith by appropriate proceedings conducted and for the payment of which the relevant party has made adequate reserves; and (c) statutory Liens incidental to the conduct of the business of the relevant party which were not incurred in connection with the borrowing of money or the obtaining of advances or credits and that do not in the aggregate materially detract from the value of its property or materially impair the use thereof in the operation of its business.

Person. "Person" shall mean any individual, Entity or Governmental Body.

Pre-Closing Period. "Pre-Closing Period" shall mean the period commencing as of the date of the Agreement and ending on the Closing Date.

<u>Proceeding</u>. "Proceeding" shall mean any action, suit, litigation, arbitration, proceeding (including any civil, criminal, administrative, investigative or appellate proceeding and any informal proceeding), prosecution, contest, hearing, inquiry, inquest, audit, examination or investigation, commenced, brought, conducted or heard by or before, or otherwise has involved, any Governmental Body or any arbitrator or arbitration panel.

Representatives. "Representatives" of a specified party shall mean officers, directors, employees, attorneys, accountants, advisors and representatives of such party, including, without limitation, all subsidiaries of such specified party, and all such Persons with respect to such subsidiaries. The Related Persons of Edible Garden shall be deemed to be "Representatives" of Edible Garden, as applicable.

SEC. "SEC" shall mean the United States Securities and Exchange Commission.

Securities Act. "Securities Act" shall mean the United States Securities Act of 1933, as amended.

Taxes. "Taxes" shall mean all foreign, federal, state or local taxes, charges, fees, levies, imposts, duties and other assessments, as applicable, including, but not limited to, any income, alternative minimum or add-on, estimated, gross income, gross receipts, sales, use, transfer, transactions, intangibles, ad valorem, value-added, franchise, registration, title, license, capital, paid-up capital, profits, withholding, payroll, employment, unemployment, excise, severance, stamp, occupation, premium, real property, recording, personal property, federal highway use, commercial rent, environmental (including, but not limited to, taxes under Section 59A of the Code) or windfall profit tax, custom, duty or other tax, governmental fee or other like assessment or charge of any kind whatsoever, together with any interest, penalties or additions to tax with respect to any of the foregoing; and "Tax" means any of the foregoing Taxes.

<u>Tax Group</u>. "Tax Group" shall mean any federal, state, local or foreign consolidated, affiliated, combined, unitary or other similar group of which Edible Garden is now or was formerly a member.

Tax Return. "Tax Return" shall mean any return, declaration, report, claim for refund or credit, information return, statement or other similar document filed with any Governmental Body with respect to Taxes, including any schedule or attachment thereto, and including any amendment thereof.

<u>Transaction Agreements</u>. "Transactional Agreements" shall mean this Agreement, the Securities Purchase Agreement between Derek Peterson, Amy Almsteier and the Shareholders, and any agreement or document to be executed pursuant to this Agreement.

SCHEDULE 1 Edible Garden Shareholders

Name	Number of Shares of Common Stock of Edible Garden Held	Percent
Ken Vande Vrede	23	23.0%
Mike Vande Vrede	23	23.0%
Steve Vande Vrede	23	23.0%
Dan Vande Vrede	23	23.0%
Beverly Willekes	4	4.0%
David Vande Vrede	4	4.0%
TOTAL	100	100.0%





Articles of Exchange

(PURSUANT TO NRS 92A.200)

Page 1

USE BLACK INK ONLY - DO NOT HIGHLIGHT

ABOVE SPACE IS FOR OFFICE USE ONLY

Articles of Exchange (Pursuant to NRS Chapter 92A - excluding 92A.200(4b))

poration
•
•
tity type *
poration
tity type *

2) The undersigned declares that a plan of exchange has been adopted by each constituent entity (NRS 92A.200).

FILING FEE: \$350.00

This form must be accompanied by appropriate fees.

Nevada Secretary of State 92A Exchange Page 1 Revised: 8-31-11

^{*} Corporation, non-profit corporation, limited partnership, limited-liability limited partnership, limited-liability company or business trust.



Articles of Exchange

(PURSUANT TO NRS 92A.200)

Page 2

USE BLACK INK ONLY - DO NOT HIGHLIGHT

A BOVE SPACE IS FOR OFFICE USE ONLY

	han two constituent entities the entities continued fron	s, please check box and attach an 8 1 n article three.
)wner's approval was	s not required from	

Name of <i>acquired</i>	entity, if applicable	
and, or;		
Terra Tech Corp.		
Name of <i>acquiring</i>	entity, if applicable	
he plan was approve	ed by the required consent (of the owners of *
Edible Garden Corp.		
Name of <i>acquired</i>	entity, if applicable	
and, or;		

*Unless otherwise provided in the certificate of trust or governing instrument of a business trust, an exchange must be approved by all the trustees and beneficial owners of each business trust that is a constituent entity in the exchange.

This form must be accompanied by appropriate fees.

Nevada Secretary of State 92 A Exchange Page 2 Revised: 8-31-11



Articles of Exchange

(PURSUANT TO NRS 92A.200)

Page 3

USE BLACK INK ONLY - DO NOT HIGHLIGHT

A BOVE SPACE IS FOR OFFICE USE ONLY

	plan of exchange has been approved by the directors of the corporation and by each lic officer or other person whose approval of the plan of exchange is required by the les of incorporation of the domestic corporation.
Nan	ne of acquired entity, if applicable
and	, or,
Nan	ne of acquiring entity, if applicable
tio	n of Plan of Exchange (check a or b):
	(a) The entire plan of exchange is attached;
O.	



Articles of Exchange (PURSUANT TO NRS 92A.200)

Page 4

USE BLACK INK ONLY - DO NOT HIGHLIGHT

ABOVE SPACE IS FOR OFFICE USE ONLY

	Time:	
Nevada limited partnership; All ge manager of each Nevada limited-li A trustee of each Nevada busines	An officer of each Nevada corporation; eneral partners of each Nevada limited-lability company with managers or a me s trust (NRS 92A.230); ** constituent entities, please check box a ies continued from article six.	liability limited partnership; A ember if there are no Manage
Edible Garden Corp.		
Name of acquired entity		,
x	President	04/15/13
Signature	Title	Date
Terra Tech Corp.		
Name of acquiring entity		
X	Chief Executive Officer	04/15/13
Signature	Title	Date

COMMON STOCK PURCHASE AGREEMENT

Dated as of April 26, 2013

by and between

TERRA TECH CORP.,

a Nevada corporation

and

HANOVER HOLDINGS I, LLC, a New York limited liability company

1

TABLE OF CONTENTS

	Page
ARTICLE I DEFINITIONS	6
ARTICLE II PURCHASE AND SALE OF COMMON STOCK	6
Section 2.1. Purchase and Sale of Stock	6
Section 2.2. Closing Date; Settlement Dates	7
Section 2.3. Initial Public Announcements and Required Filings	7
ARTICLE III DRAW DOWN TERMS	8
Section 3.1. Draw Down Notice	8
Section 3.2. Limitation of Draw Downs	8
Section 3.3. Reduction of Commitment	8
Section 3.4. Below Floor Price	9
Section 3.5. Settlement	9
Section 3.6. Failure to Deliver Shares	9
Section 3.7. Certain Limitations	9
Section 3.8. Blackout Periods	10
ARTICLE IV REPRESENTATIONS, WARRANTIES AND COVENANTS OF THE INVESTOR	11
Section 4.1. Organization and Standing of the Investor	11
Section 4.2. Authorization and Power	11
Section 4.3. No Conflicts	11
Section 4.4. Investment Purpose	11
Section 4.5. Accredited Investor Status	12
Section 4.6. Reliance on Exemptions	12
Section 4.7. Information	12
Section 4.8. No Governmental Review	12
Section 4.9. No General Solicitation	12
Section 4.10. Not an Affiliate	12
Section 4.11. Statutory Underwriter Status	13
Section 4.12. Resales of Securities	13
ARTICLE V REPRESENTATIONS, WARRANTIES AND COVENANTS OF THE COMPANY	13
Section 5.1. Organization, Good Standing and Power	13
Section 5.2. Authorization, Enforcement	13
Section 5.3. Capitalization and Voting Rights	14
Section 5.4. Issuance of Securities	14
2	
2	

	Section 5.5. No Conflicts	14
	Section 5.6. Commission Documents, Financial Statements	15
	Section 5.7. Subsidiaries	16
	Section 5.8. No Material Adverse Effect	17
	Section 5.9. No Undisclosed Liabilities	17
	Section 5.10. No Undisclosed Events or Circumstances	17
	Section 5.11. Indebtedness; Solvency	17
	Section 5.12. Title To Assets	18
	Section 5.13. Actions Pending	18
	Section 5.14. Compliance With Law	18
	Section 5.15. Certain Fees	18
	Section 5.16. Disclosure	19
	Section 5.17. Operation of Business	19
	Section 5.18. Environmental Compliance	20
	Section 5.19. Material Agreements	20
	Section 5.20. Transactions With Affiliates	21
	Section 5.21. Employees	21
	Section 5.22. Use of Proceeds	21
	Section 5.23. Investment Company Act Status	21
	Section 5.24. ERISA	21
	Section 5.25. Taxes	22
	Section 5.26. Insurance	22
	Section 5.27. U.S. Real Property Holding Corporation	22
	Section 5.28. Exemption from Registration; Valid Issuances	22
	Section 5.29. No General Solicitation or Advertising	23
	Section 5.30. No Integrated Offering	23
	Section 5.31. Dilutive Effect	23
	Section 5.32. Manipulation of Price	23
	Section 5.33. Securities Act	23
	Section 5.34. Listing and Maintenance Requirements	24
	Section 5.35. Application of Takeover Protections	24
	Section 5.36. Foreign Corrupt Practices Act	24
	Section 5.37. Money Laundering Laws	24
	Section 5.38. OFAC	25
	Section 5.39. [Intentionally Omitted]	25
	Section 5.40. Royalties and Commissions	25
	Section 5.41. Acknowledgement Regarding Investor's Acquisition of Securities	25
	·	
ART	TICLE VI ADDITIONAL COVENANTS	25
	Section 6.1. Securities Compliance	25
	Section 6.2. Reservation of Common Stock	26
	Section 6.3. Registration and Listing	26
	Section 6.4. Compliance with Laws	26
	Section 6.5. Keening of Records and Books of Account: Due Diligence	27

Section 6.6. Limitations on Holdings and Issuances	
Section 6.7. Other Agreements and Alternate Transactions	
Section 6.8. Corporate Existence	30
Section 6.9. Fundamental Transaction	30
Section 6.10. Delivery of Registration Statement and Prospectus; Subsequent Changes	30
Section 6.11. Amendments to the Registration Statement; Prospectus Supplements	31
Section 6.12. Stop Orders	31
Section 6.13. Selling Restrictions	32
Section 6.14. Effective Registration Statement	32
Section 6.15. Blue Sky	32
Section 6.16. Non-Public Information	33
Section 6.17. Broker/Dealer	33
Section 6.18. Disclosure Schedule	33
ARTICLE VII CONDITIONS TO CLOSING AND CONDITIONS TO THE SALE AND PURCHASE OF THE SHARES	34
Section 7.1. Conditions Precedent to Closing	34
Section 7.2. Conditions Precedent to a Draw Down	35
ARTICLE VIII TERMINATION	
Section 8.1. Termination	38
Section 8.2. Other Termination	38
Section 8.3. Effect of Termination	39
ARTICLE IX INDEMNIFICATION	40
Section 9.1. Indemnification of Investor	40
Section 9.2. Indemnification Procedures	41
$\it \Delta$	

ARTICLE X MISCELLANEOUS	42
Section 10.1. Fees and Expenses	42
Section 10.2. Specific Enforcement, Consent to Jurisdiction, Waiver of Jury Trial	44
Section 10.3. Entire Agreement; Amendment	45
Section 10.4. Notices	45
Section 10.5. Waivers	46
Section 10.6. Headings	46
Section 10.7. Construction	46
Section 10.8. Successors and Assigns	47
Section 10.9. No Third Party Beneficiaries	47
Section 10.10. Governing Law	47
Section 10.11. Survival	47
Section 10.12. Counterparts	47
Section 10.13. Publicity	47
Section 10.14. Severability	48
Section 10.15. Further Assurances	48
Annex I. Definitions	50

COMMON STOCK PURCHASE AGREEMENT

This **COMMON STOCK PURCHASE AGREEMENT** is made and entered into as of March xx, 2013 (this "Agreement"), by and between Hanover Holdings I, LLC, a New York limited liability company (the "Investor"), and Terra Tech Corp., a corporation organized and existing under the laws of the state of Nevada (the "Company").

RECITALS

WHEREAS, the parties desire that, upon the terms and subject to the conditions and limitations set forth herein, the Company may issue and sell to the Investor, from time to time as provided herein, and the Investor shall purchase from the Company, up to \$5,000,000 of newly issued shares of the Company's common stock, \$0.001 par value ("Common Stock");

WHEREAS, such investments will be made in reliance upon the provisions of Section 4(a)(2) of the Securities Act ('Section 4(a)(2)'') and Rule 506 of Regulation D promulgated by the Commission under the Securities Act (''Regulation D''), and upon such other exemption from the registration requirements of the Securities Act as may be available with respect to any or all of the investments in Common Stock to be made hereunder;

WHEREAS, the parties hereto are concurrently entering into a Registration Rights Agreement in the form of Exhibit A hereto (the "Registration Rights Agreement"), pursuant to which the Company shall register the Registrable Securities (as defined in the Registration Rights Agreement), upon the terms and subject to the conditions set forth therein; and

WHEREAS, in consideration for the Investor's execution and delivery of this Agreement, the Company is concurrently causing its transfer agent to issue to the Investor the Initial Commitment Shares, upon the terms and subject to the conditions set forth in this Agreement;

NOW, THEREFORE, the parties hereto, intending to be legally bound, hereby agree as follows:

ARTICLE I DEFINITIONS

Capitalized terms used in this Agreement shall have the meanings ascribed to such terms in <u>Annex I</u> hereto, and hereby made a part hereof, or as otherwise set forth in this Agreement.

ARTICLE II PURCHASE AND SALE OF COMMON STOCK

Section 2.1. Purchase and Sale of Stock. Upon the terms and subject to the conditions of this Agreement, during the Investment Period, the Company in its discretion may issue and sell to the Investor, and the Investor shall purchase from the Company, up to \$5,000,000 (the "Total Commitment") of duly authorized, validly issued, fully paid and nonassessable shares of Common Stock (the "Aggregate Limit"), by the delivery to the Investor of Draw Down Notices as provided in Article III hereof.

Section 2.2. Closing Date; Settlement Dates. This Agreement shall become effective and binding (the "Closing") upon payment of the Document Preparation Fee prior to the Closing pursuant to Sections 7.1 and 10.1, the delivery of irrevocable instructions to issue the Initial Commitment Shares to the Investor or its designees as provided in Sections 7.1 and 10.1, the delivery of counterpart signature pages of this Agreement and the Registration Rights Agreement executed by each of the parties hereto and thereto, and the delivery of all other documents, instruments and writings required to be delivered at the Closing, in each case as provided in Section 7.1, to the offices of Sichenzia Ross Friedman Ference LLP, 61 Broadway, New York, NY 10006, at 5:00 p.m., New York City time, on the Closing Date. In consideration of and in express reliance upon the representations, warranties and covenants contained in, and upon the terms and subject to the conditions of, this Agreement, during the Investment Period the Company shall issue and sell to the Investor, and the Investor shall purchase from the Company, the Shares in respect of each Draw Down. The issuance and sale of Shares to the Investor pursuant to any Draw Down shall occur on the applicable Settlement Date in accordance with Section 3.5, provided that all of the conditions precedent thereto set forth in Article VII theretofore shall have been fulfilled on or prior to such Settlement Date.

Section 2.3. <u>Initial Public Announcements and Required Filings</u>. The Company shall, at or before 8:30 a.m., New York City time, on the second Trading Day after the Closing, issue a press release (the "Press Release") reasonably acceptable to the Investor disclosing the execution of this Agreement and the Registration Rights Agreement by the Company and the Investor and the issuance of the Initial Commitment Shares to the Investor, and briefly describing the transactions contemplated thereby. At or before 8:30 a.m., New York City time, on the second Trading Day following the Closing Date, the Company shall file a Current Report on Form 8-K describing all the material terms of the transactions contemplated by the Transaction Documents in the form required by the Exchange Act and attaching copies of each of this Agreement, the Registration Rights Agreement and the Press Release as exhibits thereto (including all exhibits thereto, the "Current Report"). The Company shall provide the Investor a reasonable opportunity to comment on a draft of the Current Report prior to filing the Current Report with the Commission, shall give due consideration to all such comments and shall not file the Current Report to the extent the Investor reasonably objects to the form or content thereof (provided, however, that the failure of the Investor to make such objection shall not relieve the Company of any obligation or liability under this Agreement or affect the Investor's right to rely on the representations and warranties made by the Company in this Agreement). From and after the issuance of the Press Release and the filing of the Current Report, the Company shall have disclosed all material, nonpublic information delivered to the Investor (or the Investor's representatives or agents) by the Company or any of its Subsidiaries, or any of their respective officers, directors, employees, agents or representatives (if any) in connection with the transactions contemplated by the Transaction Documents. The Investor covenants that until such time as the transactions contemplated by this Agreement are publicly disclosed by the Company as described in this Section 2.3, the Investor will maintain the confidentiality of all disclosures made to it in connection with the transactions contemplated by the Transaction Documents (including the existence and terms of the transactions), except that the Investor may disclose the terms of such transactions to its financial, accounting, legal and other advisors (provided that the Investor directs such Persons to maintain the confidentiality of such information). Not later than 15 calendar days following the Closing Date, the Company shall file a Form D with respect to the Securities in accordance with Regulation D and shall provide a copy thereof to the Investor promptly after such filing. The Company shall prepare and file with the Commission the Registration Statement (including the Prospectus) covering only the resale by the Investor of the Registrable Securities in accordance with the Securities Act and the Registration Rights Agreement. At or before 8:30 a.m. (New York City time) on the Trading Day immediately following the Effective Date, the Company shall file with the Commission in accordance with Rule 424(b) under the Securities Act the final Prospectus to be used in connection with sales pursuant to the Registration Statement. If the transactions contemplated by any Draw Down are material to the Company (individually or collectively with all other prior Draw Downs, the consummation of which have not previously been reported in any Prospectus Supplement filed with the Commission under Rule 424(b) under the Securities Act or in any report, statement or other document filed by the Company with the Commission under the Exchange Act), or if otherwise required under the Securities Act (or the interpretations of the Commission thereof), in each case as reasonably determined by the Company or the Investor, then, on the first Trading Day immediately following the last Trading Day of the Pricing Period with respect to such Draw Down, the Company shall file with the Commission a Prospectus Supplement pursuant to Rule 424(b) under the Securities Act with respect to the applicable Draw Down(s), disclosing the total Draw Down Amount Requested pursuant to such Draw Down(s), the total number of Shares that are to be (and, if applicable, have been) issued and sold to the Investor pursuant to such Draw Down(s), the total purchase price for the Shares subject to such Draw Down(s), the applicable Discount Price(s) for such Shares and the net proceeds that are to be (and, if applicable, have been) received by the Company from the sale of such Shares. To the extent not previously disclosed in the Prospectus or a Prospectus Supplement, the Company shall disclose in its Quarterly Reports on Form 10-Q and in its Annual Reports on Form 10-K the information described in the immediately preceding sentence relating to all Draw Down(s) consummated during the relevant fiscal quarter, and include each such Quarterly Report on Form 10-Q and Annual Report on Form 10-K in a Prospectus Supplement and file such Prospectus Supplement with the Commission under Rule 424(b) under the Securities Act.

ARTICLE III DRAW DOWN TERMS

Subject to the satisfaction of the conditions set forth in this Agreement, the parties agree as follows:

- Section 3.1. Draw Down Notice. From time to time during the Investment Period, the Company may, in its sole discretion, no later than 9:30 a.m. (New York City time) on the first Trading Day of the Pricing Period, provide to the Investor a Draw Down Notice, substantially in the form attached hereto as Exhibit B (the "Draw Down Notice"), which Draw Down Notice shall become effective at 9:30 a.m. (New York City time) on the first Trading Day of the Pricing Period specified in the Draw Down Notice; provided, however, that if the Company delivers the Draw Down Notice to the Investor later than 9:30 a.m. (New York City time) on a Trading Day, then the first Trading Day of such Pricing Period shall not be the Trading Day on which the Investor received such Draw Down Notice, but rather shall be the immediately following Trading Day (unless a subsequent Trading Day is therein specified). The date on which the Company delivers any Draw Down Notice in accordance with this Section 3.1 hereinafter shall be referred to as a "Draw Down Exercise Date". The Draw Down Notice shall specify the Draw Down Amount Requested (which shall not exceed the Maximum Draw Down Amount Requested), establish the Floor Price for such Draw Down and designate the first and last Trading Day of the Pricing Period. Upon the terms and subject to the conditions of this Agreement, the Investor is obligated to accept each Draw Down Notice prepared and delivered in accordance with the provisions of this Agreement and shall purchase from the Company the Shares subject to such Draw Down Notice at the applicable Discount Price on the applicable Settlement Date. Anything to the contrary in this Agreement notwithstanding, the parties hereto acknowledge and agree that the Investor shall not be required to purchase, and shall not purchase, more than the Maximum Draw Down Amount Requested pursuant to any single Draw Down Notice.
- Section 3.2. <u>Limitation of Draw Downs</u>. The Company shall not make more than one Draw Down in each Pricing Period. At least 24 hours shall elapse between the completion of the settlement of one Draw Down in accordance with Section 3.5 below and the commencement of a Pricing Period for any other Draw Down during the Investment Period. Each Draw Down automatically shall expire immediately following the completion of the settlement thereof in accordance with Section 3.5 below.
- Section 3.3. Reduction of Commitment. On each Settlement Date, the Investor's Total Commitment under this Agreement automatically shall be reduced, on a dollar-for-dollar basis, by the total Draw Down Amount paid to the Company on such Settlement Date.

- Section 3.4. Below Floor Price. If the VWAP on any Trading Day during the applicable Pricing Period is lower than the applicable Floor Price, then for each such Trading Day, the Draw Down Amount Requested shall be reduced by a number of shares of Common Stock equal to the product of (x) 0.10 and (y) the total Draw Down Amount Requested, and no Shares shall be purchased or sold with respect to such Trading Day. If trading in the Common Stock on the Trading Market is suspended for any reason for more than three hours on any Trading Day during the applicable Pricing Period, then for each such Trading Day during the Pricing Period the Draw Down Amount Requested shall be reduced as provided in the immediately preceding sentence, and no Shares shall be purchased or sold with respect to such Trading Day.
- Section 3.5. Settlement. The payment for, against simultaneous delivery of, Shares in respect of each Draw Down shall be settled not later than the third Business Day next following the last Trading Day of each Pricing Period (the "Settlement Date"). On each Settlement Date, the Company shall, or shall cause its transfer agent to, electronically transfer the Shares purchased by the Investor by crediting the Investor's or its designees' account (provided the Investor shall have given the Company written notice of such designee prior to the Settlement Date) at DTC through its Deposit/Withdrawal at Custodian (DWAC) system, which Shares shall be freely tradable and transferable and without restriction on resale pursuant to the Registration Statement, against simultaneous payment therefor to the Company's designated account by wire transfer of immediately available funds; provided that if the Shares are received by the Investor later than 1:00 p.m., New York City time, payment therefor shall be made with next day funds. As set forth in Section 3.6, a failure by the Company or its transfer agent (if applicable) to deliver such Shares on the applicable Settlement Date shall result in the payment of partial damages by the Company to the Investor.
- Section 3.6. Failure to Deliver Shares. If the Company issues a Draw Down Notice and fails to deliver the Shares to the Investor on the applicable Settlement Date and such failure continues for 10 Trading Days, the Company shall pay the Investor, in cash, in addition to all other remedies available to the Investor, as partial damages for such failure and not as a penalty, an amount equal to 2.0% of the payment required to be paid by the Investor on such Settlement Date for the initial 30 days following such Settlement Date until the Shares have been delivered, and an additional 2.0% for each additional 30-day period thereafter until the Shares have been delivered, which amount shall be prorated for such periods less than 30 days (the "Make Whole Amount"). If the Make Whole Amount is not paid within two Trading Days following a demand therefor from the Investor, the Make Whole Amount shall accrue annual interest (on the basis of the 365 day year) compounded daily at a rate equal to the greater of (i) the prime rate of interest then in effect as published by the Wall Street Journal plus 3.0% and (ii) 10.0%, up to and including the date on which the Make Whole Amount is actually paid. The Company shall not issue a Draw Down Notice to the Investor until the Make Whole Amount, plus all accrued interest, has been paid to the Investor in full.
- Section 3.7. Certain Limitations. Notwithstanding anything to the contrary contained in this Agreement, in no event may the Company issue a Draw Down Notice to the extent that (i) the Draw Down Amount Requested in such Draw Down Notice exceeds the Maximum Draw Down Amount Requested, (ii) the sale of Shares pursuant to such Draw Down Notice would cause the Company to issue or sell or the Investor to acquire or purchase a dollar value of shares of Common Stock which, when aggregated with all Draw Down Amounts paid by the Investor pursuant to all prior Draw Down Notices issued under this Agreement, would exceed the Aggregate Limit, or (iii) the sale of Shares pursuant to such Draw Down Notice would cause the Company to sell or the Investor to purchase a number of shares of Common Stock which, when aggregated with all other shares of Common Stock then beneficially owned (as calculated pursuant to Section 13(d) of the Exchange Act and Rule 13d-3 promulgated thereunder) by the Investor and its Affiliates, would result in the beneficial ownership by the Investor or any of its Affiliates of more than 4.99% of the then issued and outstanding shares of Common Stock (the "Ownership Limitation"). If the Company issues a Draw Down Notice in which the Draw Down Amount Requested exceeds the Maximum Draw Down Amount Requested, such Draw Down Notice shall be void ab initio to the extent the Draw Down Amount Requested exceeds the Maximum Draw Down Amount Requested. If the Company issues a Draw Down Notice that otherwise would require the Investor to purchase shares of Common Stock which would cause the aggregate purchases of Common Stock by the Investor under this Agreement to exceed the Aggregate Limit, such Draw Down Notice shall be void ab initio to the extent of the amount by which the dollar value of shares of Common Stock otherwise issuable pursuant to such Draw Down Notice, together with all Draw Down Amounts paid by the Investor pursuant to all prior Draw Down Notices issued under this Agreement, would exceed the Aggregate Limit. If the Company issues a Draw Down Notice that otherwise would require the Investor to purchase shares of Common Stock which would cause the aggregate number of shares of Common Stock then beneficially owned (as calculated pursuant to Section 13(d) of the Exchange Act and Rule 13d-3 promulgated thereunder) by the Investor and its Affiliates to exceed the Ownership Limitation, such Draw Down Notice shall be void ab initio to the extent of the amount by which the number of shares of Common Stock otherwise issuable pursuant to such Draw Down Notice, together with all shares of Common Stock then beneficially owned by the Investor and its Affiliates, would exceed the Ownership Limitation.

Section 3.8. Blackout Periods. The Company shall advise the Investor in writing of any changes to its policy on insider trading. Notwithstanding any other provision of this Agreement, the Company shall not deliver any Draw Down Notice or otherwise offer or sell Shares to the Investor, and the Investor shall not be obligated to purchase any Shares pursuant to this Agreement, (i) during any period in which the Company is, or may be deemed to be, in possession of material non-public information, (ii) during any period (other than the period referred to in clause (iii) of this Section 3.8) in which the Company's insider trading policy, as it exists from time to time, would prohibit purchases or sales of Common Stock by its officers or directors (each such period, a "Blackout Period"), except with respect to this clause (ii) as expressly provided in the immediately following sentence, or (iii) except as expressly provided in this Section 3.8, at any time from and including the date (each, an "Announcement Date") on which the Company shall issue a press release containing, or shall otherwise publicly announce, its earnings, revenues or other results of operations (each, an "Earnings Announcement") through and including the time that is 24 hours after the time that the Company files (a "Filing Time") a Quarterly Report on Form 10-Q or an Annual Report on Form 10-K that includes consolidated financial statements as of and for the same period or periods, as the case may be, covered by such Earnings Announcement. If the Company wishes to deliver any Draw Down Notice or otherwise offer, sell or deliver Shares to the Investor during any Blackout Period, the Company shall, as a condition thereto, provide the Investor with the compliance certificate substantially in the form attached hereto as Exhibit D, dated the date of such Draw Down Notice, which certificate shall be deemed to remain in effect during the applicable Pricing Period through and including the applicable Settlement Date, and the "bring down" opinions in the form mutually agreed to by the parties hereto, dated the date of such Draw Down Notice. If the Company wishes to deliver any Draw Down Notice or otherwise offer, sell or deliver Shares to the Investor at any time during the period from and including an Announcement Date through and including the time that is 24 hours after the corresponding Filing Time, the Company shall, as conditions thereto, (1) prepare and deliver to the Investor (with a copy to counsel to the Investor) a report on Form 8-K which shall include substantially the same financial and related information as was set forth in the relevant Earnings Announcement (other than any earnings or other projections, similar forward-looking data and officers' quotations) (each, an "Earnings 8-K"), in form and substance reasonably satisfactory to the Investor and its counsel, (2) provide the Investor with the compliance certificate substantially in the form attached hereto as Exhibit D, dated the date of such Draw Down Notice, which certificate shall be deemed to remain in effect during the applicable Pricing Period through and including the applicable Settlement Date, and the "bring down" opinions in the form mutually agreed to by the parties hereto, dated the date of such Draw Down Notice and (3) file such Earnings 8-K with the Commission (so that it is deemed "filed" for purposes of Section 18 of the Exchange Act), include such Earnings 8-K in a Prospectus Supplement and file such Prospectus Supplement with the Commission under Rule 424(b) under the Securities Act, in each case on or prior to the date of such Draw Down Notice. The provisions of clause (iii) of this Section 3.8 shall not be applicable for the period from and after the time at which all of the conditions set forth in the immediately preceding sentence shall have been satisfied (or, if later, the time that is 24 hours after the time that the relevant Earnings Announcement was first publicly released) through and including the time that is 24 hours after the Filing Time of the relevant Quarterly Report on Form 10-Q or Annual Report on Form 10-K, as the case may be. For purposes of clarity, the parties agree that the delivery of the compliance certificate and the "bring down" opinions pursuant to this Section 3.8 shall not relieve the Company from any of its obligations under this Agreement with respect to the delivery of the compliance certificate called for by Section 7.2(ii) and the "bring down" opinions called for by Section 7.2(xv) on the applicable Settlement Date, which Sections shall have independent application.

ARTICLE IV REPRESENTATIONS, WARRANTIES AND COVENANTS OF THE INVESTOR

The Investor hereby makes the following representations, warranties and covenants to the Company:

- Section 4.1. Organization and Standing of the Investor. The Investor is a limited liability company duly organized, validly existing and in good standing under the laws of the State of New York.
- Section 4.2. <u>Authorization and Power</u>. The Investor has the requisite corporate power and authority to enter into and perform its obligations under this Agreement and the Registration Rights Agreement and to purchase or acquire the Securities in accordance with the terms hereof. The execution, delivery and performance by the Investor of this Agreement and the Registration Rights Agreement and the consummation by it of the transactions contemplated hereby and thereby have been duly authorized by all necessary corporate action, and no further consent or authorization of the Investor, its Board of Directors or its stockholders is required. Each of this Agreement and the Registration Rights Agreement has been duly executed and delivered by the Investor and constitutes a valid and binding obligation of the Investor enforceable against it in accordance with its terms, except as such enforceability may be limited by applicable bankruptcy, insolvency, reorganization, moratorium, liquidation, conservatorship, receivership, or similar laws relating to, or affecting generally the enforcement of, creditor's rights and remedies or by other equitable principles of general application (including any limitation of equitable remedies).
- Section 4.3. No Conflicts. The execution, delivery and performance by the Investor of this Agreement and the Registration Rights Agreement and the consummation by the Investor of the transactions contemplated hereby and thereby do not and shall not (i) result in a violation of such Investor's charter documents, bylaws or other applicable organizational instruments, (ii) conflict with, constitute a default (or an event which, with notice or lapse of time or both, would become a default) under, or give rise to any rights of termination, amendment, acceleration or cancellation of, any material agreement, mortgage, deed of trust, indenture, note, bond, license, lease agreement, instrument or obligation to which the Investor is a party or is bound, (iii) create or impose any lien, charge or encumbrance on any property of the Investor under any agreement or any commitment to which the Investor is party or under which the Investor is bound or under which any of its properties or assets are bound, or (iv) result in a violation of any federal, state, local or foreign statute, rule, or regulation, or any order, judgment or decree of any court or governmental agency applicable to the Investor or by which any of its properties or assets are bound or affected, except, in the case of clauses (ii), (iii) and (iv), for such conflicts, defaults, terminations, amendments, acceleration, cancellations and violations as would not, individually or in the aggregate, prohibit or otherwise interfere with, in any material respect, the ability of the Investor to enter into and perform its obligations under this Agreement and the Registration Rights Agreement. The Investor is not required under any applicable federal, state, local or foreign law, rule or regulation to obtain any consent, authorization or order of, or make any filing or registration with, any court or governmental agency in order for it to execute, deliver or perform any of its obligations under this Agreement and the Registration Rights Agreement or to purchase the Securi
- Section 4.4. <u>Investment Purpose</u>. The Investor is acquiring the Securities for its own account, for investment purposes and not with a view towards, or for resale in connection with, the public sale or distribution thereof, except pursuant to sales registered under or exempt from the registration requirements of the Securities Act; <u>provided</u>, <u>however</u>, that by making the representations herein, the Investor does not agree, or make any representation or warranty, to hold any of the Securities for any minimum or other specific term and reserves the right to dispose of the Securities at any time in accordance with or pursuant to a registration statement or an exemption under the Securities Act. The Investor does not presently have any agreement or understanding, directly or indirectly, with any Person to distribute any of the Securities.

- Section 4.5. Accredited Investor Status. The Investor is an "accredited investor" as that term is defined in Rule 501(a) of Regulation D.
- Section 4.6. Reliance on Exemptions. The Investor understands that the Securities are being offered and sold to it in reliance on specific exemptions from the registration requirements of U.S. federal and state securities laws and that the Company is relying in part upon the truth and accuracy of, and the Investor's compliance with, the representations, warranties, agreements, acknowledgments and understandings of the Investor set forth herein in order to determine the availability of such exemptions and the eligibility of the Investor to acquire the Securities.
- Section 4.7. Information. All materials relating to the business, financial condition, management and operations of the Company and materials relating to the offer and sale of the Securities which have been requested by the Investor have been furnished or otherwise made available to the Investor or its advisors, including, without limitation, the Commission Documents. The Investor understands that its investment in the Securities and has such knowledge and experience in financial and business matters that it is capable of evaluating the merits and risks of a proposed investment in the Securities. The Investor and its advisors have been afforded the opportunity to ask questions of and receive answers from representatives of the Company concerning the financial condition and business of the Company and other matters relating to an investment in the Securities. Neither such inquiries nor any other due diligence investigations conducted by the Investor or its advisors, if any, or its representatives shall modify, amend or affect the Investor's right to rely on the Company's representations and warranties contained in this Agreement or in any other Transaction Document to which the Company is a party or the Investor's right to rely on any other document or instrument executed and/or delivered in connection with this Agreement or the consummation of the transaction contemplated hereby (including, without limitation, the opinions of the Company's counsel delivered pursuant to Sections 7.1(iv) and 7.2(xv)). The Investor has sought such accounting, legal and tax advice as it has considered necessary to make an informed investment decision with respect to its acquisition of the Securities. The Investor understands that it (and not the Company) shall be responsible for its own tax liabilities that may arise as a result of this investment or the transactions contemplated by this Agreement.
- Section 4.8. No Governmental Review. The Investor understands that no United States federal or state agency or any other government or governmental agency has passed on or made any recommendation or endorsement of the Securities or the fairness or suitability of the investment in the Securities nor have such authorities passed upon or endorsed the merits of the offering of the Securities.
- Section 4.9. <u>No General Solicitation</u>. The Investor is not purchasing the Securities as a result of any form of general solicitation or general advertising (within the meaning of Regulation D) in connection with the offer or sale of the Securities.
 - Section 4.10. Not an Affiliate. The Investor is not an officer, director or an Affiliate of the Company.

- Section 4.11. <u>Statutory Underwriter Status</u>. The Investor acknowledges that it will be disclosed as an "underwriter" and a "selling stockholder" in the Registration Statement and in any Prospectus contained therein to the extent required by applicable law and to the extent the Prospectus is related to the resale of Registrable Securities.
- Section 4.12. Resales of Securities. The Investor represents, warrants and covenants that unless the Securities are eligible for resale pursuant to Rule 144, it will resell such Securities only pursuant to the Registration Statement, in a manner described under the caption "Plan of Distribution" in the Registration Statement, and in a manner in compliance with all applicable U.S. federal and state securities laws, rules and regulations, including, without limitation, any applicable prospectus delivery requirements of the Securities Act.

ARTICLE V REPRESENTATIONS, WARRANTIES AND COVENANTS OF THE COMPANY

Except as set forth in the disclosure schedule delivered by the Company to the Investor (which is hereby incorporated by reference in, and constitutes an integral part of, this Agreement) (the "Disclosure Schedule"), the Company hereby makes the following representations, warranties and covenants to the Investor:

- Section 5.1. Organization, Good Standing and Power. The Company and each of its Subsidiaries is a corporation duly organized, validly existing and in good standing under the laws of the jurisdiction of its incorporation and has the requisite corporate power and authority to own, lease and operate its properties and assets and to conduct its business as it is now being conducted and as presently proposed to be conducted. The Company and each Subsidiary is duly qualified as a foreign corporation to do business and is in good standing in every jurisdiction in which the nature of the business conducted or property owned by it makes such qualification necessary, except for any jurisdiction in which the failure to be so qualified would not have a Material Adverse Effect.
- Section 5.2. Authorization, Enforcement. The Company has the requisite corporate power and authority to enter into and perform its obligations under each of the Transaction Documents to which it is a party and to issue the Securities in accordance with the terms hereof and thereof. Except for approvals of the Company's Board of Directors or a committee thereof as may be required in connection with any issuance and sale of Securities to the Investor hereunder (which approvals shall be obtained prior to the delivery of any Draw Down Notice), the execution, delivery and performance by the Company of each of the Transaction Documents to which it is a party and the consummation by it of the transactions contemplated hereby and thereby have been duly and validly authorized by all necessary corporate action, and no further consent or authorization of the Company and constitutes a valid and binding obligation of the Company enforceable against the Company in accordance with its terms, except as such enforceability may be limited by applicable bankruptcy, insolvency, reorganization, moratorium, liquidation, conservatorship, receivership or similar laws relating to, or affecting generally the enforcement of, creditor's rights and remedies or by other equitable principles of general application (including any limitation of equitable remedies).

Section 5.3. Capitalization and Voting Rights. The authorized capital stock of the Company and the shares thereof issued and outstanding were as set forth in the Commission Documents as of the dates reflected therein. All of the outstanding shares of Common Stock have been duly authorized and validly issued, and are fully paid and nonassessable. Except as set forth in the Commission Documents, this Agreement and the Registration Rights Agreement, there are no agreements or arrangements under which the Company is obligated to register the sale of any securities under the Securities Act. Except as set forth in the Commission Documents, no shares of Common Stock are entitled to preemptive rights and there are no outstanding debt securities and no contracts, commitments, understandings, or arrangements by which the Company is or may become bound to issue additional shares of the capital stock of the Company or options, warrants, scrip, rights to subscribe to, calls or commitments of any character whatsoever relating to, or securities or rights convertible into or exchangeable for, any shares of capital stock of the Company other than those issued or granted in the ordinary course of business pursuant to the Company's equity incentive and/or compensatory plans or arrangements. Except for customary transfer restrictions contained in agreements entered into by the Company to sell restricted securities or as set forth in the Commission Documents, the Company is not a party to, and it has no Knowledge of, any agreement restricting the voting or transfer of any shares of the capital stock of the Company. Except as set forth in the Commission Documents, the offer and sale of all capital stock, convertible or exchangeable securities, rights, warrants or options of the Company issued prior to the Closing Date complied with all applicable federal and state securities laws, and no stockholder has any right of rescission or damages or any "put" or similar right with respect thereto that would have a Material Adverse Effect. Ex

Section 5.4. <u>Issuance of Securities</u>. The Initial Commitment Shares have been, and the Shares to be issued under this Agreement have been or will be (prior to the delivery of any Draw Down Notice to the Investor hereunder), duly authorized by all necessary corporate action on the part of the Company. The Commitment Shares, when issued in accordance with the terms of this Agreement, and the Shares, when issued and paid for in accordance with the terms of this Agreement, shall be validly issued and outstanding, fully paid and nonassessable and free from all liens, charges, taxes, security interests, encumbrances, rights of first refusal, preemptive or similar rights and other encumbrances with respect to the issue thereof.

Section 5.5. No Conflicts. The execution, delivery and performance by the Company of each of the Transaction Documents to which it is a party and the consummation by the Company of the transactions contemplated hereby and thereby do not and shall not (i) result in a violation of any provision of the Company's Charter or Bylaws, (ii) conflict with or result in a breach or violation of any of the terms or provisions of, or constitute a default (or an event which, with notice or lapse of time or both, would become a default) under, or give rise to any rights of termination, amendment, acceleration or cancellation of, any material agreement, mortgage, deed of trust, indenture, note, bond, license, lease agreement, instrument or obligation to which the Company or any of its Significant Subsidiaries is a party or is bound, (iii) create or impose a lien, charge or encumbrance on any property or assets of the Company or any of its Significant Subsidiaries under any agreement or any commitment to which the Company or any of its Significant Subsidiaries is a party or by which the Company or any of its Significant Subsidiaries is bound or to which any of their respective properties or assets is subject, or (iv) result in a violation of any federal, state, local or foreign statute, rule, regulation, order, judgment or decree applicable to the Company or any of its Subsidiaries or by which any property or asset of the Company or any of its Subsidiaries are bound or affected (including federal and state securities laws and regulations and the rules and regulations of the Trading Market), except, in the case of clauses (ii), (iii) and (iv), for such conflicts, defaults, terminations, amendments, acceleration, cancellations, liens, charges, encumbrances and violations as would not, individually or in the aggregate, have a Material Adverse Effect. Except as specifically contemplated by this Agreement or the Registration Rights Agreement and as required under the Securities Act and any applicable state securities laws, the Company is not required under any federal, state, local or foreign law, rule or regulation to obtain any consent, authorization or order of, or make any filing or registration with, any court or governmental agency (including, without limitation, the Trading Market) in order for it to execute, deliver or perform any of its obligations under the Transaction Documents to which it is a party, or to issue the Securities to the Investor in accordance with the terms hereof and thereof (other than such consents, authorizations, orders, filings or registrations as have been obtained or made prior to the Closing Date); provided, however, that, for purposes of the representation made in this sentence, the Company is assuming and relying upon the accuracy of the representations and warranties of the Investor in this Agreement and the compliance by it with its covenants and agreements contained in this Agreement and the Registration Rights Agreement.

Section 5.6. Commission Documents, Financial Statements. (a) The Company has timely filed (giving effect to permissible extensions in accordance with Rule 12b-25 under the Exchange Act) all Commission Documents. The Company has delivered or made available to the Investor via EDGAR or otherwise true and complete copies of the Commission Documents filed with or furnished to the Commission prior to the Closing Date (including, without limitation, the 2012 Form 10-K). No Subsidiary of the Company is required to file or furnish any report, schedule, registration, form, statement, information or other document with the Commission. As of its filing date, each Commission Document filed with or furnished to the Commission prior to the Closing Date complied in all material respects with the requirements of the Securities Act or the Exchange Act, as applicable, and other federal, state and local laws, rules and regulations applicable to it, and, as of its filing date (or, if amended or superseded by a filing prior to the Closing Date, on the date of such amended or superseded filing), such Commission Document did not contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading. The Registration Statement, on the date it is filed with the Commission, on the date it is declared effective by the Commission, on each Draw Down Exercise Date and on each Settlement Date, shall comply in all material respects with the requirements of the Securities Act (including, without limitation, Rule 415 under the Securities Act) and shall not contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary in order to make the statements therein not misleading, except that this representation and warranty shall not apply to statements in or omissions from the Registration Statement made in reliance upon and in conformity with information relating to the Investor furnished to the Company in writing by or on behalf of the Investor expressly for use therein. The Prospectus and each Prospectus Supplement required to be filed pursuant to this Agreement or the Registration Rights Agreement after the Closing Date, when taken together, on its date, on each Draw Down Exercise Date and on each Settlement Date, shall comply in all material respects with the requirements of the Securities Act (including, without limitation, Rule 424(b) under the Securities Act) and shall not contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading, except that this representation and warranty shall not apply to statements in or omissions from the Prospectus or any Prospectus Supplement made in reliance upon and in conformity with information relating to the Investor furnished to the Company in writing by or on behalf of the Investor expressly for use therein. Each Commission Document (other than the Registration Statement, the Prospectus or any Prospectus Supplement) to be filed with or furnished to the Commission after the Closing Date and incorporated by reference in the Registration Statement, the Prospectus or any Prospectus Supplement required to be filed pursuant to this Agreement or the Registration Rights Agreement (including, without limitation, the Current Report), when such document is filed with or furnished to the Commission and, if applicable, when such document becomes effective, as the case may be, shall comply in all material respects with the requirements of the Securities Act or the Exchange Act, as applicable, and other federal, state and local laws, rules and regulations applicable to it, and shall not contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading. The Company has delivered or made available to the Investor via EDGAR or otherwise true and complete copies of all comment letters and substantive correspondence received by the Company from the Commission relating to the Commission Documents filed with or furnished to the Commission as of the Closing Date, together with all written responses of the Company thereto in the form such responses were filed via EDGAR. There are no outstanding or unresolved comments or undertakings in such comment letters received by the Company from the Commission. The Commission has not issued any stop order or other order suspending the effectiveness of any registration statement filed by the Company under the Securities Act or the Exchange Act.

- (b) The financial statements, together with the related notes and schedules, of the Company included in the Commission Documents comply as to form in all material respects with all applicable accounting requirements and the published rules and regulations of the Commission and all other applicable rules and regulations with respect thereto. Such financial statements, together with the related notes and schedules, have been prepared in accordance with GAAP applied on a consistent basis during the periods involved (except (i) as may be otherwise indicated in such financial statements or the notes thereto or (ii) in the case of unaudited interim statements, to the extent they may not include footnotes or may be condensed or summary statements), and fairly present in all material respects the financial condition of the Company and its consolidated Subsidiaries as of the dates thereof and the results of operations and cash flows for the periods then ended (subject, in the case of unaudited statements, to normal year-end audit adjustments).
- (c) The Company has timely filed with the Commission and made available to the Investor via EDGAR or otherwise all certifications and statements required by (x) Rule 13a-14 or Rule 15d-14 under the Exchange Act or (y) 18 U.S.C. Section 1350 (Section 906 of the Sarbanes-Oxley Act of 2002 ("SOXA")) with respect to all relevant Commission Documents. The Company is in compliance in all material respects with the provisions of SOXA applicable to it as of the date hereof. The Company maintains disclosure controls and procedures required by Rule 13a-15 or Rule 15d-15 under the Exchange Act; such controls and procedures are effective to ensure that all material information concerning the Company and its Subsidiaries is made known on a timely basis to the individuals responsible for the timely and accurate preparation of the Company's Commission filings and other public disclosure documents. As used in this Section 5.6(c), the term "file" shall be broadly construed to include any manner in which a document or information is furnished, supplied or otherwise made available to the Commission.
- (d) Tarvaran, Askelson & Company, LLP, who shall express their opinion on the audited financial statements and related schedules to be included or incorporated by reference in the Registration Statement and the Prospectus are, with respect to the Company, independent public accountants as required by the Securities Act and is an independent registered public accounting firm within the meaning of SOXA as required by the rules of the Public Company Accounting Oversight Board. Tarvaran, Askelson & Company, LLP has not been engaged by the Company to perform any "prohibited activities" (as defined in Section 10A of the Exchange Act).
- Section 5.7. <u>Subsidiaries</u>. The Current Report on Form 8-K dated February 9, 2012, sets forth each Subsidiary of the Company as of the Closing Date, showing its jurisdiction of incorporation or organization and the percentage of the Company's ownership of the outstanding capital stock or other ownership interests of such Subsidiary, and the Company does not have any other Subsidiaries as of the Closing Date.

- Section 5.8. No Material Adverse Effect. Except as disclosed in any Commission Documents filed since November 19, 2012, since November 19, 2012, the Company has not experienced or suffered any Material Adverse Effect, and there exists no current state of facts, condition or event which would have a Material Adverse Effect.
- Section 5.9. No Undisclosed Liabilities. Neither the Company nor any of its Subsidiaries has any liabilities, obligations, claims or losses (whether liquidated or unliquidated, secured or unsecured, absolute, accrued, contingent or otherwise) that would be required to be disclosed on a balance sheet of the Company or any Subsidiary (including the notes thereto) in conformity with GAAP and are not disclosed in the Commission Documents, other than those incurred in the ordinary course of the Company's or its Subsidiaries respective businesses since November 19, 2012 and which, individually or in the aggregate, do not or would not have a Material Adverse Effect.
- Section 5.10. No Undisclosed Events or Circumstances. No event or circumstance has occurred or information exists with respect to the Company or any of its Subsidiaries or its or their business, properties, liabilities, operations (including results thereof) or conditions (financial or otherwise), which, under applicable law, rule or regulation, requires public disclosure or announcement by the Company at or before the Closing but which has not been so publicly announced or disclosed, except for events or circumstances which, individually or in the aggregate, do not or would not have a Material Adverse Effect.
- Section 5.11. Indebtedness; Solvency. The Company's Quarterly Report on Form 10-Q for its fiscal quarter ended September 30, 2012 sets forth, as of September 30, 2012, all outstanding secured and unsecured Indebtedness of the Company or any Subsidiary, or for which the Company or any Subsidiary has commitments through such date. For the purposes of this Agreement, "Indebtedness" shall mean (a) any liabilities for borrowed money or amounts owed in excess of \$1,000 (other than trade accounts payable incurred in the ordinary course of business), (b) all guaranties, endorsements, indemnities and other contingent obligations in respect of Indebtedness of others in excess of \$1,000, whether or not the same are or should be reflected in the Company's balance sheet (or the notes thereto), except guaranties by endorsement of negotiable instruments for deposit or collection or similar transactions in the ordinary course of business; and (c) the present value of any lease payments in excess of \$1,000 due under leases required to be capitalized in accordance with GAAP. There is no existing or continuing default or event of default in respect of any Indebtedness of the Company or any of its Subsidiaries. The Company has not taken any steps, and does not currently expect to take any steps, to seek protection pursuant to Title 11 of the United States Code or any similar federal or state bankruptcy law or law for the relief of debtors, nor does the Company have any Knowledge that its creditors intend to initiate involuntary bankruptcy, insolvency, reorganization or liquidation proceedings or other proceedings for relief under Title 11 of the United States Code or any other federal or state bankruptcy law or any law for the relief of debtors. The Company is financially solvent and is generally able to pay its debts as they become due.

Section 5.12. <u>Title To Assets</u>. Each of the Company and its Subsidiaries has good and valid title to, or has valid rights to lease or otherwise use, all of their respective real and personal property reflected in the Commission Documents, free of mortgages, pledges, charges, liens, security interests or other encumbrances, except for those indicated in the Commission Documents and those that would not have a Material Adverse Effect. All real property and facilities held under lease by the Company or any of its Subsidiaries are held by them under valid, subsisting and enforceable leases with such exceptions as are not material and do not interfere with the use made and proposed to be made of such property and buildings by the Company or any of its Subsidiaries.

Section 5.13. Actions Pending. To the Knowledge of the Company, there is no action, suit, claim, investigation or proceeding pending or threatened against the Company or any Subsidiary which questions the validity of the Transaction Documents or the transactions contemplated thereby or any action taken or to be taken pursuant thereto. Except as set forth in the Commission Documents, to the Knowledge of the Company, there is no action, suit, claim, investigation or proceeding pending or threatened against or involving the Company, any Subsidiary or any of their respective properties or assets, or involving any officers or directors of the Company or any of its Subsidiaries, including, without limitation, any securities class action lawsuit or stockholder derivative lawsuit related to the Company, in each case which, if determined adversely to the Company, its Subsidiary or any officer or director of the Company or its Subsidiaries, would have a Material Adverse Effect. Except as set forth in the Commission Documents, no judgment, order, writ, injunction or decree or award has been issued by or, to the Knowledge of the Company, requested of any court, arbitrator or governmental agency which would be reasonably expected to result in a Material Adverse Effect.

Section 5.14. Compliance With Law. The business of the Company and the Subsidiaries has been and is presently being conducted in compliance with all applicable federal, state, local and foreign governmental laws, rules, regulations and ordinances, except as set forth in the Commission Documents and except for such non-compliance which, individually or in the aggregate, would not have a Material Adverse Effect. Neither the Company nor any of its Subsidiaries is in violation of any judgment, decree or order or any statute, ordinance, rule or regulation applicable to the Company or any of its Subsidiaries, and neither the Company nor any of its Subsidiaries will conduct its business in violation of any of the foregoing, except in all cases for possible violations which could not, individually or in the aggregate, have a Material Adverse Effect. Without limiting the generality of the foregoing, the Company has maintained all requirements for the continued listing or quotation of its Common Stock on the Trading Market, and the Company is not in violation of any of the rules, regulations or requirements of the Trading Market and has no Knowledge of any facts or circumstances that could reasonably lead to delisting or suspension of the Common Stock by the Trading Market in the foreseeable future.

Section 5.15. Certain Fees. No brokers, finders or financial advisory fees or commissions are or shall be payable by the Company or any Subsidiary (or any of their respective Affiliates) with respect to the transactions contemplated by the Transaction Documents, other than fees payable to Aegis Capital Corp..

Section 5.16. Disclosure. The Company confirms that neither it nor any other Person acting on its behalf has provided the Investor or any of its agents, advisors or counsel with any information that constitutes or could reasonably be expected to constitute material, nonpublic information concerning the Company or any of its Subsidiaries, other than the existence of the transactions contemplated by the Transaction Documents. The Company understands and confirms that the Investor will rely on the foregoing representations in effecting transactions in securities of the Company. All disclosure provided to Investor regarding the Company and its Subsidiaries, their businesses and the transaction contemplated by the Transaction Documents (including, without limitation, the representations and warranties of the Company contained in the Transaction Documents to which it is a party (as modified by the Disclosure Schedule)) furnished by or on behalf of the Company or any of its Subsidiaries, taken together, is true and correct and does not contain any untrue statement of a material fact or omit to state any material fact necessary in order to make the statements made therein, in the light of the circumstances under which they were made, not misleading. Each press release issued by the Company or any of its Subsidiaries during the 12 months preceding the Closing Date did not at the time of release (or, if amended or superseded by a later dated press release issued by the Company or any of its Subsidiaries prior to the Closing Date or by a later dated Commission Document filed with or furnished to the Commission by the Company prior to the Closing Date, at the time of issuance of such later dated press release or filing or furnishing of such Commission Document, as applicable) contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary in order to make the statements therein, in the light of the circumstances under which they are made, not misleading.

Section 5.17. Operation of Business. (a) The Company or one or more of its Subsidiaries possesses such permits, licenses, approvals, consents and other authorizations (including licenses, accreditation and other similar documentation or approvals of any local health departments) issued by the appropriate federal, state, local or foreign regulatory agencies or bodies as are necessary to conduct the business now operated by it (collectively, "Governmental Licenses"), except where the failure to possess such Governmental Licenses, individually or in the aggregate, would not have a Material Adverse Effect. The Company and its Subsidiaries are in compliance with the terms and conditions of all such Governmental Licenses, except where the failure to so comply, individually or in the aggregate, would not have a Material Adverse Effect or except as otherwise disclosed in the Commission Documents. All of the Governmental Licenses are valid and in full force and effect, except where the invalidity of such Governmental Licenses or the failure of such Governmental Licenses to be in full force and effect, individually or in the aggregate, would not have a Material Adverse Effect or except as otherwise disclosed in the Commission Documents. Except as set forth in the Commission Documents, neither the Company nor any of its Subsidiaries has received any written notice of proceedings relating to the revocation or modification of any such Governmental Licenses which, if the subject of any unfavorable decision, ruling or finding, individually or in the aggregate, would have a Material Adverse Effect. This Section 5.17 does not relate to environmental matters, such items being the subject of Section 5.18.

(b) The Company or one or more of its Subsidiaries owns or possesses adequate patents, patent rights, licenses, inventions, copyrights, know-how (including trade secrets and other unpatented and/or unpatentable proprietary or confidential information, systems or procedures), trademarks, service marks, trade names, trade dress, logos, copyrights and other intellectual property, including, without limitation, all of the intellectual property described in the Commission Documents as being owned or licensed by the Company (collectively, "Intellectual Property"), necessary to carry on the business now operated by it. Except as set forth in the Commission Documents, there are no actions, suits or judicial proceedings pending, or to the Company's Knowledge threatened, relating to patents or proprietary information to which the Company or any of its Subsidiaries is a party or of which any property of the Company or any of its Subsidiaries is aware of any infringement of or conflict with asserted rights of others with respect to any Intellectual Property or of any facts or circumstances which could render any Intellectual Property invalid or inadequate to protect the interest of the Company and its Subsidiaries therein, and which infringement or conflict (if the subject of any unfavorable decision, ruling or finding) or invalidity or inadequacy, individually or in the aggregate, would have a Material Adverse Effect.

Section 5.18. Environmental Compliance. Except as disclosed in the Commission Documents, the Company and each of its Subsidiaries have obtained all material approvals, authorization, certificates, consents, licenses, orders and permits or other similar authorizations of all governmental authorities, or from any other person, that are required under any Environmental Laws, except for any approvals, authorization, certificates, consents, licenses, orders and permits or other similar authorizations the failure of which to obtain does not or would not have a Material Adverse Effect. "Environmental Laws" shall mean all applicable laws relating to the protection of the environment including, without limitation, all requirements pertaining to reporting, licensing, permitting, controlling, investigating or remediating emissions, discharges, releases or threatened releases of hazardous substances, chemical substances, pollutants, contaminants or toxic substances, materials or wastes, whether solid, liquid or gaseous in nature, into the air, surface water, groundwater or land, or relating to the manufacture, processing, distribution, use, treatment, storage, disposal, transport or handling of hazardous substances, chemical substances, pollutants, contaminants or toxic substances, whether solid, liquid or gaseous in nature. Except for such instances as would not, individually or in the aggregate, have a Material Adverse Effect, to the Company's Knowledge, there are no past or present events, conditions, circumstances, incidents, actions or omissions relating to or in any way affecting the Company or its Subsidiaries that violate or could reasonably be expected to violate any Environmental Law after the Closing Date or that could reasonably be expected to give rise to any environmental liability, or otherwise form the basis of any claim, action, demand, suit, proceeding, hearing, study or investigation (i) under any Environmental Law, or (ii) based on or related to the manufacture, processing, distribution, use, treatment,

Section 5.19. Material Agreements. Except as set forth in the Commission Documents, neither the Company nor any Subsidiary of the Company is a party to any written or oral contract, instrument, agreement commitment, obligation, plan or arrangement, a copy of which would be required to be filed with the Commission as an exhibit to an annual report on Form 10-K (collectively, "Material Agreements"). Except as set forth in the Commission Documents, the Company and each of its Subsidiaries have performed in all material respects all the obligations then required to be performed by them under the Material Agreements, have received no notice of default or an event of default by the Company or any of its Subsidiaries thereunder and are not aware of any basis for the assertion thereof, and neither the Company or any of its Subsidiaries nor, to the Knowledge of the Company, any other contracting party thereto are in default under any Material Agreement now in effect, the result of which would have a Material Adverse Effect. Except as set forth in the Commission Documents, each of the Material Agreements is in full force and effect, and constitutes a legal, valid and binding obligation enforceable in accordance with its terms against the Company and/or any of its Subsidiaries and, to the Knowledge of the Company, each other contracting party thereto, except as such enforceability may be limited by applicable bankruptcy, insolvency, reorganization, moratorium, liquidation, conservatorship, receivership or similar laws relating to, or affecting generally the enforcement of, creditor's rights and remedies or by other equitable principles of general application.

- Section 5.20. <u>Transactions With Affiliates</u>. Except as set forth in the Commission Documents, there are no loans, leases, agreements, contracts, royalty agreements, management contracts, service arrangements or other continuing transactions exceeding \$120,000 between (a) the Company or any Subsidiary, on the one hand, and (b) any person or entity who would be covered by Item 404(a) of Regulation S-K, on the other hand. Except as disclosed in the Commission Documents, there are no outstanding amounts payable to or receivable from, or advances by the Company or any of its Subsidiaries to, and neither the Company or any of its Subsidiaries is otherwise a creditor of or debtor to, any beneficial owner of more than 5% of the outstanding shares of Common Stock, or any director, employee or affiliate of the Company or any of its Subsidiaries, other than (i) reimbursement for reasonable expenses incurred on behalf of the Company or any of its Subsidiaries or (ii) as part of the normal and customary terms of such person's employment or service as a director with the Company or any of its Subsidiaries.
- Section 5.21. Employees. Neither the Company nor any Subsidiary of the Company has any collective bargaining arrangements or agreements covering any of its employees, except as set forth in the Commission Documents. Except as disclosed in the Commission Documents, no officer, consultant or key employee of the Company or any Subsidiary whose termination, either individually or in the aggregate, would reasonably be expected to have a Material Adverse Effect, has terminated or, to the Knowledge of the Company, has any present intention of terminating his or her employment or engagement with the Company or any Subsidiary.
- Section 5.22. <u>Use of Proceeds</u>. The proceeds from the sale of the Shares shall be used by the Company and its Subsidiaries as set forth in the Prospectus and any Prospectus Supplement filed pursuant to Section 2.3 of this Agreement and pursuant to the Registration Rights Agreement.
- Section 5.23. <u>Investment Company Act Status</u>. The Company is not, and as a result of the consummation of the transactions contemplated by the Transaction Documents and the application of the proceeds from the sale of the Shares as set forth in the Prospectus and any Prospectus Supplement shall not be required to be registered as, an "investment company" or a company "controlled" by an "investment company," within the meaning of the Investment Company Act of 1940, as amended.
- Section 5.24. ERISA. No liability to the Pension Benefit Guaranty Corporation has been incurred with respect to any Plan by the Company or any of its Subsidiaries which has had or would have a Material Adverse Effect. No "prohibited transaction" (as defined in Section 406 of ERISA or Section 4975 of the Code) or "accumulated funding deficiency" (as defined in Section 302 of ERISA) or any of the events set forth in Section 4043(b) of ERISA has occurred with respect to any Plan which has had or would have a Material Adverse Effect, and the execution and delivery of this Agreement and the issuance and sale of the Securities hereunder shall not result in any of the foregoing events. Each Plan is in compliance in all material respects with applicable law, including ERISA and the Code; the Company has not incurred and does not expect to incur liability under Title IV of ERISA with respect to the termination of, or withdrawal from, any Plan; and each Plan for which the Company would have any liability that is intended to be qualified under Section 401(a) of the Code is so qualified in all material respects and nothing has occurred, whether by action or failure to act, which would cause the loss of such qualifications. As used in this Section 5.24, the term "Plan" shall mean an "employee pension benefit plan" (as defined in Section 3 of ERISA) which is or has been established or maintained, or to which contributions are or have been made, by the Company or any Subsidiary or by any trade or business, whether or not incorporated, which, together with the Company or any Subsidiary, is under common control, as described in Section 414(b) or (c) of the Code.

- Section 5.25. Taxes. The Company and each of its Subsidiaries (i) has filed all necessary federal, state and foreign income and franchise tax returns or has duly requested extensions thereof, except for those the failure of which to file would not have a Material Adverse Effect, (ii) has paid all federal, state, local and foreign taxes due and payable for which it is liable, except to the extent that any such taxes are being contested in good faith and by appropriate proceedings, except for such taxes the failure of which to pay would not have a Material Adverse Effect, and (iii) does not have any tax deficiency or claims outstanding or assessed or, to the Company's Knowledge, proposed against it which would have a Material Adverse Effect. There are no unpaid taxes in any material amount claimed to be due by the taxing authority of any jurisdiction, and the officers of the Company and its Subsidiaries know of no basis for any such claim. The Company is not operated in such a manner as to qualify as a passive foreign investment company, as defined in Section 1297 of the Code.
- Section 5.26. Insurance. The Company and its Subsidiaries are insured by insurers of recognized financial responsibility against such losses and risks and in such amounts as management of the Company believes to be prudent and customary in the businesses in which the Company and its Subsidiaries are engaged. Neither the Company nor any such Subsidiary has been refused any insurance coverage sought or applied for, and neither the Company nor any such Subsidiary has any reason to believe that it will be unable to renew its existing insurance coverage as and when such coverage expires or to obtain similar coverage from similar insurers as may be necessary to continue its business at a cost that would not have a Material Adverse Effect.
- Section 5.27. <u>U.S. Real Property Holding Corporation</u>. Neither the Company nor any of its Subsidiaries is, or has ever been, and so long as any of the Securities are held by the Investor, shall become a U.S. real property holding corporation within the meaning of Section 897 of the Code.
- Section 5.28. Exemption from Registration; Valid Issuances. Subject to, and in reliance on, the representations, warranties and covenants made herein by the Investor, the offer and sale of the Securities in accordance with the terms and conditions of this Agreement is exempt from the registration requirements of the Securities Act pursuant to Section 4(a)(2) and Rule 506 of Regulation D; provided, however, that at the request of and with the express agreement of the Investor, the Shares will be delivered to the Investor via book entry through DTC and will not bear legends noting restrictions as to resale of such securities under federal or state securities laws, nor will any such securities be subject to stop transfer instructions. Neither the offer or sale of the Securities pursuant to, nor the Company's performance of its obligations under, the Transaction Documents to which it is a party shall (i) result in the creation or imposition of any liens, charges, claims or other encumbrances upon the Securities, or (ii) entitle the holders of any outstanding shares of capital stock of the Company to preemptive or other rights to subscribe to or acquire the shares of Common Stock or other securities of the Company.

- Section 5.29. No General Solicitation or Advertising. Neither the Company, nor any of its Subsidiaries or Affiliates, nor any Person acting on its or their behalf, has engaged in any form of general solicitation or general advertising (within the meaning of Regulation D) in connection with the offer or sale of the Securities.
- Section 5.30. No Integrated Offering. None of the Company, its Subsidiaries or any of their Affiliates, nor any Person acting on their behalf has, directly or indirectly, made any offers or sales of any security or solicited any offers to buy any security, under circumstances that would require registration of the issuance of any of the Securities under the Securities Act, whether through integration with prior offerings or otherwise, or cause this offering of the Securities to require approval of stockholders of the Company under any applicable stockholder approval provisions, including, without limitation, under the rules and regulations of the Trading Market. None of the Company, its Subsidiaries, their Affiliates nor any Person acting on their behalf will take any action or steps referred to in the preceding sentence that would require registration of the issuance of any of the Securities under the Securities Act or cause the offering of any of the Securities to be integrated with other offerings.
- Section 5.31. <u>Dilutive Effect</u>. The Company is aware and acknowledges that issuance of the Securities could cause dilution to existing stockholders and could significantly increase the outstanding number of shares of Common Stock.
- Section 5.32. Manipulation of Price. Neither the Company nor any of its officers, directors or Affiliates has, and, to the Knowledge of the Company, no Person acting on their behalf has, (i) taken, directly or indirectly, any action designed or intended to cause or to result in the stabilization or manipulation of the price of any security of the Company, or which caused or resulted in, or which would in the future reasonably be expected to cause or result in, the stabilization or manipulation of the price of any security of the Company, in each case to facilitate the sale or resale of any of the Securities, or (ii) sold, bid for, purchased, or paid any compensation for soliciting purchases of, any of the Securities. Neither the Company nor any of its officers, directors or Affiliates will during the term of this Agreement, and, to the Knowledge of the Company, no Person acting on their behalf will during the term of this Agreement, take any of the actions referred to in the immediately preceding sentence.
- Section 5.33. Securities Act. The Company has complied and shall comply with all applicable federal and state securities laws in connection with the offer, issuance and sale of the Securities hereunder, including, without limitation, the applicable requirements of the Securities Act. The Registration Statement, upon filing with the Commission and at the time it is declared effective by the Commission, shall satisfy all of the requirements of the Securities Act to register the resale of the Registrable Securities by the Investor in accordance with the Registration Rights Agreement on a delayed or continuous basis under Rule 415 under the Securities Act at then-prevailing market prices, and not fixed prices. The Company is not, and has not previously been at any time, an issuer identified in, or subject to, Rule 144(i).

- Section 5.34. <u>Listing and Maintenance Requirements</u>. The Company's Common Stock is registered pursuant to Section 12(b) or 12(g) of the Exchange Act, and the Company has taken no action designed to, or which to its Knowledge is likely to have the effect of, terminating the registration of the Common Stock under the Exchange Act, nor has the Company received any notification that the Commission is contemplating terminating such registration. The Company has not, in the 12 months preceding the Closing Date, received notice from any Trading Market on which the Common Stock is or has been listed or quoted to the effect that the Company is not in compliance with the listing or maintenance requirements of such Trading Market. As of the Closing Date, the Company is in compliance with all such listing and maintenance requirements.
- Section 5.35. Application of Takeover Protections. The Company and its Board of Directors have taken all necessary action, if any, in order to render inapplicable any control share acquisition, business combination, poison pill (including any distribution under a rights agreement) or other similar anti-takeover provision under the Company's Charter or the laws of its state of incorporation that is or could become applicable to the Investor as a result of the Investor and the Company fulfilling their respective obligations or exercising their respective rights under the Transaction Documents (as applicable), including, without limitation, as a result of the Company's issuance of the Securities and the Investor's ownership of the Securities.
- Section 5.36. Foreign Corrupt Practices Act. None of the Company, any Subsidiary or, to the Knowledge of the Company, any director, officer, agent, employee, affiliate or other Person acting on behalf of the Company or any of its Subsidiaries, is aware of or has taken any action, directly or indirectly, that would result in a violation by such Persons of the Foreign Corrupt Practices Act of 1977, as amended, and the rules and regulations thereunder (collectively, the "FCPA"), including, without limitation, making use of the mails or any means or instrumentality of interstate commerce corruptly in furtherance of an offer, payment, promise to pay or authorization of the payment of any money, or other property, gift, promise to give, or authorization of the giving of anything of value to any "foreign official" (as such term is defined in the FCPA) or any foreign political party or official thereof or any candidate for foreign political office, in contravention of the FCPA. The Company and the Subsidiaries have conducted their respective businesses in compliance with the FCPA.
- Section 5.37. Money Laundering Laws. The operations of the Company and its Subsidiaries are and have been conducted at all times in compliance with applicable financial recordkeeping and reporting requirements of the Currency and Foreign Transactions Reporting Act of 1970, as amended, the money laundering statutes of all jurisdictions, the rules and regulations thereunder and any related or similar rules, regulations or guidelines, issued, administered or enforced by any governmental agency (collectively, the "Money Laundering Laws") and, to the Knowledge of the Company, no action, suit or proceeding by or before any court or governmental agency, authority or body or any arbitrator involving the Company or any of its Subsidiaries with respect to the Money Laundering Laws is pending or threatened.

Section 5.38. OFAC. None of the Company, any Subsidiary or, to the Knowledge of the Company, any director, officer, agent, employee, affiliate or Person acting on behalf of the Company or any of its Subsidiaries is currently subject to any U.S. sanctions administered by the Office of Foreign Assets Control of the U.S. Treasury Department ("OFAC"); and the Company will not directly or indirectly use the proceeds of the offering, or lend, contribute or otherwise make available such proceeds to any subsidiary, joint venture partner or other Person, for the purpose of financing the activities of any Person currently subject to any U.S. sanctions administered by OFAC.

Section 5.39. [intentionally omitted]

Section 5.40. Royalties and Commissions. Except as set forth in the Commission Documents, neither the Company nor any of its Subsidiaries has any responsibility or obligation to pay or have paid on their behalf any material commission, royalty or similar payment to any person with respect to their property rights.

Section 5.41. Acknowledgement Regarding Investor's Acquisition of Securities. The Company acknowledges and agrees that the Investor is acting solely in the capacity of an arm's length purchaser with respect to this Agreement and the transactions contemplated by the Transaction Documents. The Company further acknowledges that the Investor is not acting as a financial advisor or fiduciary of the Company (or in any similar capacity) with respect to this Agreement and the transactions contemplated by the Transaction Documents, and any advice given by the Investor or any of its representatives or agents in connection therewith is merely incidental to the Investor's acquisition of the Securities. The Company further represents to the Investor that the Company's decision to enter into the Transaction Documents to which it is a party has been based solely on the independent evaluation of the transactions contemplated thereby by the Company and its representatives. The Company acknowledges and agrees that the Investor has not made and does not make any representations or warranties with respect to the transactions contemplated by the Transaction Documents other than those specifically set forth in Article IV of this Agreement.

ARTICLE VI ADDITIONAL COVENANTS

The Company covenants with the Investor, and the Investor covenants with the Company, as follows, which covenants of one party are for the benefit of the other party, during the Investment Period:

Section 6.1. Securities Compliance. The Company shall notify the Commission and the Trading Market, if and as applicable, in accordance with their respective rules and regulations, of the transactions contemplated by the Transaction Documents, and shall take all necessary action, undertake all proceedings and obtain all registrations, permits, consents and approvals for the legal and valid issuance of the Securities to the Investor in accordance with the terms of the Transaction Documents, as applicable.

Section 6.2. Reservation of Common Stock. The Company has available and the Company shall reserve and keep available at all times, free of preemptive and other similar rights of stockholders, the requisite aggregate number of authorized but unissued shares of Common Stock to enable the Company to timely effect the issuance, sale and delivery in full to the Investor of all Securities to be issued and delivered under this Agreement, in any case prior to the issuance to the Investor of such Securities. The number of shares of Common Stock so reserved from time to time, as theretofore increased or reduced as hereinafter provided, may be reduced by the number of shares of Common Stock actually delivered pursuant to this Agreement.

Section 6.3. Registration and Listing. The Company shall take all action necessary to cause the Common Stock to continue to be registered as a class of securities under Sections 12(b) or 12(g) of the Exchange Act, shall comply with its reporting and filing obligations under the Exchange Act, and shall not take any action or file any document (whether or not permitted by the Securities Act or the Exchange Act) to terminate or suspend such registration or to terminate or suspend its reporting and filing obligations under the Exchange Act or Securities Act, except as permitted herein. The Company shall use its reasonable best efforts to continue the listing and trading of its Common Stock and the listing of the Securities purchased or acquired by the Investor hereunder on the Trading Market and to comply with the Company's reporting, filing and other rules and regulations of the Trading Market. The Company shall not take any action which could be reasonably expected to result in the delisting or suspension of the Common Stock on the Trading Market. If the Company receives any final and non-appealable notice that the listing or quotation of the Common Stock on the Trading Market shall be terminated on a date certain, the Company shall promptly (and in any case within 48 hours) notify the Investor of such fact in writing and shall use its reasonable best efforts to cause the Common Stock to be listed or quoted on another Trading Market prior to such date certain.

Section 6.4. Compliance with Laws.

(i) The Company shall comply, and cause each Subsidiary to comply, (a) with all laws, rules, regulations and orders applicable to the business and operations of the Company and its Subsidiaries, except as would not have a Material Adverse Effect and (b) with all applicable provisions of the Securities Act and the Exchange Act and the rules and regulations of the Trading Market. Without limiting the foregoing, neither the Company, nor any of its Subsidiaries, nor to the Knowledge of the Company, any of their respective directors, officers, agents, employees or any other Persons acting on their behalf shall, in connection with the operation of the Company's and its Subsidiaries' respective businesses, (1) use any corporate funds for unlawful contributions, payments, gifts or entertainment or to make any unlawful expenditures relating to political activity to government officials, candidates or members of political parties or organizations, (2) pay, accept or receive any unlawful contributions, payments, expenditures or gifts, or (3) violate or operate in noncompliance with any export restrictions, anti-boycott regulations, embargo regulations or other applicable domestic or foreign laws and regulations, including, without limitation, the FCPA and the Money Laundering Laws.

(ii) The Investor shall comply with all laws, rules, regulations and orders applicable to the performance by it of its obligations under this Agreement and its investment in the Securities, except as would not, individually or in the aggregate, prohibit or otherwise interfere with the ability of the Investor to enter into and perform its obligations under this Agreement in any material respect. Without limiting the foregoing, the Investor shall comply with all applicable provisions of the Securities Act and the Exchange Act, including Regulation M thereunder, and any applicable securities laws of any non-U.S. jurisdictions.

Section 6.5. Keeping of Records and Books of Account; Due Diligence.

- (i) The Company shall keep and cause each Subsidiary to keep adequate records and books of account, in which complete entries shall be made in accordance with GAAP consistently applied, reflecting all financial transactions of the Company and its Subsidiaries, and in which, for each fiscal year, all proper reserves for depreciation, depletion, obsolescence, amortization, taxes, bad debts and other purposes in connection with its business shall be made. The Company shall maintain a system of internal accounting controls that (a) pertain to the maintenance of records that in reasonable detail accurately and fairly reflect the transactions and dispositions of the assets of the Company; (b) provide reasonable assurance that transactions are recorded as necessary to permit preparation of financial statements in accordance with generally accepted accounting principles, and that receipts and expenditures of the Company are being made only in accordance with authorizations of management and directors of the Company; and (c) provide reasonable assurance regarding prevention or timely detection of unauthorized acquisition, use or disposition of the Company's assets that could have a material effect on the Company's financial statements (it being acknowledged and agreed that the identification by the Company and/or its independent registered public accounting firm of any "significant deficiencies" or "material weaknesses" (each as defined by the Public Company Accounting Oversight Board) in the Company's internal controls over its financial reporting shall not, in and of itself, constitute a breach of this Section 6.5(i)).
- (ii) Subject to the requirements of Section 6.16 of this Agreement, from time to time from and after the Closing Date, the Company shall make available for inspection and review by the Investor during normal business hours and after reasonable notice, customary documentation reasonably requested by the Investor and/or its appointed counsel or advisors to conduct due diligence; <u>provided</u>, <u>however</u>, that after the Closing Date, the Investor's continued due diligence shall not be a condition to the issuance of any Draw Down Notice or the settlement of any Draw Down.
- Section 6.6. <u>Limitations on Holdings and Issuances</u>. The Company shall not be obligated to issue and the Investor shall not be obligated to purchase any shares of Common Stock which would cause the aggregate number of shares of Common Stock then beneficially owned (as calculated pursuant to Section 13(d) of the Exchange Act and Rule 13d-3 promulgated thereunder) by the Investor and its Affiliates to exceed the Ownership Limitation. Promptly following any request by the Company, the Investor shall inform the Company of the number of shares of Common Stock then beneficially owned by the Investor and its Affiliates.

Section 6.7. Other Agreements and Alternate Transactions.

(i) The Company shall not enter into, announce or recommend to its stockholders any agreement, plan, arrangement or transaction in or of which the terms thereof would restrict, materially delay, conflict with or impair the ability or right of the Company to perform its obligations under the Transaction Documents to which it is a party, including, without limitation, the obligation of the Company to deliver (i) the Initial Commitment Shares to the Investor not later than 4:00 p.m. (New York time) on the second Trading Day immediately following the Closing Date, and (ii) the Shares to the Investor in respect of a Draw Down on the applicable Settlement Date. For the avoidance of doubt, nothing in this Section 6.7(i) shall in any way limit the Company's right to terminate this Agreement in accordance with Section 8.1 (subject in all cases to Section 8.3).

(ii) If the Company enters into any agreement, plan, arrangement or transaction with a third party or seeks to utilize any existing agreement, plan or arrangement with a third party, in each case the principal purpose of which is to implement, effect or consummate, at any time during the period beginning on the first Trading Day of any Pricing Period and ending on the second Trading Day next following the applicable Settlement Date (the "Reference Period"), an Alternate Transaction that does not constitute an Acceptable Transaction, the Company shall provide prompt notice thereof (an "Alternate Transaction Notice") to the Investor; provided, however, that such Alternate Transaction Notice must be received by the Investor not later than the earlier of (a) 48 hours after the Company's execution of any agreement, plan, arrangement or transaction relating to such Alternate Transaction (or, with respect to any existing agreement, plan or arrangement to implement, effect or consummate such Other Financing) and (b) the second Trading Day immediately preceding the applicable Settlement Date with respect to the applicable Draw Down Notice. If required under applicable law, including, without limitation, Regulation FD promulgated by the Commission, or under the applicable rules and regulations of the Trading Market, the Company shall simultaneously publicly disclose the information included in any Alternate Transaction Notice in accordance with Regulation FD and the applicable rules and regulations of the Trading Market. For purposes of this Section 6.7(ii), any press release issued by, or Commission Document filed by, the Company shall constitute sufficient notice, provided that it is issued or filed, as the case may be, within the time requirements set forth in the first sentence (including the provisos thereto) of this Section 6.7(ii) for an Alternate Transaction Notice. With respect to any Reference Period for which the Company is required to provide an Alternate Transaction Notice pursuant to the first sentence of t

(iii) For all purposes of this Agreement, an "Alternate Transaction" shall mean (w) the issuance of Common Stock for a purchase price less than, or the issuance of securities convertible into or exchangeable for Common Stock at an exercise or conversion price (as the case may be) less than, the then Current Market Price of the Common Stock (including, without limitation, pursuant to any "equity line" or other financing that is substantially similar to the financing provided for under this Agreement, or pursuant to any other transaction in which the purchase, conversion or exchange price for such Common Stock is determined using a floating discount or other post-issuance adjustable discount to the then Current Market Price (any such transaction, a "Similar Financing")), in each case, after all fees, discounts, Warrant Value and commissions associated with the transaction (a "Below Market Offering"); (x) an "at-the-market" offering of Common Stock or securities convertible into or exchangeable for Common Stock pursuant to Rule 415(a)(4) under the Securities Act (an "ATM"); (y) the implementation by the Company of any mechanism in respect of any securities convertible into or exchangeable for Common Stock for the reset of the purchase price of the Common Stock to below the then Current Market Price of the Common Stock (including, without limitation, any antidilution or similar adjustment provisions in respect of any Company securities, but specifically excluding customary antidilution adjustments for stock splits, stock dividends, stock combinations, recapitalizations, reclassifications and similar events) (a "Price Reset Provision"); or (z) the issuance of options, warrants or similar rights of subscription or the issuance of convertible equity or debt securities, in each case not constituting an Acceptable Transaction. For all purposes of this Agreement, an "Acceptable Transaction" shall mean the issuance by the Company of: (1) debt securities or any class or series of preferred stock of the Company, in each case that are not convertible into or exchangeable for Common Stock or securities convertible into or exchangeable for Common Stock; (2) shares of Common Stock or securities convertible into or exchangeable for Common Stock other than in connection with a Below Market Offering or an ATM, and the issuance of shares of Common Stock upon the conversion, exercise or exchange thereof; (3) shares of Common Stock or securities convertible into or exchangeable for Common Stock in connection with an underwritten public offering of equity securities of the Company or a registered direct public offering of equity securities of the Company, in each case where the price per share of such Common Stock (or the conversion or exercise price of such securities, as applicable) is fixed concurrently with the execution of definitive documentation relating to such offering, and the issuance of shares of Common Stock upon the conversion, exercise or exchange thereof; (4) shares of Common Stock or securities convertible into or exchangeable for Common Stock in connection with awards under the Company's benefit and equity plans and arrangements or shareholder rights plan (as applicable), and the issuance of shares of Common Stock upon the conversion, exercise or exchange thereof; (5) shares of Common Stock issuable upon the conversion, exercise or exchange of equity awards or convertible, exercisable or exchangeable securities outstanding as of the Closing Date; (6) shares of Common Stock in connection with stock splits, stock dividends, stock combinations, recapitalizations, reclassifications and similar events; (7) shares of Common Stock or securities convertible into or exercisable or exchangeable for Common Stock issued in connection with the acquisition, license or sale of one or more other companies, equipment, technologies, other assets or lines of business, and the issuance of shares of Common Stock upon the conversion, exercise or exchange thereof; (8) shares of Common Stock or securities convertible into or exercisable or exchangeable for Common Stock or similar rights to subscribe for the purchase of shares of Common Stock in connection with technology sharing, collaboration, partnering, licensing, research and joint development agreements (or amendments thereto) with third parties, and the issuance of shares of Common Stock upon the conversion, exercise or exchange thereof; (9) shares of Common Stock or securities convertible into or exchangeable for Common Stock to employees, consultants and/or advisors as consideration for services rendered or to be rendered, and the issuance of shares of Common Stock upon the conversion, exercise or exchange thereof; and (10) shares of Common Stock or securities convertible into or exchangeable for Common Stock issued in connection with capital or equipment financings and/or real property lease arrangements, and the issuance of shares of Common Stock upon the conversion, exercise or exchange thereof.

Section 6.8. <u>Corporate Existence</u>. The Company shall take all steps necessary to preserve and continue the corporate existence of the Company; <u>provided</u>, <u>however</u>, that, except as provided in Section 6.9, nothing in this Agreement shall be deemed to prohibit the Company from engaging in any Fundamental Transaction with another Person. For the avoidance of doubt, nothing in this Section 6.8 shall in any way limit the Company's right to terminate this Agreement in accordance with Section 8.1 (subject in all cases to Section 8.3).

Section 6.9. <u>Fundamental Transaction</u>. If a Draw Down Notice has been delivered to the Investor and the transactions contemplated therein have not yet been fully settled in accordance with the terms and conditions of this Agreement, the Company shall not effect any Fundamental Transaction until the expiration of five Trading Days following the Settlement Date with respect to such Draw Down Notice.

Section 6.10. Delivery of Registration Statement and Prospectus; Subsequent Changes. In accordance with the Registration Rights Agreement, the Company shall deliver or make available to the Investor and its counsel, without charge, an electronic copy of the Registration Statement, the Prospectus and all amendments and supplements to the Registration Statement or Prospectus that are filed with the Commission during any period in which a Prospectus (or in lieu thereof, the notice referred to in Rule 173(a) under the Securities Act) is required by the Securities Act to be delivered in connection with resales of the Registrable Securities, in each case as soon as reasonably practicable after the filing thereof with the Commission. The Company shall provide the Investor a reasonable opportunity to comment on a draft of each such document and shall give due consideration to all such comments. The Company consents to the use of the Prospectus (and of any Prospectus Supplement thereto) in accordance with the provisions of the Securities Act and with the securities or "Blue Sky" laws of the jurisdictions in which the Registrable Securities may be sold by the Investor, in connection with the resale of the Registrable Securities and for such period of time thereafter as the Prospectus (or in lieu thereof, the notice referred to in Rule 173(a) under the Securities Act) is required by the Securities Act to be delivered in connection with resales of the Registrable Securities. If during such period of time any event shall occur that in the reasonable judgment of the Company and its counsel is required to be set forth in the Registration Statement, the Prospectus or any Prospectus Supplement or should be set forth therein in order to make the statements made therein (in the case of the Prospectus or any Prospectus or any Prospectus Supplement or comply with the Securities Act or any other applicable law or regulation, the Company shall forthwith (i) notify the Investor to suspend the resale of Registrable Securities during su

Section 6.11. Amendments to the Registration Statement; Prospectus Supplements. Except as provided in this Agreement and other than periodic reports required to be filed pursuant to the Exchange Act, the Company shall not file with the Commission any amendment to the Registration Statement that relates to the Investor, the Transaction Documents or the transactions contemplated thereby or file with the Commission any Prospectus Supplement that relates to the Investor, the Transaction Documents or the transactions contemplated thereby with respect to which (a) the Investor shall not previously have been advised, (b) the Company shall not have given due consideration to any comments thereon received from the Investor or its counsel, or (c) the Investor shall reasonably object after being so advised, unless it is necessary to amend the Registration Statement or make any supplement to the Prospectus to comply with the Securities Act or any other applicable law or regulation, in which case the Company shall expeditiously furnish to the Investor shall be provided with a reasonable opportunity to review and comment upon any disclosure relating to the Investor and the Company shall expeditiously furnish to the Investor an electronic copy thereof. In addition, for so long as, in the reasonable opinion of counsel for the Investor, the Prospectus (or in lieu thereof, the notice referred to in Rule 173(a) under the Securities Act) is required to be delivered in connection with any sales of Registrable Securities by the Investor, the Company shall not file any Prospectus Supplement without delivering or making available a copy of such Prospectus Supplement to the Investor promptly. Upon receipt of amendment to the Registration Statement or Prospectus Supplement from the Company or its counsel, the Investor shall promptly review such document and provide comments to the Company or its counsel regarding such document, if any, within a reasonable period of time.

Section 6.12. Stop Orders. The Company shall notify the Investor as soon as possible (but in no event later than 24 hours), and confirm in writing, upon its becoming aware of the occurrence of any of the following events in respect of the Registration Statement or related Prospectus Supplement relating to an offering of Registrable Securities: (i) receipt of any request by the Commission or any other federal or state governmental authority for any additional information relating to the Registration Statement, the Prospectus or any Prospectus Supplement, or for any amendment of or supplement to the Registration Statement, the Prospectus, or any Prospectus Supplement; (ii) the issuance by the Commission or any other federal or state governmental authority of any stop order suspending the effectiveness of the Registration Statement or prohibiting or suspending the use of the Prospectus or any Prospectus Supplement, or of the suspension of qualification or exemption from qualification of the Securities for offering or sale in any jurisdiction, or the initiation or contemplated initiation of any proceeding for such purpose; and (iii) any event or the existence of any condition or state of facts, which makes any statement of a material fact made in the Registration Statement, the Prospectus or any Prospectus Supplement untrue or which requires the making of any additions to or changes to the statements then made in the Registration Statement, the Prospectus or any Prospectus Supplement in order to state a material fact required by the Securities Act to be stated therein or necessary in order to make the statements then made therein (in the case of the Prospectus or any Prospectus Supplement, in light of the circumstances under which they were made) not misleading, or which requires an amendment to the Registration Statement or a supplement to the Prospectus or any Prospectus Supplement to comply with the Securities Act or any other law (other than the transactions contemplated by any Draw Down Notice and the settlement thereof). The Company shall not be required to disclose to the Investor the substance or specific reasons of any of the events set forth in clauses (i) through (iii) of the immediately preceding sentence, but rather, shall only be required to disclose that the event has occurred. The Company shall not issue any Draw Down during the continuation of any of the foregoing events. If at any time the Commission or any other federal or state governmental authority shall issue any stop order suspending the effectiveness of the Registration Statement or prohibiting or suspending the use of the Prospectus or any Prospectus Supplement, the Company shall use commercially reasonable efforts to obtain the withdrawal of such order at the earliest possible time.

Section 6.13. Selling Restrictions. (i) Except as expressly set forth below, the Investor covenants that from and after the Closing Date through and including the Trading Day next following the expiration or termination of this Agreement (the "Restricted Period"), neither the Investor nor any of its Affiliates nor any entity managed or controlled by the Investor (collectively, the "Restricted Persons" and each of the foregoing is referred to herein as a "Restricted Person") shall, directly or indirectly, (x) engage in any Short Sales involving the Company's securities or (y) grant any option to purchase, or acquire any right to dispose of or otherwise dispose for value of, any shares of Common Stock or any securities convertible into or exercisable or exchangeable for any shares of Common Stock, or enter into any swap, hedge or other similar agreement that transfers, in whole or in part, the economic risk of ownership of the Common Stock. Notwithstanding the foregoing, it is expressly understood and agreed that nothing contained herein shall (without implication that the contrary would otherwise be true) prohibit any Restricted Person during the Restricted Period from: (1) selling "long" (as defined under Rule 200 promulgated under Regulation SHO) the Securities; or (2) selling a number of shares of Common Stock equal to the number of Shares that such Restricted Person is or may be obligated to purchase under a pending Draw Down Notice but has not yet taken possession of so long as such Restricted Person (or the Broker-Dealer, as applicable) delivers the Shares purchased pursuant to such Draw Down Notice to the purchaser thereof or the applicable Broker-Dealer; as applicable) shall not be required to so deliver any such Shares subject to such Draw Down Notice if the Company fails for any reason to deliver such Shares to the Investor on the applicable Settlement Date upon the terms and subject to the provisions of this Agreement.

(ii) In addition to the foregoing, in connection with any sale of Securities (including any sale permitted by paragraph (i) above), the Investor shall comply in all respects with all applicable laws, rules, regulations and orders, including, without limitation, the requirements of the Securities Act and the Exchange Act.

Section 6.14. <u>Effective Registration Statement</u>. During the Investment Period, the Company shall use its best efforts to maintain the continuous effectiveness of the Registration Statement under the Securities Act.

Section 6.15. <u>Blue Sky.</u> The Company shall take such action, if any, as is necessary in order to obtain an exemption for or to qualify the Securities for issuance and sale to the Investor pursuant to the Transaction Documents, at the request of the Investor, and the subsequent resale of Registrable Securities by the Investor, in each case, under applicable state securities or "Blue Sky" laws and shall provide evidence of any such action so taken to the Investor from time to time following the Closing Date; <u>provided, however</u>, that the Company shall not be required in connection therewith or as a condition thereto to (x) qualify to do business in any jurisdiction where it would not otherwise be required to qualify but for this Section 6.15, (y) subject itself to general taxation in any such jurisdiction, or (z) file a general consent to service of process in any such jurisdiction.

Section 6.16. Non-Public Information. Neither the Company or any of its Subsidiaries, nor any of their respective directors, officers, employees or agents shall disclose any material non-public information about the Company to the Investor, unless a simultaneous public announcement thereof is made by the Company in the manner contemplated by Regulation FD. In the event of a breach of the foregoing covenant by the Company or any of its Subsidiaries, or any of their respective directors, officers, employees and agents (as determined in the reasonable good faith judgment of the Investor), (i) the Investor shall promptly provide written notice of such breach to the Company and (ii) after such notice has been provided to the Company and in addition to any other remedy provided herein or in the other Transaction Documents, the Investor shall have the Company, any of its Subsidiaries, or any of their respective directors, officers, employees or agents; provided that the Company shall have failed to publicly disclose such material, non-public information within 24 hours following demand therefor by the Investor. The Investor shall not have any liability to the Company, any of its Subsidiaries, or any of their respective directors, officers, employees, stockholders or agents, for any such disclosure.

Section 6.17. <u>Broker/Dealer</u>. The Investor shall use one or more broker-dealers to effectuate all sales, if any, of Securities that it may purchase or otherwise acquire from the Company pursuant to the Transaction Documents, as applicable, which (or whom) shall be unaffiliated with the Investor and not then currently engaged or used by the Company (collectively, the "*Broker-Dealer*"). The Investor shall, from time to time, provide the Company with all information regarding the Broker-Dealer reasonably requested by the Company. The Investor shall be solely responsible for all fees and commissions of the Broker-Dealer, which shall not exceed customary brokerage fees and commissions.

Section 6.18. <u>Disclosure Schedule.</u>

- (i) The Company may, from time to time, update the Disclosure Schedule as may be required to satisfy the condition set forth in Section 7.2(i). For purposes of this Section 6.18, any disclosure made in a schedule to the Compliance Certificate substantially in the form attached hereto as Exhibit D shall be deemed to be an update of the Disclosure Schedule. Notwithstanding anything in this Agreement to the contrary, no update to the Disclosure Schedule pursuant to this Section 6.18 shall cure any breach of a representation or warranty of the Company contained in this Agreement and made prior to the update and shall not affect any of the Investor's rights or remedies with respect thereto.
- (ii) Notwithstanding anything to the contrary contained in the Disclosure Schedule or in this Agreement, the information and disclosure contained in any Schedule of the Disclosure Schedule shall be deemed to be disclosed and incorporated by reference in any other Schedule of the Disclosure Schedule as though fully set forth in such Schedule for which applicability of such information and disclosure is readily apparent on its face. The fact that any item of information is disclosed in the Disclosure Schedule shall not be construed to mean that such information is required to be disclosed by this Agreement. Except as expressly set forth in this Agreement, such information and the thresholds (whether based on quantity, qualitative characterization, dollar amounts or otherwise) set forth herein shall not be used as a basis for interpreting the terms "material" or "Material Adverse Effect" or other similar terms in this Agreement.

ARTICLE VII CONDITIONS TO CLOSING AND CONDITIONS TO THE SALE AND PURCHASE OF THE SHARES

Section 7.1. Conditions Precedent to Closing. The Closing is subject to the satisfaction of each of the conditions set forth in this Section 7.1.

- (i) Accuracy of the Investor's Representations and Warranties. The representations and warranties of the Investor contained in this Agreement (a) that are not qualified by "materiality" shall be true and correct in all material respects as of the Closing Date, except to the extent such representations and warranties are as of another date, in which case, such representations and warranties shall be true and correct as of the Closing Date, except to the extent such representations and warranties warranties are as of another date, in which case, such representations and warranties shall be true and correct as of the Closing Date, except to the extent such representations and warranties are as of another date, in which case, such representations and warranties shall be true and correct as of such other date.
- (ii) Accuracy of the Company's Representations and Warranties. The representations and warranties of the Company contained in this Agreement (a) that are not qualified by "materiality" or "Material Adverse Effect" shall be true and correct in all material respects as of the Closing Date, except to the extent such representations and warranties are as of another date, in which case, such representations and warranties shall be true and correct in all material respects as of such other date and (b) that are qualified by "materiality" or "Material Adverse Effect" shall be true and correct as of the Closing Date, except to the extent such representations and warranties are as of another date, in which case, such representations and warranties shall be true and correct as of such other date.
- (iii) Payment of Document Preparation Fee; Issuance of Initial Commitment Shares. Prior to the Closing, the Company shall have paid by wire transfer of immediately available funds to an account designated by the Investor's counsel, the total Document Preparation Fee in accordance with Section 10.1(i) hereof, all of which fees shall be non-refundable when paid regardless of whether any Draw Downs are issued by the Company or settled hereunder. On the Closing Date, the Company shall deliver irrevocable instructions to its transfer agent to issue to the Investor, not later than 4:00 p.m. (New York City time) on the second Trading Day immediately following the Closing Date, a certificate representing the Initial Commitment Shares in the name of the Investor or its designee (in which case such designee name shall have been provided to the Company prior to the Closing Date), in consideration for the Investor's execution and delivery of this Agreement. For the avoidance of doubt, all of the Initial Commitment Shares shall be fully earned as of the Closing Date regardless of whether any Draw Downs are issued by the Company or settled hereunder.

- (iv) <u>Closing Deliverables</u>. At the Closing, counterpart signature pages of this Agreement and the Registration Rights Agreement executed by each of the parties hereto shall be delivered as provided in Section 2.2. Simultaneously with the execution and delivery of this Agreement and the Registration Rights Agreement, the Investor's counsel shall have received (1) an opinion of outside counsel to the Company, dated the Closing Date, in the form mutually agreed to by the parties hereto, (2) a certificate from the Company, dated the Closing Date, in the form of <u>Exhibit C</u> hereto, and (3) a copy of the irrevocable instructions to the Company's transfer agent regarding the issuance to the Investor of the certificate representing the Initial Commitment Shares.
- Section 7.2. Conditions Precedent to a Draw Down. The right of the Company to deliver a Draw Down Notice and the obligation of the Investor to accept a Draw Down Notice and to acquire and pay for the Shares in accordance therewith is subject to the satisfaction, at each Draw Down Exercise Date and at each Settlement Date (except as otherwise expressly set forth below), of each of the conditions set forth in this Section 7.2.
- (i) Accuracy of the Company's Representations and Warranties. The representations and warranties of the Company contained in this Agreement (a) that are not qualified by "materiality" or "Material Adverse Effect" shall have been true and correct in all material respects when made and shall be true and correct in all material respects as of the applicable Draw Down Exercise Date and the applicable Settlement Date with the same force and effect as if made on such dates, except to the extent such representations and warranties are as of another date, in which case, such representations and warranties shall be true and correct in all material respects as of such other date and (b) that are qualified by "materiality" or "Material Adverse Effect" shall have been true and correct when made and shall be true and correct as of the applicable Draw Down Exercise Date and the applicable Settlement Date with the same force and effect as if made on such dates, except to the extent such representations and warranties are as of another date, in which case, such representations and warranties shall be true and correct as of such other date.
- (ii) <u>Performance of the Company</u>. The Company shall have performed, satisfied and complied in all material respects with all covenants, agreements and conditions required by this Agreement and the Registration Rights Agreement to be performed, satisfied or complied with by the Company at or prior to the applicable Draw Down Exercise Date and the applicable Settlement Date. The Company shall have delivered to the Investor on the applicable Settlement Date the Compliance Certificate substantially in the form attached hereto as <u>Exhibit D.</u>
- (iii) Registration Statement Effective. The Registration Statement covering the resale by the Investor of the Registrable Securities shall have been declared effective under the Securities Act by the Commission and shall remain effective, and the Investor shall be permitted to utilize the Prospectus therein to resell (a) all of the Initial Commitment Shares issuable pursuant to this Agreement, (b) all of the Shares issuable pursuant to the applicable Draw Down Notice.

(iv) No Material Notices. None of the following events shall have occurred and be continuing: (a) receipt of any request by the Commission or any other federal or state governmental authority for any additional information relating to the Registration Statement, the Prospectus or any Prospectus Supplement, or for any amendment of or supplement to the Registration Statement, the Prospectus, or any Prospectus Supplement; (b) the issuance by the Commission or any other federal or state governmental authority of any stop order suspending the effectiveness of the Registration Statement or prohibiting or suspending the use of the Prospectus or any Prospectus Supplement, or of the suspension of qualification or exemption from qualification of the Securities for offering or sale in any jurisdiction, or the initiation or contemplated initiation of any proceeding for such purpose; or (c) the occurrence of any event or the existence of any condition or state of facts, which makes any statement of a material fact made in the Registration Statement, the Prospectus or any Prospectus Supplement untrue or which requires the making of any additions to or changes to the statements then made in the Registration Statement, the Prospectus or any Prospectus or any Prospectus Supplement in order to state a material fact required by the Securities Act to be stated therein or necessary in order to make the statements then made therein (in the case of the Prospectus or any Prospectus Supplement to the Registration Statement or a supplement to the Prospectus or any Prospectus Supplement to the Prospectus One Notice and the settlement thereof). The Company shall have no Knowledge of any event that could reasonably be expected to have the effect of causing the suspension of the effectiveness of the Registration Statement or the prohibition or suspension of the use of the Prospectus or any Prospectus Supplement in connection with the resale of the Registrable Securities by the Investor.

(v) Other Commission Filings. The Current Report and the Form D shall have been filed with the Commission as required pursuant to Section 2.3, and the final Prospectus and all other Prospectus Supplements required to have been filed with the Commission pursuant to Section 2.3 and pursuant to the Registration Rights Agreement shall have been filed with the Commission in accordance with Section 2.3 and the Registration Rights Agreement. All reports, schedules, registrations, forms, statements, information and other documents required to have been filed by the Company with the Commission pursuant to the reporting requirements of the Exchange Act, including all material required to have been filed pursuant to Section 13(a) or 15(d) of the Exchange Act, shall have been filed with the Commission and, if any Registrable Securities are covered by a Registration Statement on Form S-3, such filings shall have been made within the applicable time period prescribed for such filing under the Exchange Act.

(vi) No Suspension of Trading in or Notice of Delisting of Common Stock. Trading in the Common Stock shall not have been suspended by the Commission, the Trading Market or the FINRA (except for any suspension of trading of limited duration agreed to by the Company, which suspension shall be terminated prior to the applicable Draw Down Exercise Date), the Company shall not have received any final and non-appealable notice that the listing or quotation of the Common Stock on the Trading Market shall be terminated on a date certain (unless, prior to such date certain, the Common Stock is listed or quoted on any other Trading Market), and trading in securities generally as reported on the Trading Market shall not have been suspended or limited, nor shall a banking moratorium have been declared either by the U.S. or New York State authorities (except for any suspension, limitation or moratorium which shall be terminated prior to the applicable Draw Down Exercise Date), nor shall there have occurred any material outbreak or escalation of hostilities or other national or international calamity or crisis that has had or would reasonably be expected to have a material adverse change in any U.S. financial, credit or securities market that is continuing as of the applicable Draw Down Exercise Date.

- (vii) Compliance with Laws. The Company shall have complied with all applicable federal, state and local governmental laws, rules, regulations and ordinances in connection with the execution, delivery and performance of this Agreement and the other Transaction Documents to which it is a party and the consummation of the transactions contemplated hereby and thereby, including, without limitation, the Company shall have obtained all permits and qualifications required by any applicable state securities or "Blue Sky" laws for the offer and sale of the Securities by the Company to the Investor and the subsequent resale of the Registrable Securities by the Investor (or shall have the availability of exemptions therefrom).
- (viii) No Injunction. No statute, regulation, order, decree, writ, ruling or injunction shall have been enacted, entered, promulgated, threatened or endorsed by any court or governmental authority of competent jurisdiction which prohibits the consummation of or which would materially modify or delay any of the transactions contemplated by the Transaction Documents.
- (ix) No Proceedings or Litigation. No action, suit or proceeding before any arbitrator or any court or governmental authority shall have been commenced or threatened, and no inquiry or investigation by any governmental authority shall have been commenced or threatened, against the Company or any Subsidiary, or any of the officers, directors or affiliates of the Company or any Subsidiary, seeking to restrain, prevent or change the transactions contemplated by the Transaction Documents, or seeking material damages in connection with such transactions.
 - (x) Aggregate Limit. The issuance and sale of the Shares issuable pursuant to such Draw Down Notice shall not violate Sections 3.1, 3.7, 3.9 and 6.6 hereof.
- (xi) <u>Securities Authorized and Delivered</u>. The Shares issuable pursuant to such Draw Down Notice shall have been duly authorized by all necessary corporate action of the Company. The Company shall have delivered all Shares relating to all prior Draw Down Notices.
- (xii) <u>Listing of Securities</u>. All of the Securities that may be issued pursuant to this Agreement shall have been approved for listing or quotation on the Trading Market as of the Closing Date, subject only to notice of issuance.
- (xiii) No Material Adverse Effect. No condition, occurrence, state of facts or event constituting a Material Adverse Effect shall have occurred and be continuing.
- (xiv) No Restrictive Legends. If requested by the Investor from and after the Effective Date, the Company shall have either (i) issued and delivered (or caused to be issued and delivered) to the Investor a certificate representing the Commitment Shares that is free from all restrictive and other legends or (ii) caused the Company's transfer agent to credit the Investor's or its designee's account at DTC through its Deposit/Withdrawal at Custodian (DWAC) system with a number of shares of Common Stock equal to the number of Commitment Shares represented by the certificate delivered by the Investor to the Company in accordance with Section 10.1(iv) of this Agreement.

(xv) <u>Opinion of Counsel; Bring-Down.</u> Prior to the first Draw Down Exercise Date, the Investor shall have received an opinion from outside counsel to the Company, in the form mutually agreed to by the parties hereto. On each Settlement Date, the Investor shall have received an opinion "bring down" from outside counsel to the Company, dated the applicable Settlement Date, in the form mutually agreed to by the parties hereto.

ARTICLE VIII TERMINATION

Section 8.1. Termination. Unless earlier terminated as provided hereunder, this Agreement shall terminate automatically on the earlier to occur of (i) the first day of the month next following the 36-month anniversary of the Effective Date (it being hereby acknowledged and agreed that such term may not be extended by the parties hereto) and (ii) the date on which the Investor shall have purchased or acquired shares of Common Stock pursuant to this Agreement equal to the Aggregate Limit. Subject to Section 8.3, the Company may terminate this Agreement effective upon one Trading Day's prior written notice to the Investor in accordance with Section 10.4; provided, however, that (A) the Company shall have paid all fees and amounts and issued all Commitment Shares owed to the Investor or its counsel, as applicable, pursuant to Section 10.1 of this Agreement, prior to such termination, and (B) prior to issuing any press release, or making any public statement or announcement, with respect to such termination, the Company shall consult with the Investor and shall obtain the Investor's consent to the form and substance of such press release or other disclosure, which consent shall not be unreasonably delayed or withheld. Subject to Section 8.3, this Agreement may be terminated at any time by the mutual written consent of the parties, effective as of the date of such mutual written consent unless otherwise provided in such written consent.

Other Termination. Subject to Section 8.3, the Investor shall have the right to terminate this Agreement effective upon one Trading Day's prior written notice to the Company in accordance with Section 10.4, if: (i) any condition, occurrence, state of facts or event constituting a Material Adverse Effect has occurred and is continuing; (ii) the Company shall have entered into any agreement, plan, arrangement or transaction with a third party or shall have determined to utilize any existing agreement, plan or arrangement with a third party, in each case the principal purpose of which is to implement, effect or consummate at any time during the Investment Period a Similar Financing, an ATM or a Price Reset Provision; (iii) a Fundamental Transaction shall have occurred; (iv) the Registration Statement is not filed by the Filing Deadline (as defined in the Registration Rights Agreement) or declared effective by the Effectiveness Deadline (as defined in the Registration Rights Agreement), or the Company is otherwise in breach or default in any material respect under any of the other provisions of the Registration Rights Agreement, and, if such failure, breach or default is capable of being cured, such failure, breach or default is not cured within 10 Trading Days after notice of such failure, breach or default is delivered to the Company pursuant to Section 10.4; (v) while the Registration Statement is required to be maintained effective pursuant to the terms of the Registration Rights Agreement and the Investor holds any Registrable Securities, the effectiveness of the Registration Statement lapses for any reason (including, without limitation, the issuance of a stop order) or the Registration Statement, the Prospectus or any Prospectus Supplement is otherwise unavailable to the Investor for the resale of all of the Registrable Securities in accordance with the terms of the Registration Rights Agreement, and such lapse or unavailability continues for a period of 20 consecutive Trading Days or for more than an aggregate of 60 Trading Days in any 365-day period, other than due to acts of the Investor; (vi) trading in the Common Stock on the Trading Market shall have been suspended or the Common Stock shall have failed to be listed or quoted on a Trading Market, and such suspension or failure continues for a period of 20 consecutive Trading Days or for more than an aggregate of 60 Trading Days in any 365-day period; (vii) the Company has filed for and/or is subject to any bankruptcy, insolvency, reorganization or liquidation proceedings or other proceedings for relief under any bankruptcy law or any law for the relief of debtors instituted by or against the Company or (viii) the Company is in material breach or default of this Agreement, and, if such breach or default is capable of being cured, such breach or default is not cured within 10 Trading Days after notice of such breach or default is delivered to the Company pursuant to Section 10.4. Unless notification thereof is required elsewhere in this Agreement (in which case such notification shall be provided in accordance with such other provision), the Company shall promptly (but in no event later than 24 hours) notify the Investor (and, if required under applicable law, including, without limitation, Regulation FD promulgated by the Commission, or under the applicable rules and regulations of the Trading Market, the Company shall publicly disclose such information in accordance with Regulation FD and the applicable rules and regulations of the Trading Market) upon becoming aware of any of the events set forth in the immediately preceding sentence.

Effect of Termination. In the event of termination by the Company or the Investor pursuant to Section 8.1 or 8.2, as applicable, written notice thereof shall forthwith be given to the other party as provided in Section 10.4 and the transactions contemplated by this Agreement shall be terminated without further action by either party. If this Agreement is terminated as provided in Section 8.1 or 8.2 herein, this Agreement shall become void and of no further force and effect, except that (i) the provisions of Article V (Representations and Warranties of the Company), Article IX (Indemnification), Article X (Miscellaneous) (excluding Section 10.1(v)) and this Article VIII (Termination) shall remain in full force and effect indefinitely notwithstanding such termination, and, (ii) so long as the Investor owns any Securities, the covenants and agreements of the Company contained in Article VI (Additional Covenants) shall remain in full force and notwithstanding such termination for a period of six months following such termination. Notwithstanding anything in this Agreement to the contrary, no termination of this Agreement by any party shall (i) become effective prior to the first Trading Day immediately following the Settlement Date related to any pending Draw Down Notice that has not been fully settled in accordance with the terms and conditions of this Agreement (it being hereby acknowledged and agreed that no termination of this Agreement shall limit, alter, modify, change or otherwise affect any of the Company's or the Investor's rights or obligations under the Transaction Documents with respect to any pending Draw Down, and that the parties shall fully perform their respective obligations with respect to any such pending Draw Down under the Transaction Documents, provided all of the conditions to the settlement thereof set forth in Article VII are timely satisfied), (ii) limit, alter, modify, change or otherwise affect the Company's or the Investor's rights or obligations under the Registration Rights Agreement, all of which shall survive any such termination, (iii) affect any Initial Commitment Shares previously issued or delivered, or any rights of any holder thereof (it being hereby acknowledged and agreed that all of the Commitment Shares shall be fully earned as of the Closing Date, regardless of whether any Draw Downs are issued by the Company or settled hereunder), or (iv) affect any cash fees paid to the Investor or its counsel pursuant to Section 10.1 (including, without limitation, the Document Preparation Fee), in each case all of which fees shall be non-refundable when paid regardless of whether any Draw Downs are issued by the Company or settled hereunder. Nothing in this Section 8.3 shall be deemed to release the Company or the Investor from any liability for any breach or default under this Agreement or any of the other Transaction Documents to which it is a party, or to impair the rights of the Company and the Investor to compel specific performance by the other party of its obligations under the Transaction Documents to which it is a party.

ARTICLE IX INDEMNIFICATION

Indemnification of Investor. In consideration of the Investor's execution and delivery of this Agreement and acquiring the Shares hereunder and in addition to all of the Company's other obligations under the Transaction Documents to which it is a party, subject to the provisions of this Section 9.1, the Company shall indemnify and hold harmless the Investor, each of its directors, officers, shareholders, members, partners, employees, representatives, agents and advisors (and any other Persons with a functionally equivalent role of a Person holding such titles notwithstanding the lack of such title or any other title), each Person, if any, who controls the Investor (within the meaning of Section 15 of the Securities Act or Section 20(a) of the Exchange Act), and the respective directors, officers, shareholders, members, partners, employees, representatives, agents and advisors (and any other Persons with a functionally equivalent role of a Person holding such titles notwithstanding the lack of such title or any other title) of such controlling Persons (each, an "Investor Party"), from and against all losses, liabilities, obligations, claims, contingencies, damages, costs and expenses (including all judgments, amounts paid in settlement, court costs, reasonable attorneys' fees and costs of defense and investigation) (collectively, "Damages") that any Investor Party may suffer or incur as a result of or relating to (a) any breach of any of the representations, warranties, covenants or agreements made by the Company in this Agreement or in the other Transaction Documents to which it is a party or (b) any action, suit, claim or proceeding (including for these purposes a derivative action brought on behalf of the Company) instituted against such Investor Party arising out of or resulting from the execution, delivery, performance or enforcement of the Transaction Documents, other than claims for indemnification within the scope of Section 6 of the Registration Rights Agreement; provided, however, that (x) the foregoing indemnity shall not apply to any Damages to the extent, but only to the extent, that such Damages resulted directly and primarily from a breach of any of the Investor's representations, warranties, covenants or agreements contained in this Agreement or the Registration Rights Agreement, and (y) the Company shall not be liable under subsection (b) of this Section 9.1 to the extent, but only to the extent, that a court of competent jurisdiction shall have determined by a final judgment (from which no further appeals are available) that such Damages resulted directly and primarily from any acts or failures to act, undertaken or omitted to be taken by such Investor Party through its fraud, bad faith, gross negligence, or willful or reckless misconduct.

The Company shall reimburse any Investor Party promptly upon demand (with accompanying presentation of documentary evidence) for all legal and other costs and expenses reasonably incurred by such Investor Party in connection with (i) any action, suit, claim or proceeding, whether at law or in equity, to enforce compliance by the Company with any provision of the Transaction Documents or (ii) any other any action, suit, claim or proceeding, whether at law or in equity, with respect to which it is entitled to indemnification under this Section 9.1; provided that the Investor shall promptly reimburse the Company for all such legal and other costs and expenses to the extent a court of competent jurisdiction determines that any Investor Party was not entitled to such reimbursement.

An Investor Party's right to indemnification or other remedies based upon the representations, warranties, covenants and agreements of the Company set forth in the Transaction Documents shall not in any way be affected by any investigation or knowledge of such Investor Party. Such representations, warranties, covenants and agreements shall not be affected or deemed waived by reason of the fact that an Investor Party knew or should have known that any representation or warranty might be inaccurate or that the Company failed to comply with any agreement or covenant. Any investigation by such Investor Party shall be for its own protection only and shall not affect or impair any right or remedy hereunder.

To the extent that the foregoing undertakings by the Company set forth in this Section 9.1 may be unenforceable for any reason, the Company shall make the maximum contribution to the payment and satisfaction of each of the Damages which is permissible under applicable law.

Section 9.2. <u>Indemnification Procedures</u>. Promptly after an Investor Party receives notice of a claim or the commencement of an action for which the Investor Party intends to seek indemnification under Section 9.1, the Investor Party will notify the Company in writing of the claim or commencement of the action, suit or proceeding; <u>provided, however</u>, that failure to notify the Company will not relieve the Company from liability under Section 9.1, except to the extent it has been materially prejudiced by the failure to give notice. The Company will be entitled to participate in the defense of any claim, action, suit or proceeding as to which indemnification is being sought, and if the Company acknowledges in writing the obligation to indemnify the Investor Party against whom the claim or action is brought, the Company may (but will not be required to) assume the defense against the claim, action, suit or proceeding with counsel satisfactory to it. After the Company notifies the Investor Party that the Company will not be liable for any further legal or other expenses incurred by the Investor Party in connection with the defense against the claim, action, suit or proceeding, the Company will not be liable for any further legal or other expenses incurred by the Investor Party in connection with the defense against the claim, action, suit or proceeding except that if, in the opinion of counsel to the Investor Party, it would be inappropriate under the applicable rules of professional responsibility for the same counsel to represent both the Company and such Investor Party. In such event, the Company will pay the reasonable fees and expenses of no more than one separate counsel for all such Investor Parties promptly as such fees and expenses are incurred. Each Investor Party, as a condition to receiving indemnification as provided in Section 9.1, will cooperate in all reasonable respects with the Company in the defense of any action or claim as to which indemnification is sought. The Company will not be liable for

The remedies provided for in this Article IX are not exclusive and shall not limit any rights or remedies which may otherwise be available to any Investor Party at law or in equity.

ARTICLE X MISCELLANEOUS

Section 10.1. Fees and Expenses.

- (i) <u>Counsel and Other Fees and Expenses</u>. Each party shall bear its own fees and expenses related to the transactions contemplated by this Agreement; <u>provided, however</u>, that the Company shall pay directly to the Investor's counsel a non-accountable and non-refundable document preparation fee in the aggregate amount of \$15,000 (collectively, the "*Document Preparation Fee*"), by wire transfer of immediately available funds to an account designated by the Investor's counsel, as follows: (a) \$5,000 of such fee shall be paid prior to the Closing (the Investor hereby acknowledges that \$5,000 of the Document Preparation Fee was paid by the Company to the Investor's counsel on March x, 2013); and (b) \$10,000 of such fee shall be paid upon execution of this Agreement. For the avoidance of doubt, the Document Preparation Fee (and any portion thereof) shall be non-refundable when paid, regardless of whether any Draw Downs are issued by the Company or settled hereunder. The Company shall pay all U.S. federal, state and local stamp and other similar transfer and other taxes and duties levied in connection with issuance of the Securities pursuant hereto.
- (ii) <u>Commitment Shares</u>. In consideration for the Investor's execution and delivery of this Agreement, concurrently with the execution and delivery of this Agreement on the Closing Date, the Company shall deliver irrevocable instructions to its transfer agent to issue to the Investor, not later than 4:00 p.m. (New York City time) on the second Trading Day immediately following the Closing Date, a certificate representing the Initial Commitment Shares in the name of the Investor or its designee (in which case such designee name shall have been provided to the Company prior to the Closing Date). Such certificate shall be delivered to the Investor by overnight courier at its address set forth in Section 10.4 hereof. For the avoidance of doubt, all of the Initial Commitment Shares shall be fully earned as of the Closing Date, regardless of whether any Draw Downs are issued by the Company or settled hereunder. Upon issuance, the Initial Commitment Shares shall constitute "restricted securities" as such term is defined in Rule 144(a)(3) under the Securities Act and, subject to the provisions of subsection (iv) of this Section 10.1, the certificates representing the Initial Commitment Shares shall be included in the Registration Statement in accordance with the terms of the Registration Rights Agreement.
- (iii) <u>Legends</u>. The certificate(s) representing the Commitment Shares issued prior to the Effective Date, except as set forth below, shall bear a restrictive legend in substantially the following form (and a stop-transfer order may be placed against transfer of such stock certificate(s)):

THE OFFER AND SALE OF THE SECURITIES REPRESENTED BY THIS CERTIFICATE HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED, OR APPLICABLE STATE SECURITIES LAWS. THE SECURITIES MAY NOT BE OFFERED FOR SALE, SOLD, TRANSFERRED OR ASSIGNED (I) IN THE ABSENCE OF (A) AN EFFECTIVE REGISTRATION STATEMENT FOR THE SECURITIES UNDER THE SECURITIES ACT OF 1933, AS AMENDED, OR (B) AN OPINION OF COUNSEL TO THE HOLDER (IF REQUESTED BY THE COMPANY), IN A FORM REASONABLY ACCEPTABLE TO THE COMPANY, THAT REGISTRATION IS NOT REQUIRED UNDER SAID ACT OR (II) UNLESS SOLD OR ELIGIBLE TO BE SOLD PURSUANT TO RULE 144 UNDER SAID ACT. NOTWITHSTANDING THE FOREGOING, THE SECURITIES MAY BE PLEDGED IN CONNECTION WITH A BONA FIDE MARGIN ACCOUNT OR OTHER LOAN OR FINANCING ARRANGEMENT SECURED BY THE SECURITIES.

Notwithstanding the foregoing and for the avoidance of doubt, all Shares to be issued in respect of any Draw Down Notice delivered to the Investor pursuant to this Agreement shall be issued to the Investor in accordance with Section 3.5 by crediting the Investor's or its designees' account at DTC through its Deposit/Withdrawal at Custodian (DWAC) system, and all such Shares shall be freely tradable and transferable and without restriction on resale (and no stop-transfer order shall be placed against transfer thereof), and the Company shall not take any action or give instructions to any transfer agent of the Company otherwise.

(iv) Removal of Legend. From and after the Effective Date, the Company shall, no later than two Trading Days following the delivery by the Investor to the Company or the Company's transfer agent (with notice to the Company) of a legended certificate representing the Commitment Shares (endorsed or with stock powers attached, signatures guaranteed, and otherwise in form necessary to affect the reissuance and/or transfer, if applicable), as directed by the Investor, either: (A) issue and deliver (or cause to be issued and delivered) to the Investor a certificate representing such Commitment Shares that is free from all restrictive and other legends or (B) cause the Company's transfer agent to credit the Investor's or its designee's account at DTC through its Deposit/Withdrawal at Custodian (DWAC) system with a number of shares of Common Stock equal to the number of Commitment Shares represented by the certificate so delivered by the Investor (the date by which such certificate is required to be delivered to the Investor or such credit is so required to be made to the account of the Investor or its designee at DTC pursuant to the foregoing is referred to herein as the "Required Delivery Date"). If the Company fails on or prior to the Required Delivery Date to either (i) issue and deliver (or cause to be issued and delivered) to the Investor a certificate representing the Commitment Shares that is free from all restrictive and other legends or (ii) cause the Company's transfer agent to credit the balance account of the Investor or its designee at DTC through its Deposit/Withdrawal at Custodian (DWAC) system with a number of Shares of Common Stock equal to the number of Commitment Shares represented by the certificate delivered by the Investor pursuant hereto, then, in addition to all other remedies available to the Investor, the Company shall pay in cash to the Investor on each day after the Required Delivery Date that the issuance or credit of such shares is not timely effected an amount equal to 2.0% of the product of (A) the sum of the number of Commitment Shares not issued to the Investor on a timely basis and to which the Investor is entitled and (B) the VWAP for the five Trading Day period immediately preceding the Required Delivery Date. In addition to the foregoing, if the Company fails to so properly deliver such unlegended certificates or so properly credit the account of the Investor or its designee at DTC by the Required Delivery Date, and if on or after the Required Delivery Date the Investor purchases (in an open market transaction or otherwise) shares of Common Stock to deliver in satisfaction of a sale by the Investor of shares of Common Stock that the Investor anticipated receiving from the Company without any restrictive legend, then the Company shall, within three Trading Days after the Investor's request, pay cash to the Investor in an amount equal to the Investor's total purchase price (including brokerage commissions, if any) for the shares of Common Stock so purchased, at which point the Company's obligation to deliver a certificate or credit such Investor's or its designee's account at DTC for such shares of Common Stock shall terminate and such shares shall be cancelled.

Section 10.2. Specific Enforcement, Consent to Jurisdiction, Waiver of Jury Trial.

(i) The Company and the Investor acknowledge and agree that irreparable damage would occur in the event that any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached. It is accordingly agreed that either party shall be entitled to an injunction or injunctions to prevent or cure breaches of the provisions of this Agreement by the other party and to enforce specifically the terms and provisions hereof (without the necessity of showing economic loss and without any bond or other security being required), this being in addition to any other remedy to which either party may be entitled by law or equity.

(ii) Each of the Company and the Investor (a) hereby irrevocably submits to the jurisdiction of the U.S. District Court and other courts of the United States sitting in the State of New York for the purposes of any suit, action or proceeding arising out of or relating to this Agreement, and (b) hereby waives, and agrees not to assert in any such suit, action or proceeding, any claim that it is not personally subject to the jurisdiction of such court, that the suit, action or proceeding is brought in an inconvenient forum or that the venue of the suit, action or proceeding is improper. Each of the Company and the Investor consents to process being served in any such suit, action or proceeding by mailing a copy thereof to such party at the address in effect for notices to it under this Agreement and agrees that such service shall constitute good and sufficient service of process and notice thereof. Nothing in this Section 10.2 shall affect or limit any right to serve process in any other manner permitted by law.

(iii) EACH OF THE COMPANY AND THE INVESTOR HEREBY WAIVES TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN RESPECT TO ANY LITIGATION DIRECTLY OR INDIRECTLY ARISING OUT OF, UNDER OR IN CONNECTION WITH THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY OR DISPUTES RELATING HERETO. EACH OF THE COMPANY AND THE INVESTOR (A) CERTIFIES THAT NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER AND (B) ACKNOWLEDGES THAT IT AND THE OTHER PARTIES HERETO HAVE BEEN TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION 10.2.

Section 10.3. Entire Agreement; Amendment. The Transaction Documents set forth the entire agreement and understanding of the parties with respect to the subject matter hereof and supersedes all prior and contemporaneous agreements, negotiations and understandings between the parties, both oral and written, with respect to such matters. There are no promises, undertakings, representations or warranties by either party relative to subject matter hereof not expressly set forth in the Transaction Documents. No provision of this Agreement may be amended by the parties from and after the date that is one Trading Day immediately preceding the initial filing of the Registration Statement with the Commission. Subject to the immediately preceding sentence, no provision of this Agreement may be amended other than by a written instrument signed by both parties hereto. The Disclosure Schedule and all exhibits to this Agreement are hereby incorporated by reference in, and made a part of, this Agreement as if set forth in full herein.

Section 10.4. Notices. Any notice, demand, request, waiver or other communication required or permitted to be given hereunder shall be in writing and shall be effective (a) upon hand delivery or facsimile (with facsimile machine confirmation of delivery received) at the address or number designated below (if delivered on a business day during normal business hours where such notice is to be received), or the first business day following such delivery (if delivered other than on a business day during normal business hours where such notice is to be received) or (b) on the second business day following the date of mailing by express courier service, fully prepaid, addressed to such address, or upon actual receipt of such mailing, whichever shall first occur. The address for such communications shall be:

If to the Company:

Terra Tech Corp. 18101 Vin Karman, Third Floor Irvine, CA 92612 Telephone Number: (855) 447-6967 Fax: () Attention: Derek Peterson, CEO

With a copy (which shall not constitute notice) to:

Law Offices of Thomas E. Puzzo, PLLC 3823 44th Ave. NE Seattle, WA 98105 Telephone Number: (206) 522-2256 Fax: (206) 260-0111

Attention: Thomas E. Puzzo, Esq.

If to the Investor:

Hanover Holdings I, LLC c/o Magna Group 5 Hanover Square New York, NY 10004 Telephone Number: (347) 491-4240 Fax: (646) 737-9948 Attention: Marc Manuel

With a copy (which shall not constitute notice) to:

Sichenzia Ross Friedman Ference LLP 61 Broadway New York, NY 10006 Telephone Number: (212) 930-9700 Fax: (212) 930-9725 Attention: Michael H. Ference, Esq.

Either party hereto may from time to time change its address for notices by giving at least 10 days advance written notice of such changed address to the other party hereto.

Section 10.5. Waivers. No provision of this Agreement may be waived by the parties from and after the date that is one Trading Day immediately preceding the initial filing of the Registration Statement with the Commission. Subject to the immediately preceding sentence, no provision of this Agreement may be waived other than in a written instrument signed by the party against whom enforcement of such waiver is sought. No failure or delay in the exercise of any power, right or privilege hereunder shall operate as a waiver thereof, nor shall any single or partial exercise of any such power, right or privilege preclude other or further exercises thereof or of any other right, power or privilege.

Section 10.6. Headings. The article, section and subsection headings in this Agreement are for convenience only and shall not constitute a part of this Agreement for any other purpose and shall not be deemed to limit or affect any of the provisions hereof. Unless the context clearly indicates otherwise, each pronoun herein shall be deemed to include the masculine, feminine, neuter, singular and plural forms thereof. The terms "including," "includes," "include" and words of like import shall be construed broadly as if followed by the words "without limitation." The terms "herein," "hereof" and words of like import refer to this entire Agreement instead of just the provision in which they are found.

Section 10.7. Construction. The parties agree that each of them and their respective counsel has reviewed and had an opportunity to revise the Transaction Documents and, therefore, the normal rule of construction to the effect that any ambiguities are to be resolved against the drafting party shall not be employed in the interpretation of the Transaction Documents. In addition, each and every reference to share prices and shares of Common Stock in any Transaction Document shall be subject to adjustment for any stock splits, stock combinations, stock dividends, recapitalizations and other similar transactions that occur on or after the date of this Agreement. Any reference in this Agreement to "Dollars" or "\$" shall mean the lawful currency of the United States of America.

- Section 10.8. Successors and Assigns. This Agreement shall be binding upon and inure to the benefit of the parties and their respective successors and assigns. The Company may not assign this Agreement or any rights or obligations hereunder to any Person without the prior written consent of the Investor, which may be withheld or delayed in the Investor's sole discretion, including by any Fundamental Transaction. The Investor may not assign its rights or obligations under this Agreement.
- Section 10.9. No Third Party Beneficiaries Except as expressly provided in Article IX, this Agreement is intended only for the benefit of the parties hereto and their respective permitted successors and assigns, and is not for the benefit of, nor may any provision hereof be enforced by, any other Person.
- Section 10.10. Governing Law. This Agreement shall be governed by and construed in accordance with the internal procedural and substantive laws of the State of New York, without giving effect to the choice of law provisions of such state that would cause the application of the laws of any other jurisdiction.
- Section 10.11. <u>Survival</u>. The representations, warranties, covenants and agreements of the Company and the Investor contained in this Agreement shall survive the execution and delivery hereof until the termination of this Agreement; <u>provided</u>, <u>however</u>, that (i) the provisions of Article V (Representations and Warranties of the Company), Article VIII (Termination), Article IX (Indemnification) and this Article X (Miscellaneous) (excluding Section 10.1(v)) shall remain in full force and effect indefinitely notwithstanding such termination, and, (ii) so long as the Investor owns any Securities, the covenants and agreements of the Company contained in Article VI (Additional Covenants), shall remain in full force and effect notwithstanding such termination for a period of six months following such termination.
- Section 10.12. Counterparts. This Agreement may be executed in counterparts, all of which taken together shall constitute one and the same original and binding instrument and shall become effective when all counterparts have been signed by each party and delivered to the other parties hereto, it being understood that all parties hereto need not sign the same counterpart. In the event any signature is delivered by facsimile, digital or electronic transmission, such transmission shall constitute delivery of the manually executed original and the party using such means of delivery shall thereafter cause four additional executed signature pages to be physically delivered to the other parties within five days of the execution and delivery hereof. Failure to provide or delay in the delivery of such additional executed signature pages shall not adversely affect the efficacy of the original delivery.
- Section 10.13. <u>Publicity</u>. The Investor shall have the right to approve, prior to issuance or filing, any press release, Commission filing or any other public disclosure made by or on behalf of the Company relating to the Investor, its purchases hereunder or any aspect of the Transaction Documents or the transactions contemplated thereby; <u>provided</u>, <u>however</u>, that except as otherwise provided in this Agreement, the Company shall be entitled, without the prior approval of the Investor, to make any press release or other public disclosure (including any filings with the Commission) with respect thereto as is required by applicable law and regulations (including the regulations of the Trading Market), so long as prior to making any such press release or other public disclosure; if reasonably practicable, the Company and its counsel shall have provided the Investor and its counsel with a reasonable opportunity to review and comment upon, and shall have consulted with the Investor and its counsel on the form and substance of, such press release or other disclosure. For the avoidance of doubt, the Company shall not be required to submit for review any such disclosure (i) contained in periodic reports filed with the Commission under the Exchange Act if it shall have previously provided the same disclosure for review in connection with a previous filing or (ii) any Prospectus Supplement if it contains disclosure that does not reference the Investor, its purchases hereunder or any aspect of the Transaction Documents or the transactions contemplated thereby.

Section 10.14. Severability. The provisions of this Agreement are severable and, in the event that any court of competent jurisdiction shall determine that any one or more of the provisions or part of the provisions contained in this Agreement shall, for any reason, be held to be invalid, illegal or unenforceable in any respect, such invalidity, illegality or unenforceability shall not affect any other provision or part of a provision of this Agreement, and this Agreement shall be reformed and construed as if such invalid or illegal or unenforceable provision, or part of such provision, had never been contained herein, so that such provisions would be valid, legal and enforceable to the maximum extent possible.

Section 10.15. <u>Further Assurances</u>. From and after the Closing Date, upon the request of the Investor or the Company, each of the Company and the Investor shall execute and deliver such instrument, documents and other writings as may be reasonably necessary or desirable to confirm and carry out and to effectuate fully the intent and purposes of this Agreement.

[Signature Page Follows]

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed by their respective authorized officer as of the date first above written.

TERRA TECH CORP.:

By: /s/ Derek Peterson

Name: Derek Peterson
Title: Chief Executive Officer

HANOVER HOLDINGS I, LLC, a New York limited liability company:

By: /s/ Joshua Sason
Name: Joshua Sason
Title: Managing Member

ANNEX I TO THE COMMON STOCK PURCHASE AGREEMENT <u>DEFINITIONS</u>

- "Acceptable Transaction" shall have the meaning assigned to such term in Section 6.7(iii) hereof.
- "Affiliate" means any Person that, directly or indirectly through one or more intermediaries, controls, is controlled by, or is under common control with a Person, as such terms are used in and construed under Rule 144. With respect to the Investor, without limitation, any Person owning, owned by, or under common ownership with the Investor, and any investment fund or managed account that is managed on a discretionary basis by the same investment manager as the Investor will be deemed to be an Affiliate.
 - "Aggregate Limit" shall have the meaning assigned to such term in Section 2.1 hereof.
 - "Agreement" shall have the meaning assigned to such term in the preamble hereof.
 - "Alternate Transaction" shall have the meaning assigned to such term in Section 6.7(iii) hereof.
 - "Alternate Transaction Notice" shall have the meaning assigned to such term in Section 6.7(ii) hereof.
 - "Announcement Date" shall have the meaning assigned to such term in Section 3.8 hereof.
 - "ATM" shall have the meaning assigned to such term in Section 6.7(iii) hereof.
- "Average Trading Volume" means the average trading volume of the Common Stock on the Trading Market for the ten Trading Days immediately prior to the applicable Draw Down Exercise Date.
 - "Below Market Offering" shall have the meaning assigned to such term in Section 6.7(iii) hereof.
 - "Blackout Period" shall have the meaning assigned to such term in Section 3.8 hereof.
 - "Broker-Dealer" shall have the meaning assigned to such term in Section 6.17 hereof.
 - "Bylaws" shall have the meaning assigned to such term in Section 5.3 hereof.
 - "Charter" shall have the meaning assigned to such term in Section 5.3 hereof.
 - "Closing" shall have the meaning assigned to such term in Section 2.2 hereof.
 - "Closing Date" means the date of this Agreement.

- "Code" means the Internal Revenue Code of 1986, as amended.
- "Commission" means the U.S. Securities and Exchange Commission or any successor entity.
- "Commission Documents" shall mean (1) all reports, schedules, registrations, forms, statements, information and other documents filed with or furnished to the Commission by the Company pursuant to the reporting requirements of the Exchange Act, including all material filed or furnished pursuant to Section 13(a), 13(c), 14 or 15(d) of the Exchange Act, and which hereafter shall be filed with or furnished to the Commission by the Company, including, without limitation, the Current Report, (2) the Registration Statement, as the same may be amended from time to time, the Prospectus and each Prospectus Supplement and (3) all information contained in such filings and all documents and disclosures that have been and heretofore shall be incorporated by reference therein.
 - "Commitment Shares" means the Initial Commitment Shares.
 - "Common Stock" shall have the meaning assigned to such term in the recitals hereof.
 - "Company" shall have the meaning assigned to such term in the preamble hereof.
- "Current Market Price" means, with respect to any particular measurement date, the closing price of a share of Common Stock as reported on the Trading Market for the Trading Day immediately preceding such measurement date.
 - "Current Report" shall have the meaning assigned to such term in Section 2.3 hereof.
 - "Damages" shall have the meaning assigned to such term in Section 9.1 hereof.
 - "Disclosure Schedule" shall have the meaning assigned to such term in the preamble to Article V hereof.
- "Discount Price" means a price equal to 92.5% of the arithmetic average of the lowest VWAPs that equal or exceed the applicable Floor Price during the applicable Pricing Period; provided, however, that if the VWAP does not equal or exceed the applicable Floor Price for at least five Trading Days during the applicable Pricing Period, then the "Discount Price" shall mean a price equal to 92.5% of the arithmetic average of all of the VWAPs that equal or exceed the applicable Floor Price (if any) during such Pricing Period.
 - "Document Preparation Fee" shall have the meaning assigned to such term in Section 10.1 hereof.
- "Draw Down" means the transactions contemplated in Article III of this Agreement with respect to any Draw Down Notice delivered by the Company in accordance with Article III of this Agreement.
 - "Draw Down Amount" means the actual amount of proceeds received by the Company pursuant to a Draw Down under this Agreement.

- "Draw Down Amount Requested" shall mean the specific numbers of shares of Common Stock requested by the Company in a Draw Down Notice delivered pursuant to Section 3.1, up to the Maximum Draw Down Amount Requested.
 - "Draw Down Exercise Date" shall have the meaning assigned to such term in Section 3.1 hereof.
 - "Draw Down Notice" shall have the meaning assigned to such term in Section 3.1 hereof.
 - "DTC" means The Depository Trust Company, or any successor thereto.
 - "Earnings Announcement" shall have the meaning assigned to such term in Section 3.8 hereof.
 - "Earnings 8-K" shall have the meaning assigned to such term in Section 3.8 hereof.
 - "EDGAR" means the Commission's Electronic Data Gathering, Analysis and Retrieval System.
- "Effective Date" means the first Trading Day immediately following the date on which the initial Registration Statement filed pursuant to Section 2(a) of the Registration Rights Agreement is declared effective by the Commission.
 - "Environmental Laws" shall have the meaning assigned to such term in Section 5.18 hereof.
 - "ERISA" shall mean the Employee Retirement Income Security Act of 1974, as amended.
 - "Exchange Act" means the Securities Exchange Act of 1934, as amended, and the rules and regulations of the Commission thereunder.
 - "FCPA" shall have the meaning assigned to such term in Section 5.36 hereof.
 - "Filing Time" shall have the meaning assigned to such term in Section 3.8 hereof.
 - "FINRA" means the Financial Industry Regulatory Authority.
- "Floor Price" is the lowest price at which the Company may sell Shares during the applicable Pricing Period as set forth in the applicable Draw Down Notice (not taking into account the applicable Discount Price).

"Fundamental Transaction" means, except with respect to Naturally Beautiful Plant Products, L.L.C., a New Jersey limited liability company; Gro-Rite, Inc., a New Jersey corporation; Gro-Rite Landscaping, Inc., a New Jersey corporation; Gro-Rite Landscaping Services Limited Liability Company, a New Jersey limited liability company; Gro-Rite Landscape Contractors Inc., a New Jersey corporation; Gro-Rite Garden Centers, Inc., a new Jersey corporation; Gro-Rite Bulk Materials, Inc., a New Jersey corporation; that (i) the Company shall, directly or indirectly, in one or more related transactions, (1) consolidate or merge with or into (whether or not the Company is the surviving corporation) another Person, with the result that the holders of the Company's capital stock immediately prior to such consolidation or merger together beneficially own less than 50% of the outstanding voting power of the surviving or resulting corporation, or (2) sell, lease, license, assign, transfer, convey or otherwise dispose of all or substantially all of the properties or assets of the Company to another Person, or (3) take action to facilitate a purchase, tender or exchange offer by another Person that is accepted by the holders of more than 50% of the outstanding shares of Common Stock (not including any shares of Common Stock held by the Person or Persons making or party to, or associated or affiliated with the Persons making or party to, such purchase, tender or exchange offer), or (4) consummate a stock or share purchase agreement or other business combination), or (5) reorganize, recapitalize or reclassify its Common Stock, or (ii) any "person" or "group" (as these terms are used for purposes of Sections 13(d) and 14(d) of the Exchange Act) is or shall become the "beneficial owner" (as defined in Rule 13d-3 under the Exchange Act), directly or indirectly, of 50% of the aggregate ordinary voting power represented by issued and outstanding Common Stock.

- "GAAP" shall mean generally accepted accounting principles in the United States of America as applied by the Company.
- "Governmental Licenses" shall have the meaning assigned to such term in Section 5.17 hereof.
- "Indebtedness" shall have the meaning assigned to such term in Section 5.11 hereof.
- "Initial Commitment Shares" means [\$125,000 of Common Stock] shares of duly authorized, validly issued, fully paid and nonassessable shares of Common Stock which, concurrently with the execution and delivery of this Agreement on the Closing Date, the Company has caused its transfer agent to issue and deliver to the Investor not later than 4:00 p.m. (New York City time) on the second Trading Day immediately following the Closing Date.
 - "Intellectual Property" shall have the meaning assigned to such term in Section 5.17(b) hereof.
 - "Investment Period" means the period commencing on the Effective Date and expiring on the date this Agreement is terminated pursuant to Article VIII hereof.
 - "Investor" shall have the meaning assigned to such term in the preamble hereof.
 - "Investor Party" shall have the meaning assigned to such term in Section 9.1 hereof.
- "Knowledge" means the actual knowledge of the Company's Chief Executive Officer or Chief Financial Officer, after reasonable inquiry of all officers, directors and employees of the Company who could reasonably be expected to have knowledge or information with respect to the matter in question.

"Make Whole Amount" shall have the meaning assigned to such term in Section 3.6 hereof.

"Material Adverse Effect' means (i) any condition, occurrence, state of facts or event having, or insofar as reasonably can be foreseen would likely have, any material adverse effect on the legality, validity or enforceability of the Transaction Documents or the transactions contemplated thereby, (ii) any condition, occurrence, state of facts or event having, or insofar as reasonably can be foreseen would likely have, any effect on the business, operations, properties or financial condition of the Company that is material and adverse to the Company and its Subsidiaries, taken as a whole, and/or (iii) any condition, occurrence, state of facts or event that would, or insofar as reasonably can be foreseen would likely, prohibit or otherwise materially interfere with or delay the ability of the Company to perform any of its obligations under any of the Transaction Documents to which it is a party; provided, however, that none of the following, individually or in the aggregate, shall be taken into account in determining whether a Material Adverse Effect has occurred or insofar as reasonably can be foreseen would likely occur: (a) changes in conditions in the U.S. or global capital, credit or financial markets generally, including changes in the availability of capital or currency exchange rates, provided such changes shall not have affected the Company in a materially disproportionate manner as compared to other similarly situated companies; (b) changes generally affecting the hydroponic equipment industry, provided such changes shall not have affected the Company in a materially disproportionate manner as compared to other similarly situated companies; (c) any effect of the announcement of, or the consummation of the transactions contemplated by, this Agreement and the other Transaction Documents on the Company's relationships, contractual or otherwise, with customers, suppliers, vendors, bank lenders, strategic venture partners or employees; and (d) the receipt of any notice that the Common Stock may be ineligible

- "Material Agreements" shall have the meaning assigned to such term in Section 5.19 hereof.
- "Maximum Draw Down Amount Requested" means the product of (i) the Average Trading Volume and (ii) 3.0.
- "Money Laundering Laws" shall have the meaning assigned to such term in Section 5.37 hereof.
- "OFAC" shall have the meaning assigned to such term in Section 5.38 hereof.
- "Ownership Limitation" shall have the meaning assigned to such term in Section 3.7 hereof.

- "Person" means any person or entity, whether a natural person, trustee, corporation, partnership, limited partnership, limited liability company, trust, unincorporated organization, business association, firm, joint venture, governmental agency or authority.
 - "Plan" shall have the meaning assigned to such term in Section 5.24 hereof.
 - "Press Release" shall have the meaning assigned to such term in Section 2.3 hereof.
 - "Price Reset Provision" shall have the meaning assigned to such term in Section 6.7(iii) hereof.
- "Pricing Period" shall mean, with respect to each Draw Down, a period of 10 consecutive Trading Days commencing on the Pricing Period start date set forth in the Draw Down Notice in accordance with Section 3.1 hereof.
- "Prospectus" means the prospectus in the form included in the Registration Statement, as supplemented from time to time by any Prospectus Supplement, including the documents incorporated by reference therein.
- "Prospectus Supplement" means any prospectus supplement to the Prospectus filed with the Commission from time to time pursuant to Rule 424(b) under the Securities Act, including the documents incorporated by reference therein.
 - "Reference Period" shall have the meaning assigned to such term in Section 6.7(ii) hereof.
 - "Registrable Securities" shall have the meaning assigned to such term in the Registration Rights Agreement.
 - "Registration Rights Agreement" shall have the meaning assigned to such term in the recitals hereof.
 - "Registration Statement" shall have the meaning assigned to such term in the Registration Rights Agreement.
 - "Regulation D" shall have the meaning assigned to such term in the recitals hereof.
 - "Required Delivery Date" shall have the meaning assigned to such term in Section 10.1(iv) hereto.
 - "Restricted Period" shall have the meaning assigned to such term in Section 6.13(i) hereof.
 - "Restricted Person" shall have the meaning assigned to such term in Section 6.13(i) hereof.
 - "Restricted Persons" shall have the meaning assigned to such term in Section 6.13(i) hereof.

- "Rule 144" means Rule 144 promulgated by the Commission pursuant to the Securities Act, as such Rule may be amended from time to time, or any similar rule or regulation hereafter adopted by the Commission having substantially the same effect.
 - "Section 4(a)(2)" shall have the meaning assigned to such term in the recitals hereof.
 - "Securities" means, collectively, the Shares and the Commitment Shares.
 - "Securities Act" shall mean the Securities Act of 1933, as amended, and the rules and regulations of the Commission thereunder.
 - "Settlement Date" shall have the meaning assigned to such term in Section 3.5 hereof.
- "Shares" shall mean the whole shares of Common Stock that are and/or may be purchased by the Investor under this Agreement pursuant to one or more Draw Downs, and not the Commitment Shares.
 - "Short Sales" shall mean "short sales" as defined in Rule 200 promulgated under Regulation SHO under the Exchange Act.
- "Significant Subsidiary" means any Subsidiary of the Company that would constitute a Significant Subsidiary of the Company within the meaning of Rule 1-02 of Regulation S-X of the Commission.
 - "Similar Financing" shall have the meaning assigned to such term in Section 6.7(iii) hereof.
 - "SOXA" shall mean the Sarbanes-Oxley Act of 2002 and the rules and regulations of the Commission thereunder.
- "Subsidiary" shall mean any corporation or other entity of which at least a majority of the securities or other ownership interest having ordinary voting power for the election of directors or other persons performing similar functions are at the time owned directly or indirectly by the Company and/or any of its other Subsidiaries.
 - "Total Commitment' shall have the meaning assigned to such term in Section 2.1 hereof.
 - "Trading Day" shall mean a full trading day (beginning at 9:30 a.m., New York City time, and ending at 4:00 p.m., New York City time) on the Trading Market.
- "Trading Market" means the OTC Bulletin Board; provided, however, that in the event the Common Stock is ever listed or quoted on the NASDAQ Global Market, the NASDAQ Global Select Market, the NASDAQ Capital Market, the New York Stock Exchange, NYSE Arca, the NYSE MKT, or the OTCQX or the OTCQB operated by OTC Markets Group Inc., than the "Trading Market" shall mean such other market or exchange or any successor to the foregoing on which the Common Stock is then listed or quoted.

"Transaction Documents" means, collectively, this Agreement (as qualified by the Disclosure Schedule) and the exhibits hereto, the Registration Rights Agreement and each of the other agreements, documents, certificates and instruments entered into or furnished by the parties hereto in connection with the transactions contemplated hereby and thereby.

"VWAP" means the volume weighted average price (the aggregate sales price of all trades of Common Stock during a Trading Day divided by the total number of shares of Common Stock traded during such Trading Day) of the Common Stock during a Trading Day as reported by Bloomberg L.P. using the AQR function.

"Warrant Value" shall mean the fair value of all warrants, options and other similar rights issued to a third party in connection with an Alternate Transaction, determined by using a standard Black-Scholes option-pricing model using a reasonable and appropriate expected volatility percentage based on applicable volatility data from an investment banking firm of nationally recognized reputation.

REGISTRATION RIGHTS AGREEMENT

This **REGISTRATION RIGHTS AGREEMENT** (this "Agreement"), dated as of April 26, 2013, is by and between Terra Tech Corp., a Nevada corporation (the "Company"), and Hanover Holdings I, LLC, a New York limited liability company (the "Investor").

RECITALS

- A. The Company and the Investor have entered into that certain Common Stock Purchase Agreement, dated as of the date hereof (the "Purchase Agreement"), pursuant to which the Company may issue, from time to time, to the Investor up to \$5,000,000 of newly issued shares of the Company's common stock, \$0.001 par value ("Common Stock"), as provided for therein.
- B. Pursuant to the terms of, and in consideration for the Investor entering into, the Purchase Agreement, the Company has issued to the Investor the Initial Commitment Shares (as defined in the Purchase Agreement) in accordance with the terms of the Purchase Agreement.
- C. Pursuant to the terms of, and in consideration for the Investor entering into, the Purchase Agreement, and to induce the Investor to execute and deliver the Purchase Agreement, the Company has agreed to provide the Investor with certain registration rights with respect to the Registrable Securities (as defined herein) as set forth herein.

AGREEMENT

NOW, THEREFORE, in consideration of the premises, the representations, warranties, covenants and agreements contained herein and in the Purchase Agreement, and for other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, intending to be legally bound hereby, the Company and the Investor hereby agree as follows:

1. <u>Definitions</u>.

Capitalized terms used herein and not otherwise defined herein shall have the respective meanings set forth in the Purchase Agreement. As used in this Agreement, the following terms shall have the following meanings:

- (a) "Business Day" means any day other than Saturday, Sunday or any other day on which commercial banks in New York, New York are authorized or required by law to remain closed.
 - (b) "Closing Date" shall mean the date of this Agreement.
 - (c) "Effective Date" means the date that the applicable Registration Statement has been declared effective by the SEC.

- (d) "Effectiveness Deadline" means (i) with respect to the initial Registration Statement required to be filed to pursuant to Section 2(a), the earlier of (A) the 90th calendar day after the earlier of (1) the Filing Deadline and (2) the date on which such initial Registration Statement is filed with the SEC and (B) the fifth Business Day after the date the Company is notified (orally or in writing, whichever is earlier) by the SEC that such Registration Statement will not be reviewed or will not be subject to further review and (ii) with respect to any additional Registration Statements that may be required to be filed by the Company pursuant to this Agreement, the earlier of (A) the 90th calendar day following the date on which the Company was required to file such additional Registration Statement and (B) the fifth Business Day after the date the Company is notified (orally or in writing, whichever is earlier) by the SEC that such Registration Statement will not be reviewed or will not be subject to further review.
- (e) "Filing Deadline" means (i) with respect to the initial Registration Statement required to be filed to pursuant to Section 2(a), April 29, 2013 and (ii) with respect to any additional Registration Statements that may be required to be filed by the Company pursuant to this Agreement, the later of (A) the 30th calendar day following the sale of substantially all of the Registration Statement, as applicable, and (B) six months following the Effective Date of the initial Registration Statement or the most recent prior additional Registration Statement, as applicable, or such earlier date as permitted by the SEC.
- (f) "Person" means any person or entity, whether a natural person, trustee, corporation, partnership, limited partnership, limited liability company, trust, unincorporated organization, business association, firm, joint venture, governmental agency or authority.
- (g) "register," "registered," and "registration" refer to a registration effected by preparing and filing one or more Registration Statements in compliance with the Securities Act and pursuant to Rule 415 and the declaration of effectiveness of such Registration Statement(s) by the SEC.
- (h) "Registrable Securities" means all of (i) the Shares, (ii) the Initial Commitment Shares, and (iii) any capital stock of the Company issued or issuable with respect to such Shares or Initial Commitment Shares, including, without limitation, (1) as a result of any stock split, stock dividend, recapitalization, exchange or similar event or otherwise and (2) shares of capital stock of the Company into which the shares of Common Stock are converted or exchanged and shares of capital stock of a successor entity into which the shares of Common Stock are converted or exchanged.
- (i) "Registration Statement" means a registration statement or registration statements of the Company filed under the Securities Act covering the resale by the Investor of Registrable Securities, as such registration statement or registration statements may be amended and supplemented from time to time (including pursuant to Rule 462(b) under the Securities Act), including all documents filed as part thereof or incorporated by reference therein.
- (j) "Rule 144" means Rule 144 promulgated by the SEC under the Securities Act, as such rule may be amended from time to time, or any other similar or successor rule or regulation of the SEC that may at any time permit the Investor to sell securities of the Company to the public without registration.

- (k) "Rule 415" means Rule 415 promulgated by the SEC under the Securities Act, as such rule may be amended from time to time, or any other similar or successor rule or regulation of the SEC providing for offering securities on a delayed or continuous basis.
 - (1) "SEC" means the U.S. Securities and Exchange Commission or any successor entity.

Registration.

- (a) <u>Mandatory Registration</u>. The Company shall prepare and, as soon as practicable, but in no event later than the Filing Deadline, file with the SEC an initial Registration Statement on Form S-1, or such other form reasonably acceptable to the Investor and Legal Counsel, covering the resale by the Investor of Registrable Securities in an amount equal to [____] shares of Common Stock, [____] of which shares of Common Stock shall be registered as Initial Commitment Shares, and the balance of which shares of Common Stock shall be registered as Shares. Such initial Registration Statement shall contain (except if otherwise directed by the Investor) the "<u>Selling Stockholder</u>" and "<u>Plan of Distribution</u>" sections in substantially the form attached hereto as <u>Exhibit B</u>. The Company shall use its commercially reasonable efforts to have such initial Registration Statement, and each other Registration Statement required to be filed pursuant to the terms hereof, declared effective by the SEC as soon as practicable, but in no event later than the applicable Effectiveness Deadline.
- (b) <u>Legal Counsel</u>. Subject to Section 5 hereof, the Investor shall have the right to select one legal counsel to review and oversee, solely on its behalf, any registration pursuant to this Section 2 ("*Legal Counsel*"), which shall be Sichenzia Ross Friedman Ference LLP or such other counsel as thereafter designated by the Investor. Except as provided under Section 10.1(i) of the Purchase Agreement, the Company shall have no obligation to reimburse the Investor for any and all legal fees and expenses of the Legal Counsel incurred in connection with the transactions contemplated hereby.

(c) Reserved

(d) <u>Sufficient Number of Shares Registered.</u> If at any time all Registrable Securities are not covered by the initial Registration Statement filed pursuant to Section 2(a) as a result of Section 2(h) or otherwise, the Company shall file with the SEC one or more additional Registration Statements (on the short form available therefor, if applicable), so as to cover all of the Registrable Securities not covered by such initial Registration Statement, in each case, as soon as practicable (taking into account any Staff position with respect to date on which the Staff will permit such additional Registration Statement(s) to be filed with the SEC), but in no event later than the applicable Filing Deadline for such additional Registration Statement(s). The Company shall use its commercially reasonable efforts to cause such additional Registration Statement(s) to become effective as soon as practicable following the filing thereof with the SEC, but in no event later than the applicable Effectiveness Deadline for such Registration Statement.

- (e) <u>Piggyback Registrations</u>. Without limiting any of the Company's obligations hereunder or under the Purchase Agreement, if there is not an effective Registration Statement covering all of the Registrable Securities and the Company shall determine to prepare and file with the SEC a registration statement relating to an offering for its own account or the account of others under the Securities Act of any of its equity securities (other than on Form S-4 or Form S-8 (each as promulgated under the Securities Act) or their then equivalents relating to equity securities to be issued solely in connection with any acquisition of any entity or business or equity securities issuable in connection with the Company's stock option or other employee benefit plans), then the Company shall deliver to the Investor a written notice of such determination and, if within five (5) days after the offer and sale of which the Investor shall so request in writing, the Company shall include in such registration statement all or any part of such Registrable Securities the offer and sale of which the Investor requests to be registered; <u>provided</u>, <u>however</u>, the Company shall not be required to register the offer and sale of any Registrable Securities pursuant to this Section 2(e) that are eligible for resale pursuant to Rule 144 without restriction (including, without limitation, volume restrictions) and without the need for current public information required by Rule 144(c)(1) (or Rule 144(i)(2), if applicable) or that are the subject of a then-effective Registration Statement.
- (f) No Inclusion of Other Securities. In no event shall the Company include any securities other than Registrable Securities on any Registration Statement pursuant to Section 2(a) or 2(d) without the prior written consent of the Investor. Subject to the proviso in Section 2(e), in connection with any offering contemplated by Section 2(e) involving an underwriting of shares, the Company shall not be required under this Section 2 or otherwise to include the Registrable Securities of any Investor therein unless such Investor accepts and agrees to the terms of the underwriting, which shall be reasonable and customary, as agreed upon between the Company and the underwriters selected by the Company. If, in connection with any offering contemplated by Section 2(e) involving an underwriting of shares, the managing underwriters advise the Company and the Investors that, in their opinion, the number of securities requested to be included in such registration exceeds the number which can be sold in such offering without adversely affecting the marketability, proposed offering price, timing, distribution method or probability of success of such offering, then the Company will include in such registration (i) first, the securities that the Company proposes to sell, (ii) second, the Registrable Securities requested to be included in such registration, which in the opinion of such underwriters can be sold without adverse effect, and (iii) third, other securities requested to be included in such registration which in the opinion of such underwriters can be sold without adverse effect, pro rata among the holders of such securities on the basis of the number of such securities owned by each such holder.
- Offering. If the staff of the SEC (the "Staff") or the SEC seeks to characterize any offering pursuant to a Registration Statement filed pursuant to this Agreement as constituting an offering of securities that does not permit such Registration Statement to become effective and be used for resales by the Investor on a delayed or continuous basis under Rule 415 at then-prevailing market prices (and not fixed prices) (or as otherwise may be acceptable to the Investor), or if after the filing of the initial Registration Statement with the SEC pursuant to Section 2(a), the Company is otherwise required by the Staff or the SEC to reduce the number of Registrable Securities included in such initial Registration Statement, then the Company shall reduce the number of Registrable Securities to be included in such initial Registration Statement (with the prior consent of the Investor and Legal Counsel as to the specific Registrable Securities to be removed therefrom) until such time as the Staff and the SEC shall so permit such Registration Statement to become effective and be used as aforesaid. Notwithstanding anything in this Agreement to the contrary, if after giving effect to the actions referred to in the immediately preceding sentence, the Staff or the SEC does not permit such Registration Statement to become effective and be used for resales by the Investor on a delayed or continuous basis under Rule 415 at then-prevailing market prices (and not fixed prices) (or as otherwise may be acceptable to the Investor), the Company shall not request acceleration of the Effective Date of such Registration Statement, the Company shall promptly (but in no event later than 48 hours) request the withdrawal of such Registration Statement pursuant to Rule 477 under the Securities Act, and the Effectiveness Deadline shall automatically be deemed to have elapsed with respect to such Registration Statement at such time as the Staff or the SEC has made a final and non-appealable determination that the SEC will not permit such Registration Statement to be so utilized (unless prior to such time the Company and the Investor have received assurances from the Staff or the SEC reasonably acceptable to Legal Counsel that a new Registration Statement filed by the Company with the SEC promptly thereafter may be so utilized). In the event of any reduction in Registrable Securities pursuant to this paragraph, the Company shall file additional Registration Statements in accordance with Section 2(d) until such time as all Registrable Securities have been included in Registration Statements that have been declared effective and the prospectus contained therein is available for use by the Investor.

Related Obligations.

The Company shall use its commercially reasonable efforts to effect the registration of the Registrable Securities in accordance with the intended method of disposition thereof, and, pursuant thereto, the Company shall have the following obligations:

(a) The Company shall promptly prepare and file with the SEC a Registration Statement with respect to the Registrable Securities (but in no event later than the applicable Filing Deadline) and use its commercially reasonable efforts to cause such Registration Statement to become effective as soon as practicable after such filing (but in no event later than the applicable Effectiveness Deadline). Subject to Allowable Grace Periods, the Company shall keep each Registration Statement effective (and the prospectus contained therein available for use) pursuant to Rule 415 for resales by the Investor on a delayed or continuous basis at then-prevailing market prices (and not fixed prices) at all times until the earlier of (i) the date as of which the Investor may sell all of the Registrable Securities required to be covered by such Registration Statement (disregarding any reduction pursuant to Section 2(g)) without restriction pursuant to Rule 144 and without the need for current public information as required by Rule 144(c)(1) (or Rule 144(i)(2), if applicable) and (ii) the date on which the Investor shall have sold all of the Registrable Securities covered by such Registration Statement (the "Registration Period"). Notwithstanding anything to the contrary contained in this Agreement (but subject to the provisions of Section 3(q) hereof), the Company shall ensure that, when filed and at all times while effective, each Registration Statement (including, without limitation, all amendments and supplements thereto) used in connection with such Registration Statement shall not contain any untrue statement of a material fact or omit to state a material fact required to be stated therein, or necessary to make the statements therein (in the case of prospectuses, in the light of the circumstances in which they were made) not misleading. The Company shall submit to the SEC, within two (2) Business Days after the later of the date that (i) the Company learns that no review of a particular Registration Statement will be made by

Subject to Section 3(q) of this Agreement, the Company shall prepare and file with the SEC such amendments (including, without limitation, post-effective amendments) and supplements to each Registration Statement and the prospectus used in connection with each such Registration Statement, which prospectus is to be filed pursuant to Rule 424 promulgated under the Securities Act, as may be necessary to keep each such Registration Statement effective (and the prospectus contained therein current and available for use) at all times during the Registration Period for such Registration Statement, and, during such period, comply with the provisions of the Securities Act with respect to the disposition of all Registrable Securities of the Company required to be covered by such Registration Statement until such time as all of such Registrable Securities shall have been disposed of in accordance with the intended methods of disposition by the seller or sellers thereof as set forth in such Registration Statement. Without limiting the generality of the foregoing, the Company covenants and agrees that (i) at or before 8:30 a.m. (New York City time) on the Trading Day immediately following each Effective Date, the Company shall file with the SEC in accordance with Rule 424(b) under the Securities Act the final prospectus to be used in connection with sales pursuant to the applicable Registration Statement, and (ii) if the transactions contemplated by any Draw Down (as defined in the Purchase Agreement) are material to the Company (individually or collectively with all other prior Draw Downs, the consummation of which have not previously been reported in any prospectus supplement filed with the SEC under Rule 424(b) under the Securities Act or in any periodic report filed by the Company with the SEC under the Securities Exchange Act of 1934, as amended (the "Exchange Act")), or if otherwise required under the Securities Act, in each case as reasonably determined by the Company or the Investor, then, on the first Trading Day immediately following the last Trading Day of the Pricing Period with respect to such Draw Down, the Company shall file with the SEC a prospectus supplement pursuant to Rule 424(b) under the Securities Act with respect to the applicable Draw Down(s), disclosing the total Draw Down Amount Requested pursuant to such Draw Down(s), the total number of Shares that have been (or are to be) issued and sold to the Investor pursuant to such Draw Down(s), the total purchase price for the Shares subject to such Draw Down(s), the applicable Discount Price(s) for such Shares and the net proceeds that have been (or are to be) received by the Company from the sale of such Shares. To the extent not previously disclosed in the prospectus or a prospectus supplement, the Company shall disclose in its Quarterly Reports on Form 10-Q and in its Annual Reports on Form 10-K the information described in the immediately preceding sentence relating to any Draw Down(s) consummated during the relevant fiscal quarter. In the case of amendments and supplements to any Registration Statement or prospectus which are required to be filed pursuant to this Agreement (including, without limitation, pursuant to this Section 3(b)) by reason of the Company filing a report on Form 8-K, Form 10-Q or Form 10-K or any analogous report under the Exchange Act, the Company shall have incorporated such report by reference into such Registration Statement and prospectus, if applicable, or shall file such amendments or supplements to the Registration Statement or prospectus with the SEC on the same day on which the Exchange Act report is filed which created the requirement for the Company to amend or supplement such Registration Statement or prospectus, for the purpose of including or incorporating such report into such Registration Statement and prospectus. The Company consents to the use of the prospectus (including, without limitation, any supplement thereto) included in each Registration Statement in accordance with the provisions of the Securities Act and with the securities or "Blue Sky" laws of the jurisdictions in which the Registrable Securities may be sold by the Investor, in connection with the resale of the Registrable Securities and for such period of time thereafter as such prospectus (including, without limitation, any supplement thereto) (or in lieu thereof, the notice referred to in Rule 173(a) under the Securities Act) is required by the Securities Act to be delivered in connection with resales of Registrable Securities.

- (c) The Company shall (A) permit Legal Counsel to review and comment upon (i) each Registration Statement at least five (5) Business Days prior to its filing with the SEC (or such shorter period as may be agreed to by the Investor and Legal Counsel) and (ii) all amendments and supplements to each Registration Statement (including, without limitation, the prospectus contained therein) (except for Annual Reports on Form 10-K, Quarterly Reports on Form 10-Q, Current Reports on Form 8-K, and any similar or successor reports or prospectus supplements the contents of which is limited to that set forth in such reports) within a reasonable number of days prior to their filing with the SEC, and (B) not file any Registration Statement or amendment or supplement thereto or to any prospectus contained therein in a form to which Legal Counsel reasonably objects. The Company shall not submit a request for acceleration of the effectiveness of a Registration Statement or any amendment or supplement thereto without the prior consent of Legal Counsel, which consent shall not be unreasonably withheld. The Company shall promptly furnish to Legal Counsel, without charge, (i) electronic copies of any correspondence from the SEC or the Staff to the Company or its representatives relating to each Registration Statement (which correspondence shall be redacted to exclude any material, non-public information regarding the Company or any of its Subsidiaries), (ii) after the same is prepared and filed with the SEC, one (1) electronic copy of each Registration Statement and any amendment(s) and supplement(s) thereto, including, without limitation, financial statements and schedules, all documents incorporated therein by reference, if requested by the Investor, and all exhibits and (iii) upon the effectiveness of each Registration Statement, one (1) electronic copy of the prospectus included in such Registration Statement and all amendments and supplements thereto. The Company shall reasonably cooperate with Legal Counsel in performing the C
- (d) Without limiting any obligation of the Company under the Purchase Agreement, the Company shall promptly furnish to the Investor, without charge, (i) after the same is prepared and filed with the SEC, at least one (1) electronic copy of each Registration Statement and any amendment(s) and supplement(s) thereto, including, without limitation, financial statements and schedules, all documents incorporated therein by reference, if requested by the Investor, all exhibits and each preliminary prospectus, (ii) upon the effectiveness of each Registration Statement, one (1) electronic copy of the prospectus included in such Registration Statement and all amendments and supplements thereto (or such other number of copies as the Investor may reasonably request from time to time) and (iii) such other documents, including, without limitation, copies of any preliminary or final prospectus, as the Investor may reasonably request from time to time in order to facilitate the disposition of the Registrable Securities owned by the Investor.

- (e) The Company shall take such action as is necessary to (i) register and qualify, unless an exemption from registration and qualification applies, the resale by the Investor of the Registrable Securities covered by a Registration Statement under such other securities or "Blue Sky" laws of all applicable jurisdictions in the United States, (ii) prepare and file in those jurisdictions, such amendments (including, without limitation, post-effective amendments) and supplements to such registrations and qualifications as may be necessary to maintain the effectiveness thereof during the Registration Period, (iii) take such other actions as may be necessary to maintain such registrations and qualifications in effect at all times during the Registration Period, and (iv) take all other actions reasonably necessary or advisable to qualify the Registrable Securities for sale in such jurisdictions; provided, however, the Company shall not be required in connection therewith or as a condition thereto to (x) qualify to do business in any jurisdiction where it would not otherwise be required to qualify but for this Section 3(e), (y) subject itself to general taxation in any such jurisdiction, or (z) file a general consent to service of process in any such jurisdiction. The Company shall promptly notify Legal Counsel and the Investor of the receipt by the Company of any notification with respect to the suspension of the registration or qualification of any of the Registrable Securities for sale under the securities or "Blue Sky" laws of any jurisdiction in the United States or its receipt of actual notice of the initiation or threatening of any proceeding for such purpose.
- (f) The Company shall notify Legal Counsel and the Investor in writing of the happening of any event, as promptly as practicable after becoming aware of such event, as a result of which the prospectus included in a Registration Statement, as then in effect, includes an untrue statement of a material fact or omission to state a material fact required to be stated therein or necessary to make the statements therein, in the light of the circumstances under which they were made, not misleading (provided that in no event shall such notice contain any material, non-public information regarding the Company or any of its Subsidiaries), and, subject to Section 3(q), promptly prepare a supplement or amendment to such Registration Statement and such prospectus contained therein to correct such untrue statement or omission and deliver one (1) electronic copy of such supplement or amendment to Legal Counsel and the Investor (or such other number of copies as Legal Counsel or the Investor may reasonably request). The Company shall also promptly notify Legal Counsel and the Investor in writing (i) when a prospectus or any prospectus supplement or post-effective amendment has become effective (notification of such effectiveness shall be delivered to Legal Counsel and the Investor by facsimile or e-mail on the same day of such effectiveness and by overnight mail), and when the Company receives written notice from the SEC that a Registration Statement or any post-effective amendment will be reviewed by the SEC, (ii) of any request by the SEC for amendments or supplements to a Registration Statement or related prospectus or related information, (iii) of the Company's reasonable determination that a post-effective amendment to a Registration Statement or any amendment or supplement thereto or any related or any other federal or state governmental authority for any additional information relating to the Registration Statement or any amendment or any amendment thereto. Nothing in this Section 3(f) shall limit any obligation of the C

- (g) The Company shall (i) use its reasonable best efforts to prevent the issuance of any stop order or other suspension of effectiveness of a Registration Statement or the use of any prospectus contained therein, or the suspension of the qualification, or the loss of an exemption from qualification, of any of the Registrable Securities for sale in any jurisdiction and, if such an order or suspension is issued, to obtain the withdrawal of such order or suspension at the earliest possible time and (ii) notify Legal Counsel and the Investor of the issuance of such order and the resolution thereof or its receipt of actual notice of the initiation or threat of any proceeding for such purpose.
- (h) Upon the written request of the Investor, the Company shall, within a reasonable time period following such request, make available for inspection during normal business hours by (i) the Investor, (ii) Legal Counsel and (iii) one (1) firm of accountants or other agents retained by such Investor (collectively, the "Inspectors"), all pertinent financial and other records, and pertinent corporate documents and properties of the Company (collectively, the "Records"), as shall be reasonably deemed necessary by each Inspector, and cause the Company's officers, directors and employees to supply all information which any Inspector may reasonably request; provided, however, each Inspector shall agree in writing to hold in strict confidence and not to make any disclosure (except to the Investor) or use of any Record or other information which the Company's board of directors determines in good faith to be confidential, and of which determination the Inspectors are so notified, unless (a) the disclosure of such Records is necessary to avoid or correct a misstatement or omission in any Registration Statement or is otherwise required under the Securities Act, (b) the release of such Records is ordered pursuant to a final, non-appealable subpoena or order from a court or government body of competent jurisdiction, or (c) the information in such Records has been made generally available to the public other than by disclosure in violation of this Agreement or any other Transaction Document (as defined in the Purchase Agreement). The Investor agrees that it shall, upon learning that disclosure of such Records is sought in or by a court or governmental body of competent jurisdiction or through other means, give prompt notice to the Company and allow the Company, at its expense, to undertake appropriate action to prevent disclosure of, or to obtain a protective order for, the Records deemed confidential. Nothing herein (or in any other confidentiality agreement between the Company and the Investor, if any) shall be deemed to l
- (i) The Company shall hold in confidence and not make any disclosure of information concerning the Investor provided to the Company unless (i) disclosure of such information is necessary to comply with federal or state securities laws, (ii) the disclosure of such information is necessary to avoid or correct a misstatement or omission in any Registration Statement or is otherwise required to be disclosed in such Registration Statement pursuant to the Securities Act, (iii) the release of such information is ordered pursuant to a subpoena or other final, non-appealable order from a court or governmental body of competent jurisdiction, or (iv) such information has been made generally available to the public other than by disclosure in violation of this Agreement or any other Transaction Document. The Company agrees that it shall, upon learning that disclosure of such information concerning the Investor is sought in or by a court or governmental body of competent jurisdiction or through other means, give prompt written notice to the Investor and allow the Investor, at the Investor's expense, to undertake appropriate action to prevent disclosure of, or to obtain a protective order for, such information.

- (j) Without limiting any obligation of the Company under the Purchase Agreement, the Company shall use its reasonable best efforts either to (i) cause all of the Registrable Securities covered by each Registration Statement to be listed on each securities exchange on which securities of the same class or series issued by the Company are then listed, if any, if the listing of such Registrable Securities is then permitted under the rules of such exchange or (ii) secure designation and quotation of all of the Registrable Securities covered by each Registration Statement on another Trading Market, or (iii) if, despite the Company's reasonable best efforts to satisfy the preceding clauses (i) or (ii) the Company is unsuccessful in satisfying the preceding clauses (i) or (ii), without limiting the generality of the foregoing, to use its reasonable best efforts to arrange for at least two market makers to register with the Financial Industry Regulatory Authority (f/k/a the National Association of Securities Dealers, Inc.) ("FINRA") as such with respect to such Registrable Securities. In addition, the Company shall cooperate with the Investor and any Broker-Dealer through which the Investor proposes to sell its Registrable Securities in effecting a filing with FINRA pursuant to FINRA Rule 5110 as requested by the Investor. The Company shall pay all fees and expenses in connection with satisfying its obligation under this Section 3(j).
- (k) The Company shall cooperate with the Investor and, to the extent applicable, facilitate the timely preparation and delivery of certificates (not bearing any restrictive legend) representing the Registrable Securities to be offered pursuant to a Registration Statement and enable such certificates to be in such denominations or amounts (as the case may be) as the Investor may reasonably request from time to time and registered in such names as the Investor may request. Certificates for Registrable Securities free from all restrictive legends may be transmitted by the transfer agent to the Investor by crediting an account at DTC as directed by the Investor.
- (l) If requested by the Investor, the Company shall as soon as practicable after receipt of notice from the Investor and subject to Section 3(q) hereof, (i) incorporate in a prospectus supplement or post-effective amendment such information as the Investor reasonably requests to be included therein relating to the sale and distribution of Registrable Securities, including, without limitation, information with respect to the number of Registrable Securities being offered or sold, the purchase price being paid therefor and any other terms of the offering of the Registrable Securities to be sold in such offering; (ii) make all required filings of such prospectus supplement or post-effective amendment after being notified of the matters to be incorporated in such prospectus supplement or post-effective amendment; and (iii) supplement or make amendments to any Registration Statement or prospectus contained therein if reasonably requested by the Investor.
- (m) The Company shall use its reasonable best efforts to cause the Registrable Securities covered by a Registration Statement to be registered with or approved by such other governmental agencies or authorities as may be necessary to consummate the disposition of such Registrable Securities.

- (n) The Company shall make generally available to its security holders as soon as practical, but not later than ninety (90) days after the close of the period covered thereby, an earnings statement (in form complying with, and in the manner provided by, the provisions of Rule 158 under the Securities Act) covering a twelve-month period beginning not later than the first day of the Company's fiscal quarter next following the applicable Effective Date of each Registration Statement.
- (o) The Company shall otherwise use its reasonable best efforts to comply with all applicable rules and regulations of the SEC in connection with any registration hereunder.
- (p) Within one (1) Business Day after each Registration Statement which covers Registrable Securities is declared effective by the SEC, the Company shall deliver, and shall cause legal counsel for the Company to deliver, to the transfer agent for such Registrable Securities (with copies to the Investor) confirmation that such Registration Statement has been declared effective by the SEC in the form attached hereto as Exhibit A.
- Notwithstanding anything to the contrary herein (but subject to the last sentence of this Section 3(q)), at any time after the Effective Date of a particular Registration Statement, the Company may delay the disclosure of material, non-public information concerning the Company or any of its Subsidiaries the disclosure of which at the time is not, in the good faith opinion of the board of directors of the Company, in the best interest of the Company and, in the opinion of counsel to the Company, otherwise required (a "Grace Period"), provided that the Company shall promptly, but in no event later than 9:30 a.m. (New York City time) on the second Trading Day immediately prior to the commencement of any Grace Period (except for such case where it is impossible to provide such two-Trading Day advance notice, in which case the Company shall provide such notice as soon as possible), notify the Investor in writing of the (i) existence of material, non-public information giving rise to a Grace Period (provided that in each such notice the Company shall not disclose the content of such material, non-public information to the Investor) and the date on which such Grace Period will begin and (ii) date on which such Grace Period ends, provided further that (I) no Grace Period shall exceed 20 consecutive Trading Days and during any 365-day period all such Grace Periods shall not exceed an aggregate of 60 Trading Days; provided, further, that the Company shall not register any securities for the account of itself or any other stockholder during any such Grace Period (other than pursuant to a registration statement on Form S-4 or S-8), (II) the first day of any Grace Period must be at least three Trading Days (or such shorter period as may be agreed by the parties) after the last day of any prior Grace Period and (III) no Grace Period may exist during (A) the first 10 consecutive Trading Days after the Effective Date of the particular Registration Statement or (B) the five-Trading Day period following each Settlement Date (each, an "Allowable Grace Period"). For purposes of determining the length of a Grace Period above, such Grace Period shall begin on and include the date set forth in the notice referred to in clause (i) above, provided that such notice is received by the Investor not later than 9:30 a.m. (New York City time) on the second Trading Day immediately prior to such commencement date (except for such case where it is impossible to provide such two-Trading Day advance notice, in which case the Company shall provide such notice as soon as possible) and shall end on and include the later of the date the Investor receives the notice referred to in clause (ii) above and the date referred to in such notice. The provisions of Section 3(1) hereof shall not be applicable during the period of any Allowable Grace Period. Upon expiration of each Grace Period, the Company shall again be bound by the first sentence of Section 3(f) with respect to the information giving rise thereto unless such material, non-public information is no longer applicable. Notwithstanding anything to the contrary contained in this Section 3(q), the Company shall cause its transfer agent to deliver unlegended shares of Common Stock to a transferee of the Investor in accordance with the terms of the Purchase Agreement in connection with any sale of Registrable Securities with respect to which the Investor has entered into a contract for sale, and delivered a copy of the prospectus included as part of the particular Registration Statement to the extent applicable, prior to the Investor's receipt of the notice of a Grace Period and for which the Investor has not yet settled.

(r) The Company shall take all other reasonable actions necessary to expedite and facilitate disposition by the Investor of its Registrable Securities pursuant to each Registration Statement.

Obligations of the Investor.

- (a) At least five Business Days prior to the first anticipated filing date of each Registration Statement (or such shorter period to which the parties agree), the Company shall notify the Investor in writing of the information the Company requires from the Investor with respect to such Registration Statement. It shall be a condition precedent to the obligations of the Company to complete the registration pursuant to this Agreement with respect to the Registrable Securities of the Investor that the Investor shall furnish to the Company such information regarding itself, the Registrable Securities held by it and the intended method of disposition of the Registrable Securities held by it, as shall be reasonably required to effect and maintain the effectiveness of the registration of such Registrable Securities and shall execute such documents in connection with such registration as the Company may reasonably request.
- (b) The Investor, by its acceptance of the Registrable Securities, agrees to cooperate with the Company as reasonably requested by the Company in connection with the preparation and filing of each Registration Statement hereunder, unless the Investor has notified the Company in writing of the Investor's election to exclude all of the Investor's Registrable Securities from such Registration Statement.
- (c) The Investor agrees that, upon receipt of any notice from the Company of the happening of any event of the kind described in Section 3(g) or the first sentence of 3(f), the Investor will immediately discontinue disposition of Registrable Securities pursuant to any Registration Statement(s) covering such Registrable Securities until the Investor's receipt of the copies of the supplemented or amended prospectus contemplated by Section 3(g) or the first sentence of Section 3(f) or receipt of notice that no supplement or amendment is required. Notwithstanding anything to the contrary in this Section 4(c), the Company shall cause its transfer agent to deliver unlegended shares of Common Stock to a transferee of the Investor in accordance with the terms of the Purchase Agreement in connection with any sale of Registrable Securities with respect to which the Investor has entered into a contract for sale prior to the Investor's receipt of a notice from the Company of the happening of any event of the kind described in Section 3(g) or the first sentence of Section 3(f) and for which the Investor has not yet settled.
- (d) The Investor covenants and agrees that it will comply with the prospectus delivery and other requirements of the Securities Act as applicable to it in connection with sales of Registrable Securities pursuant to a Registration Statement.

Expenses of Registration.

All reasonable expenses, other than underwriting discounts and commissions, incurred in connection with registrations, filings or qualifications pursuant to Sections 2 and 3, including, without limitation, all registration, listing and qualifications fees, printers and accounting fees, FINRA filing fees (if any) and fees and disbursements of counsel for the Company shall be paid by the Company.

Indemnification.

In the event any Registrable Securities are included in any Registration Statement under this Agreement, to the fullest extent permitted by law, the Company will, and hereby does, indemnify, hold harmless and defend the Investor, each of its directors, officers, shareholders, members, partners, employees, agents, advisors, representatives (and any other Persons with a functionally equivalent role of a Person holding such titles notwithstanding the lack of such title or any other title) and each Person, if any, who controls the Investor within the meaning of the Securities Act or the Exchange Act and each of the directors, officers, shareholders, members, partners, employees, agents, advisors, representatives (and any other Persons with a functionally equivalent role of a Person holding such titles notwithstanding the lack of such title or any other title) of such controlling Persons (each, an "Investor Party" and collectively, the "Investor Parties"), against any losses, obligations, claims, damages, liabilities, contingencies, judgments, fines, penalties, charges, costs (including, without limitation, court costs, reasonable attorneys' fees, costs of defense and investigation), amounts paid in settlement or expenses, joint or several, (collectively, "Claims") incurred in investigating, preparing or defending any action, claim, suit, inquiry, proceeding, investigation or appeal taken from the foregoing by or before any court or governmental, administrative or other regulatory agency, body or the SEC, whether pending or threatened, whether or not an Investor Party is or may be a party thereto ("Indemnified Damages"), to which any of them may become subject insofar as such Claims (or actions or proceedings, whether commenced or threatened, in respect thereof) arise out of or are based upon: (i) any untrue statement or alleged untrue statement of a material fact in a Registration Statement or any post-effective amendment thereto or in any filing made in connection with the qualification of the offering under the securities or other "Blue Sky" laws of any jurisdiction in which Registrable Securities are offered ("Blue Sky Filing"), or the omission or alleged omission to state a material fact required to be stated therein or necessary to make the statements therein not misleading or (ii) any untrue statement or alleged untrue statement of a material fact contained in any prospectus (as amended or supplemented) or in any prospectus supplement or the omission or alleged omission to state therein any material fact necessary to make the statements made therein, in light of the circumstances under which the statements therein were made, not misleading (the matters in the foregoing clauses (i) and (ii) being, collectively, "Violations"). Subject to Section 6(c), the Company shall reimburse the Investor Parties, promptly as such expenses are incurred and are due and payable, for any legal fees or other reasonable expenses incurred by them in connection with investigating or defending any such Claim. Notwithstanding anything to the contrary contained herein, the indemnification agreement contained in this Section 6(a): (i) shall not apply to a Claim by an Investor Party arising out of or based upon a Violation which occurs in reliance upon and in conformity with information furnished in writing to the Company by such Investor Party for such Investor Party expressly for use in connection with the preparation of such Registration Statement, prospectus or prospectus supplement or any such amendment thereof or supplement thereto; (ii) shall not be available to the Investor to the extent such Claim is based on a failure of the Investor to deliver or to cause to be delivered the prospectus (as amended or supplemented) made available by the Company (to the extent applicable), including, without limitation, a corrected prospectus, if such prospectus (as amended or supplemented) or corrected prospectus was timely made available by the Company pursuant to Section 3(d) and then only if, and to the extent that, following the receipt of the corrected prospectus no grounds for such Claim would have existed; and (iii) shall not apply to amounts paid in settlement of any Claim if such settlement is effected without the prior written consent of the Company, which consent shall not be unreasonably withheld or delayed. Such indemnity shall remain in full force and effect regardless of any investigation made by or on behalf of the Investor Party and shall survive the transfer of any of the Registrable Securities by the Investor pursuant to Section 9.

- (b) In connection with any Registration Statement in which the Investor is participating, the Investor agrees to severally and not jointly indemnify, hold harmless and defend, to the same extent and in the same manner as is set forth in Section 6(a), the Company, each of its directors, each of its officers who signs the Registration Statement and each Person, if any, who controls the Company within the meaning of the Securities Act or the Exchange Act (each, an "Company Party"), against any Claim or Indemnified Damages to which any of them may become subject, under the Securities Act, the Exchange Act or otherwise, insofar as such Claim or Indemnified Damages arise out of or are based upon any Violation, in each case, to the extent, and only to the extent, that such Violation occurs in reliance upon and in conformity with written information relating to the Investor furnished to the Company by the Investor expressly for use in connection with such Registration Statement; and, subject to Section 6(c) and the below provisos in this Section 6(b), the Investor will reimburse a Company Party any legal or other expenses reasonably incurred by such Company Party in connection with investigating or defending any such Claim; provided, however, the indemnity agreement contained in this Section 6(b) and the agreement with respect to contribution contained in Section 7 shall not apply to amounts paid in settlement of any Claim if such settlement is effected without the prior written consent of the Investor, which consent shall not be unreasonably withheld or delayed, provided further that the Investor shall be liable under this Section 6(b) for only that amount of a Claim or Indemnified Damages as does not exceed the net proceeds to the Investor as a result of the applicable sale of Registrable Securities pursuant to such Registration Statement. Such indemnity shall remain in full force and effect regardless of any investigation made by or on behalf of such Company Party and shall survive the transfer of any of the Registr
- Promptly after receipt by an Investor Party or Company Party (as the case may be) under this Section 6 of notice of the commencement of any action or proceeding (including, without limitation, any governmental action or proceeding) involving a Claim, such Investor Party or Company Party (as the case may be) shall, if a Claim in respect thereof is to be made against any indemnifying party under this Section 6, deliver to the indemnifying party a written notice of the commencement thereof, and the indemnifying party shall have the right to participate in, and, to the extent the indemnifying party so desires, jointly with any other indemnifying party similarly noticed, to assume control of the defense thereof with counsel mutually satisfactory to the indemnifying party and the Investor Party or the Company Party (as the case may be); provided, however, an Investor Party or Company Party (as the case may be) shall have the right to retain its own counsel with the fees and expenses of such counsel to be paid by the indemnifying party if: (i) the indemnifying party has agreed in writing to pay such fees and expenses; (ii) the indemnifying party shall have failed promptly to assume the defense of such Claim and to employ counsel reasonably satisfactory to such Investor Party or Company Party (as the case may be) in any such Claim; or (iii) the named parties to any such Claim (including, without limitation, any impleaded parties) include both such Investor Party or Company Party (as the case may be) and the indemnifying party, and such Investor Party or such Company Party (as the case may be) shall have been advised by counsel that a conflict of interest is likely to exist if the same counsel were to represent such Investor Party or such Company Party and the indemnifying party (in which case, if such Investor Party or such Company Party (as the case may be) notifies the indemnifying party in writing that it elects to employ separate counsel at the expense of the indemnifying party, then the indemnifying party shall not have the right to assume the defense thereof on behalf of the indemnified party and such counsel shall be at the expense of the indemnifying party, provided further that in the case of clause (iii) above the indemnifying party shall not be responsible for the reasonable fees and expenses of more than one (1) separate legal counsel for all Investor Parties or Company Parties (as the case may be). The Company Party or Investor Party (as the case may be) shall reasonably cooperate with the indemnifying party in connection with any negotiation or defense of any such action or Claim by the indemnifying party and shall furnish to the indemnifying party all information reasonably available to the Company Party or Investor Party (as the case may be) which relates to such action or Claim. The indemnifying party shall keep the Company Party or Investor Party (as the case may be) reasonably apprised at all times as to the status of the defense or any settlement negotiations with respect thereto. No indemnifying party shall be liable for any settlement of any action, claim or proceeding effected without its prior written consent; provided, however, the indemnifying party shall not unreasonably withhold, delay or condition its consent. No indemnifying party shall, without the prior written consent of the Company Party or Investor Party (as the case may be), consent to entry of any judgment or enter into any settlement or other compromise which does not include as an unconditional term thereof the giving by the claimant or plaintiff to such Company Party or Investor Party (as the case may be) of a release from all liability in respect to such Claim or litigation, and such settlement shall not include any admission as to fault on the part of the Company Party. For the avoidance of doubt, the immediately preceding sentence shall apply to Sections 6(a) and 6(b) hereof. Following indemnification as provided for hereunder, the indemnifying party shall be subrogated to all rights of the Company Party or Investor Party (as the case may be) with respect to all third parties, firms or corporations relating to the matter for which indemnification has been made. The failure to deliver written notice to the indemnifying party within a reasonable time of the commencement of any such action shall not relieve such indemnifying party of any liability to the Investor Party or Company Party (as the case may be) under this Section 6, except to the extent that the indemnifying party is materially and adversely prejudiced in its ability to defend such action.

- (d) No Person involved in the sale of Registrable Securities who is guilty of fraudulent misrepresentation (within the meaning of Section 11(f) of the Securities Act) in connection with such sale shall be entitled to indemnification from any Person involved in such sale of Registrable Securities who is not guilty of fraudulent misrepresentation.
- (e) The indemnification required by this Section 6 shall be made by periodic payments of the amount thereof during the course of the investigation or defense, as and when bills are received or Indemnified Damages are incurred; <u>provided</u> that the Investor shall promptly reimburse the Company for all such payments to the extent a court of competent jurisdiction determines that any Investor Party was not entitled to such payments.
- (f) The indemnity and contribution agreements contained herein shall be in addition to (i) any cause of action or similar right of the Company Party or Investor Party against the indemnifying party or others, and (ii) any liabilities the indemnifying party may be subject to pursuant to the law.

Contribution.

To the extent any indemnification by an indemnifying party is prohibited or limited by law, the indemnifying party agrees to make the maximum contribution with respect to any amounts for which it would otherwise be liable under Section 6 to the fullest extent permitted by law; provided, however: (i) no contribution shall be made under circumstances where the maker would not have been liable for indemnification under the fault standards set forth in Section 6 of this Agreement, (ii) no Person involved in the sale of Registrable Securities which Person is guilty of fraudulent misrepresentation (within the meaning of Section 11(f) of the Securities Act) in connection with such sale shall be entitled to contribution from any Person involved in such sale of Registrable Securities who was not guilty of fraudulent misrepresentation; and (iii) contribution by any seller of Registrable Securities shall be limited in amount to the amount of net proceeds received by such seller from the applicable sale of such Registrable Securities pursuant to such Registration Statement. Notwithstanding the provisions of this Section 7, the Investor shall not be required to contribute, in the aggregate, any amount in excess of the amount by which the net proceeds actually received by the Investor from the applicable sale of the Registrable Securities subject to the Claim exceeds the amount of any damages that the Investor has otherwise been required to pay, or would otherwise be required to pay under Section 6(b), by reason of such untrue or alleged untrue statement or omission or alleged omission.

8. Reports Under the Exchange Act

With a view to making available to the Investor the benefits of Rule 144, the Company agrees to:

- (a) use its reasonable best efforts to make and keep public information available, as those terms are understood and defined in Rule 144;
- (b) use its reasonable best efforts to file with the SEC in a timely manner all reports and other documents required of the Company under the Securities Act and the Exchange Act so long as the Company remains subject to such requirements (it being understood that nothing herein shall limit any of the Company's obligations under the Purchase Agreement) and the filing of such reports and other documents is required for the applicable provisions of Rule 144;
- (c) furnish to the Investor so long as the Investor owns Registrable Securities, promptly upon request, (i) a written statement by the Company, if true, that it has complied with the reporting, submission and posting requirements of Rule 144 and the Exchange Act, (ii) a copy of the most recent annual or quarterly report of the Company and such other reports and documents so filed by the Company with the SEC if such reports are not publicly available via EDGAR, and (iii) such other information as may be reasonably requested to permit the Investor to sell such securities pursuant to Rule 144 without registration; and
- (d) take such additional action as is reasonably requested by the Investor to enable the Investor to sell the Registrable Securities pursuant to Rule 144, including, without limitation, delivering all such legal opinions, consents, certificates, resolutions and instructions to the Company's Transfer Agent as may be reasonably requested from time to time by the Investor and otherwise fully cooperate with Investor and Investor's broker to effect such sale of securities pursuant to Rule 144.

Assignment of Registration Rights.

All or any portion of the rights under this Agreement shall be automatically assignable by the Investor to any transferee or assignee of all or any portion of the Investor's Registrable Securities if: (i) the Investor agrees in writing with such transferee or assignee to assign all or any portion of such rights, and a copy of such agreement is furnished to the Company within a reasonable time after such assignment; (ii) the Company is, within a reasonable time after such transfer or assignment, furnished with written notice of (a) the name and address of such transferee or assignee, and (b) the securities with respect to which such registration rights are being transferred or assigned; (iii) immediately following such transfer or assignment the further disposition of such securities by such transferee or assignee is restricted under the Securities Act or applicable state securities laws if so required; (iv) at or before the time the Company receives the written notice contemplated by clause (ii) of this sentence such transferee or assignee agrees in writing with the Company to be bound by all of the provisions contained herein; (v) such transfer or assignment shall have been made in accordance with the applicable requirements of the Purchase Agreement; and (vi) such transfer or assignment shall have been conducted in accordance with all applicable federal and state securities laws. The term "Investor" in this Agreement shall also include all such transferees and assignees.

10. Amendment or Waiver.

No provision of this Agreement may be amended or waived by the parties from and after the date that is one Trading Day immediately preceding the initial filing of the Registration Statement with the SEC. Subject to the immediately preceding sentence, no provision of this Agreement may be (i) amended other than by a written instrument signed by both parties hereto or (ii) waived other than in a written instrument signed by the party against whom enforcement of such waiver is sought. Failure of any party to exercise any right or remedy under this Agreement or otherwise, or delay by a party in exercising such right or remedy, shall not operate as a waiver thereof.

11. Miscellaneous.

- (a) Solely for purposes of this Agreement, a Person is deemed to be a holder of Registrable Securities whenever such Person owns or is deemed to own of record such Registrable Securities. If the Company receives conflicting instructions, notices or elections from two or more Persons with respect to the same Registrable Securities, the Company shall act upon the basis of instructions, notice or election received from such record owner of such Registrable Securities.
- (b) Any notices, consents, waivers or other communications required or permitted to be given under the terms of this Agreement shall be given in accordance with Section 10.4 of the Purchase Agreement.
- (c) Failure of any party to exercise any right or remedy under this Agreement or otherwise, or delay by a party in exercising such right or remedy, shall not operate as a waiver thereof. The Company and the Investor acknowledge and agree that irreparable damage would occur in the event that any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached. It is accordingly agreed that either party shall be entitled to an injunction or injunctions to prevent or cure breaches of the provisions of this Agreement by the other party and to enforce specifically the terms and provisions hereof (without the necessity of showing economic loss and without any bond or other security being required), this being in addition to any other remedy to which either party may be entitled by law or equity.
- (d) All questions concerning the construction, validity, enforcement and interpretation of this Agreement shall be governed by the internal laws of the State of New York, without giving effect to any choice of law or conflict of law provision or rule (whether of the State of New York or any other jurisdictions) that would cause the application of the laws of any jurisdictions other than the State of New York. Each party hereby irrevocably submits to the exclusive jurisdiction of the federal courts sitting in The City of New York, Borough of Manhattan, for the adjudication of any dispute hereunder or in connection herewith or with any transaction contemplated hereby or discussed herein, and hereby irrevocably waives, and agrees not to assert in any suit, action or proceeding, any claim that it is not personally subject to the jurisdiction of any such court, that such suit, action or proceeding is brought in an inconvenient forum or that the venue of such suit, action or proceeding is improper. Each party hereby irrevocably waives personal service of process and consents to process being served in any such suit, action or proceeding by mailing a copy thereof to such party at the address for such notices to it under this Agreement and agrees that such service shall constitute good and sufficient service of process and notice thereof. Nothing contained herein shall be deemed to limit in any way any right to serve process in any manner permitted by law. If any provision of this Agreement shall be invalid or unenforceable in any jurisdiction, such invalidity or unenforceability of the remainder of this Agreement in that jurisdiction or the validity or enforceability of any provision of this Agreement in any other jurisdiction. EACH PARTY HEREBY IRREVOCABLY WAIVES ANY RIGHT IT MAY HAVE TO, AND AGREES NOT TO REQUEST, A JURY TRIAL FOR THE ADJUDICATION OF ANY DISPUTE HEREUNDER OR IN CONNECTION HEREWITH OR ARISING OUT OF THIS AGREEMENT OR ANY TRANSACTION CONTEMPLATED HEREBY.

(e) The Transaction Documents set forth the entire agreement and understanding of the parties solely with respect to the subject matter thereof and supersedes a
prior and contemporaneous agreements, negotiations and understandings between the parties, both oral and written, solely with respect to such matters. There are no promise
undertakings, representations or warranties by either party relative to subject matter hereof not expressly set forth in the Transaction Documents. Notwithstanding anything in the
Agreement to the contrary and without implication that the contrary would otherwise be true, nothing contained in this Agreement shall limit, modify or affect in any mann
whatsoever (i) the conditions precedent to a Draw Down contained in Article VII of the Purchase Agreement, including, without limitation, the condition precedent contained in
Section 7.2(iii) thereof or (ii) any of the Company's obligations under the Purchase Agreement.

- (f) Subject to compliance with Section 9, this Agreement shall inure to the benefit of and be binding upon the permitted successors and assigns of each of the parties hereto. This Agreement is not for the benefit of, nor may any provision hereof be enforced by, any Person, other than the parties hereto, their respective permitted successors and assigns and the Persons referred to in Sections 6 and 7 hereof.
- (g) The headings in this Agreement are for convenience of reference only and shall not limit or otherwise affect the meaning hereof. Unless the context clearly indicates otherwise, each pronoun herein shall be deemed to include the masculine, feminine, neuter, singular and plural forms thereof. The terms "including," "includes," "include" and words of like import shall be construed broadly as if followed by the words "without limitation." The terms "herein," "hereunder," "hereof" and words of like import refer to this entire Agreement instead of just the provision in which they are found.
- (h) This Agreement may be executed in two or more identical counterparts, all of which shall be considered one and the same agreement and shall become effective when counterparts have been signed by each party and delivered to the other party. If any signature is delivered by facsimile transmission or by an e-mail which contains a portable document format (.pdf) file of an executed signature page, such signature page shall create a valid and binding obligation of the party executing (or on whose behalf such signature is executed) with the same force and effect as if such signature page were an original thereof.
- (i) Each party shall do and perform, or cause to be done and performed, all such further acts and things, and shall execute and deliver all such other agreements, certificates, instruments and documents as any other party may reasonably request in order to carry out the intent and accomplish the purposes of this Agreement and the consummation of the transactions contemplated hereby.
- (j) The language used in this Agreement will be deemed to be the language chosen by the parties to express their mutual intent and no rules of strict construction will be applied against any party.

[signature pages follow]

IN WITNESS WHEREOF, Investor and the Company have caused their respective signature page to this Registration Rights Agreement to be duly executed as of the date first written above.

COMPANY:

TERRA TECH CORP.

By: /s/ Derek Peterson

Name: Derek Peterson Title: Chief Executive Officer

IN WITNESS WHEREOF, Investor and the Company have caused their respective signature page to this Registration Rights Agreement to be duly executed as of the date first written above.

INVESTOR:

HANOVER HOLDINGS I, LLC, a New York limited liability company

By: /s/Joshua Sason

Name: Joshua Sason Its: Managing Member

STOCK REDEMPTION AGREEMENT

This Stock Redemption Agreement (this "Agreement") is made between Terra Tech Corp., a Nevada corporation (the "Company"), and Amy Almsteier (the "Selling Shareholder") this 6th day of May 2013.

RECITALS

- A. The Selling Shareholder is the owner of 11,700,000 shares of common stock (the "Redemption Shares"), par value \$.001 per share, of the Company.
- B. The Company desires to redeem the Redemption Shares from the Selling Shareholder, and the Selling Shareholder desires to have the Redemption Shares redeemed by the Company, upon the terms and conditions set forth in this Agreement.

ACCORDINGLY, the parties agree as follows:

- 1. Redemption Price. The Company shall redeem the Redemption Shares from the Selling Shareholder for the aggregate redemption price of US \$5.00, which amount is hereby acknowledged as having been received in cash by the Company from the Selling Shareholder (the "Redemption Price"). Selling Shareholder hereby assigns, separate from certificate, all right, title and interest in and to the Redemption Shares to the Corporation.
- 2. Selling Shareholder's Representations and Warranties.
- 2.1. The Selling Shareholder represents and warrants to the Company that: (i) the Selling Shareholder owns and holds the Redemption Shares free and clear of all liens, encumbrances and claims of other persons or entities whatsoever and subject to no options, warrants, contracts, agreements, arrangements or understandings of any kind; and (ii) the Selling Shareholder has full power and authority to transfer and deliver the Redemption Shares to the Company in accordance with the terms of this Agreement, and the consummation of the redemption transaction provided for in this Agreement shall not constitute the breach of any term or provision of, or constitute a default under, any agreement or other instrument to which the Selling Shareholder is a party.
- 2.2 The Selling Shareholder further represents and warrants to the Company that the Selling Shareholder has been advised to consult with, and has consulted or chosen not to consult with, independent advisers with respect to the fairness of the Redemption Price and the other terms of this Agreement.
- 3. <u>Successors and Assigns</u>. This Agreement shall be binding on and shall inure to the benefit of the parties to this Agreement and their respective spouses, successors, assignees, heirs and personal representatives.

- 4. Entire Agreement. This Agreement supersedes any and all oral or written agreements previously made relating to the subject matter of this Agreement, and constitutes the entire agreement of the parties with respect to such subject matter.
- 5. Amendment. This Agreement may be modified or amended only in writing signed by both parties.
- 6. <u>Further Assurances</u>. Each party shall execute and deliver any and all additional documents and instruments and shall take all actions reasonably requested by the other party in order to carry out the intent of this Agreement.
- 7. Counterparts. This Agreement may be executed in counterparts and by facsimile or scanned e-mail attachment, each of which shall considered an original, but both of which together shall constitute the same document.

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be executed as of the date first written above.

COMPANY:

TERRA TECH CORP.:

By: /s/ Derek Peterson

Name: Derek Peterson

Title: President and Chief Executive Officer

SELLING SHAREHOLDER:

By: /s/ Amy Almsteier

Name: Amy Almsteier